Western SGraduate & Postdoctoral Studies

Western University Scholarship@Western

Electronic Thesis and Dissertation Repository

8-20-2015 12:00 AM

The Role of Androgens in Life History Theories of Attachment

Janani S. Sankar The University of Western Ontario

Supervisor Dr. Elizabeth Hampson *The University of Western Ontario*

Graduate Program in Psychology A thesis submitted in partial fulfillment of the requirements for the degree in Doctor of Philosophy © Janani S. Sankar 2015

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Biological Psychology Commons

Recommended Citation

Sankar, Janani S., "The Role of Androgens in Life History Theories of Attachment" (2015). *Electronic Thesis and Dissertation Repository*. 3060. https://ir.lib.uwo.ca/etd/3060

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

THE ROLE OF ANDROGENS IN LIFE HISTORY THEORIES OF ATTACHMENT

(Thesis format: Monograph)

by

Janani S. Sankar

Graduate Program in Psychology

A thesis submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy

The School of Graduate and Postdoctoral Studies The University of Western Ontario London, Ontario, Canada

© Janani S. Sankar 2015

Abstract

Life history theories (LHTs) of attachment address how attachments to caregivers in infancy/childhood and to romantic partners in adulthood are used to negotiate mating and reproductive choices. Greater insecure-avoidant attachment has been suggested to be associated with the adoption of a low-investment, short-term reproductive strategy. The role of sex hormones, including the androgen testosterone (T), in the development of attachment-related reproductive strategies has been speculated in some LHTs. This research tested an integrated-LHT model of early environment, attachment, and reproductive strategies in men, using structural equation modeling. Androgen-related effects were hypothesized to occur prenatally and/or in adulthood, consistent with various LHTs of attachment. A sample of 195 young men (M = 21.06 years) from the University of Western Ontario completed self-report paper-and-pencil questionnaires assessing their romantic attachment style, retrospective attachment to caregivers in childhood, sexual variables (e.g., sociosexuality, age at first intercourse), and personality variables such as aggression, impulsivity, and risk-taking. Testosterone was measured in saliva, while an indirect estimate of prenatal T was derived from the 2D:4D finger length ratio. Degree of androgen receptor (AR) sensitivity, as indexed by the CAG repeat polymorphism in the AR gene, was also obtained. Results showed that adult romantic attachment style (avoidant vs. anxious) mediated the relationship between childhood attachment insecurity and men's sexual reproductive strategy. Greater avoidance predicted a more opportunistic sexual strategy and greater anxiety predicted lower levels of the same strategy. Degree of childhood attachment insecurity, as retrospectively reported, mediated the relationship between quality of early family structure and engagement in non-sexual evocative behaviours believed to be associated with a more opportunistic reproductive strategy (non-sexual reproductive strategy). Adult T was an independent positive predictor of avoidant attachment, and of the non-sexual reproductive strategy, while weaker AR responsivity predicted higher levels of romantic attachment anxiety. Furthermore, romantic attachment configurations were found to mediate the relationship between and rogenic variables and sexual behaviour. These findings highlight the figural role of attachment in life history based models of mating strategies, and provide some of

ii

the first empirical support for the hypothesis that romantic attachment in men is, in part, sex-hormonally-based.

Keywords: Attachment, romantic attachment, life history theory, testosterone, sex hormone, androgen, digit ratio, CAG repeat, men, sexual strategy

Acknowledgments

"Appreciation is a wonderful thing. It makes what is excellent in others belong to us as well."

-- Voltaire

First and foremost, I must convey my sincerest gratitude to my research advisor Dr. Elizabeth Hampson for her integral role in the development and execution of the work described in this thesis, and for her guidance, encouragement, and patience throughout my graduate schooling. I am forever appreciative of your dedication and enthusiasm towards fostering my research interests and clinical passions. I have become a more thoughtful researcher and clinician under your guidance and your mentorship has been, and will continue to be, invaluable as I move forward with my career.

I must also give special thanks to Dr. Paul Tremblay for his assistance and valuable consultations related to the data analysis for this study. I am truly appreciative of the time and effort you took to field my many questions as I navigated a new and complex statistical method.

To the members of my advisory committee, Dr. Lorne Campbell and Dr. Greg Moran, thank you for sharing your expertise and providing helpful suggestions at various stages during this doctoral work.

I must recognize Bavani Rajakumar for completing all the hormonal assays that were so integral to this research. Similarly, a special thank you to The Centre for Applied Genomics in Toronto, ON who completed the DNA genotyping for this study.

The support of family and friends during this long and eventful process cannot go unnoticed. I must extend a heartfelt thanks to my fellow lab members past and present (Kelly, Adrian, Mark, Mia, Bavani), with whom I have shared a unique and meaningful experience. Words do not adequately express the gratitude I have for family and friends who have stood by me through it all. I am indebted to the support I have received, in all shapes and sizes, and for the strength of character you have built in me. Finally, it goes without saying that these years of hard work would not have reached their successful culmination without the unyielding support of my husband James. You have been the silent partner through this journey and the sounding board when I needed it. I have drawn on your intellect and your spirit, both of which gave me the courage to persevere in the face of what sometimes felt like an insurmountable task. I am so very lucky to have you by my side. In reaching this ultimate academic goal, it is our journey together that has, and continues to be, the greatest learning experience of them all.

Table of Contents

Abstract	ii
Acknowledgments	iv
Table of Contents	vi
List of Tables	ix
List of Figures	X
List of Appendices	xi
1. Introduction	1
1.1. Research Overview	4
1.2. Attachment within a Life History Theory Perspective	5
1.2.1. Belsky, Steinberg, and Draper Model (Psychosocial Acceleration Theory	v) 5
1.2.2. Chisholm Model	10
1.2.3. Del Giudice Model	13
1.2.4. Miller and Pasta Model	14
1.3. Behavioural Correlates of Attachment Style	16
1.3.1. Sexual/Reproductive Behaviours	16
1.3.2. Externalizing Behaviours	18
1.3.3. Internalizing Behaviours	22
1.4. The Biological Basis of Attachment	23
1.5. Testosterone: Mechanisms of Action	27
1.6. Is There an Effect of Prenatal Testosterone on Attachment and Related	
Behaviours?	29
1.6.1. Prenatal Androgen Exposure and Attachment Behaviours	29
1.6.2. Prenatal Androgen Exposure and Attachment-Related Personality Profi	le:
Evidence from Non-Human Animal Studies	30

1.6.3. Prenatal Androgen Exposure and Attachment-Related	Personality Profile:
Evidence from Human Studies	
1.7. Is There an Effect of Adult Testosterone on Attachment an	d Related Behaviours?
1.7.1. Sexual/Relationship Behaviour	
1.7.2. Aggressive Behaviour	
1.7.3. Other Externalizing Behaviours	
1.7.4. Internalizing Behaviours	
1.7.5. Attachment Behaviours	
1.7.6. Summary	
1.8. A Consideration of Androgen Receptor (AR) Functionality	<i>r</i>
1.9. Model Presentation	
2. Method	53
2.1. Sample and Recruitment	
2.2. Procedure	
2.3. Instruments and Measures	
2.3.1. Attachment Scales	
2.3.2. Personality and Motivational Scales	
2.4. Hand-Related Measurements	
2.5. AR Genotyping	
2.6. Testosterone Measurement	
2.7. Data Analytic Approach	
3. Results	72
3.1. Data Treatment	
3.2. Analysis of Life History Theory-Based Model	
3.2.1. Overview of Indicator Variables	
3.2.2. Descriptive Statistics	

3.2.3. Evaluation of Assumptions	
3.2.4. Correlations among Model Variables	85
3.2.5. Measurement Model	91
3.2.6. Full Structural Equation Model Analyses	101
4. General Discussion	111
4.1. Overview of Research Goals	111
4.2. Summary of Research Findings	113
4.3. The Integrated Life History Theory-Based Model	114
4.4. The Role of Androgens Within an Integrated Life History Model	119
4.5. Limitations	126
4.6. Study Contributions and Directions for Future Research	129
4.7. Conclusion	131
5. References	132
Appendices	175
Curriculum Vitae	

List of Tables

Table 1: Summary of LHTs of Attachment	6
Table 2: Demographics of Study Participants	55
Table 3: Descriptive Statistics for Model Variables	82
Table 4: Correlations among LHT-Based Model Variables	86
Table 5: Standardized Factor Loadings for Early Family Environment Latent	
Variable	93
Table 6a: Standardized Factor Loadings for Sexual Reproductive Strategy Latent	
Variable	95
Table 6b: Standardized Factor Loadings for Modified Sexual Reproductive Strategy	
Latent Variable	95
Table 7a: Standardized Factor Loadings for Adult Attachment Latent Variable	96
Table 7b: Standardized Factor Loadings for Insecure/Avoidant Attachment Styles	96
Table 7c: Standardized Factor Loadings for Modified Avoidant Attachment Latent	
Variable	96
Table 8: Standardized Factor Loadings for Non-Sexual Reproductive Strategy Latent	
Variable	99
Table 9: CFA with Standardized Factor Loadings and Correlations	.100

List of Figures

Figure 1:	Integrated life history based model	49
Figure 2:	Illustration of hand-related measurements	69
Figure 3:	Indicators of Early Family Environment latent construct	75
Figure 4:	Indicators of Non-Sexual Reproductive Strategy latent construct	77
Figure 5:	Indicators of Sexual Reproductive Strategy latent construct	79
Figure 6:	Structural model	104
Figure 7:	Structural model run with only significant pathways	106
Figure 8:	Summary of mediation analyses	108

List of Appendices

Appendix A: Poster for Recruitment from Western Campus	175
Appendix B: Order of Questionnaire Presentation	176
Appendix C: Demographics Questionnaire and Dating History Questionnaire.	177
Appendix D: Detailed Description of Indicators of Latent Constructs in the LF	IT-Based
Model	
Appendix E: Ethics Approval Forms	191

1. Introduction

Attachment theory (Bowlby, 1969/1982; 1973; 1980) provides a framework through which to understand the nature and development of an individual's close emotional bonds with a caregiver or loved one. Originating in infancy, attachment is conceptualized as a "behavioural system", the purpose of which is to innately motivate the child to form affective bonds with caregivers by organizing his/her own behaviours in order to maintain proximity to the caregiver. Infants' biological predisposition for proximity seeking (Bowlby, 1969/1982) is viewed as evolutionarily adaptive, serving to increase the likelihood of being protected from danger, thereby resulting in a survival advantage. While emotional and/or physical proximity seeking is central to all attachment relationships, there is, however, individual variation in the types of attachment behaviours used to achieve this goal. Differences in the local environment in which humans develop, including early family experience, lead individuals to differ in the careeliciting behaviours used to maximize caregiver ability. The result is the development of different learned *attachment styles*.

Two broad categories of attachment have been recognized: secure and insecure (Ainsworth et al., 1978; Main & Goldwyn, 1984). Children who are securely attached generally experience consistently available, sensitive caregiving. Such relational experience allows these children to use the caregiver as a "secure base" for exploration of the world and to seek help and comfort from him or her when distressed. Children who are insecurely attached can be further classified into three distinct types: avoidant, anxious/ambivalent, and disorganized. Insecure-avoidant children generally experience a rejecting, cold, and uninvolved caregiver and treat the caregiver as unavailable and do not ask for help or comfort when distressed. The insecure-anxious/ambivalent style is characterized by inconsistent caregiver availability, which results in a child who is easily distressed and asking vigorously for help and comfort but not easily calmed. Finally, children exposed to frightening or threatening parenting behaviour may develop attachment disorganization. This pattern is associated with both approach and avoidance behaviours as caregivers are simultaneously seen as sources of comfort and fear. These behaviours, although beginning in infancy, are not limited to this period in human development. Bowlby (1969/1982,1979) speculated that attachment is an integral part of human existence "from cradle to the grave", and that attachment patterns formed in infancy and childhood represent internal working models (IWMs) that are used to anticipate, interpret, and guide current and *future* interpersonal interactions. Research has demonstrated continuity in attachment patterns from childhood to adulthood (correlation size \approx .40) (e.g. Hamilton, 2000; Waters et al., 2000; Fraley, 2002), with level of stability in patterns dependent on the persistence of IWMs formed in childhood (e.g., Bowlby, 1969/1982; Sroufe, 1979; Sroufe, Egeland, & Kreutzer, 1990; for further review see Fraley, 2002) and exposure to factors that can disrupt them (i.e., environmental changes or seminal events experienced later in development) (e.g., Thompson, Lamb, & Estes, 1982; Trivers, 1985; Scharfe & Bartholomew, 1994, Lewis, 1997, Waters et al., 2000). In adulthood, the concept of attachment can be applied to romantic/love relationships (Hazan & Shaver, 1987), with the romantic partner viewed as the prototypical instantiation of an attachment figure (Hazan & Shaver, 1987; 1994; Ainsworth, 1991).

Assessment of attachment to romantic partners in adulthood is mostly accomplished through self-report questionnaires, based on the social psychological tradition, although interview-based protocols do exist (e.g., Bartholomew & Horowitz, 1991). In their original conceptualization, Hazan and Shaver (1987) extrapolated the main features of the three major infant attachment styles described by Ainsworth and colleagues (1978) to assess individuals' alignment with secure, anxious-ambivalent, and avoidant patterns of attachment in a romantic context. Since then, factor analytic study of many self-report attachment questionnaires has revealed two robust dimensions of anxiety and avoidance that underlie romantic attachment patterns (Brennan, Clark, & Shaver, 1998). Thus, adult romantic attachment is generally conceptualized along the two dimensions of anxiety and avoidance (Fraley & Shaver, 2000; Brennan et al., 1998, Bartholomew & Horowitz, 1991), such that securely attached adults exhibit a pattern of little worry or concern about the availability and willingness of significant others to meet attachment needs (low anxiety) and little discomfort and avoidance of emotional intimacy (low avoidance). Among the patterns of insecure attachment, *dismissing-avoidant* adults, as so-named by Bartholomew and Horowitz (1991), display low anxiety and high avoidance. These individuals distance themselves from their romantic partners, show a low need for closeness, and describe themselves as self-sufficient. In contrast, *preoccupied* (anxious-ambivalent) adults display a pattern of high anxiety and low avoidance, report a strong desire for intimacy, feel uncomfortable when not involved in close relationships, and worry about being rejected by their partners. Finally, what Bartholomew and Horowitz (1991) called *fearful-avoidant* adults are both highly anxious and highly avoidant and display a mix of desire for closeness and fear of rejection.

Bartholomew (1990) suggested that the dimensions of anxiety and avoidance may overlap with individuals' images of themselves and of others, which Bowlby (1973) viewed as defining features of attachment working models. Positivity vs. negativity of the 'self' model-the degree to which individuals have internalized a sense of their own self-worth, and which may correspond to levels of anxiety/dependency in close relationships, intersects with positivity vs. negativity of the 'other' model-the degree to which others are generally expected to be available and supportive, and which may correspond with the tendency to seek out or avoid closeness in relationships. Thus a positive model of the self and others corresponds with a secure attachment style, while a positive model of the self combined with a negative model of others corresponds to a dismissing-avoidant style, a negative self model and positive other model corresponds to a preoccupied style, and negative self and other models corresponds to the fearfulavoidant style. Factor analytic study and multidimensional scaling analyses have provided empirical support for the four-group model (Bartholomew & Horowitz, 1991; see Feeney, 2008 for review). This is consistent with reports of a fourth infant attachment style marked by characteristics of both avoidance and anxiety (e.g., Main, Kaplan, & Cassidy, 1985; Main & Soloman, 1990).

Given that attachment behaviours are not simply a characteristic of early development, but also extend into adulthood, it is important to understand their function in adult life. The emotional bond formed between an individual and his/her romantic partner can also be viewed as evolutionarily adaptive (Zeifman & Hazan, 2008). Although not providing survival value, as in infancy or childhood, attachment to a romantic partner is reproductively advantageous. By increasing the likelihood that one's genes will be passed on to the next generation, romantic attachment may be viewed as a proximate mechanism to enhance reproductive success (Zeifman & Hazan, 2008). Consequently, an individual's approach to attachment with romantic partners in adulthood may have important consequences for mating and reproduction (Del Giudice & Belsky, 2010).

1.1. Research Overview

The current dissertation was directed toward two broad aims. The first purpose was to empirically investigate the role of attachment in the development of mating and reproductive strategies in men. As will be described in Section 1.2, various *life history theories* of attachment exist, which have postulated how early environmental experiences and attachment relationships influence the development of corresponding approaches to sexual behaviour and mating. Although various theories have been proposed, there has been little attempt to put these models to an empirical test and as a result the construct of adult romantic attachment and how it relates to various developmental antecedents and reproductive strategies has been poorly articulated. The present research represented a first, systematic attempt to assess the function and derivation of attachment within the context of an integrated-life history theory model. The integrated model to be tested and the hypothesized predictive relationships will be outlined in Section 1.9. An overview of

The second aim of the present study was to incorporate and test a novel biological perspective on attachment (see Section 1.4) by investigating the role of testosterone in the development of individual differences in attachment style, and within the context of life history theories of attachment. Life history theories have broadly implicated sex hormones, including the androgen testosterone, but specific details are lacking. As will be presented in Sections 1.3, 1.6, and 1.7, the behavioural correlates of the insecure-avoidant romantic attachment style, which has been proposed to represent a more male-typed attachment pattern, can be shown to overlap with many of the behavioural

correlates of high testosterone exposure that have been identified in the neuroendocrinology literature.

These aims will be discussed in further detail in Section 1.9, following an overview of the life history perspective.

1.2. Attachment within a Life History Theory Perspective

In the past three decades a significant body of work within the broader attachment literature has begun to integrate ideas derived from evolutionary biology with conventional thought and empirical observation on adult romantic attachment (see Simpson & Belsky, 2008). By focusing on its *reproductive* consequences, life history theories (LHTs) of attachment attempt to extend the concept of evolutionary adaptiveness of attachment across the lifespan (e.g., Belsky, Steinberg, & Draper, 1991; Chisholm, 1999; Del Giudice, 2009). LHT approaches address, in general, how an organism allocates time and resources in order to optimize his/her reproductive fitness (for review see Kaplan & Gangestad, 2005). Attachment relationships established in early life are believed to provide crucial information about the safety and predictability of an individual's local environment, which in turn are used to negotiate three strategic reproductive trade-offs later in life: (1) current *vs.* future reproduction, (2) quantity *vs.* quality of offspring, (3) mating *vs.* parenting effort.

I will now present the major tenets of the predominant LHTs of attachment that have been discussed in the literature (please refer to Table 1 which provides a summary of the various theories reviewed). Following this, I will describe some of the main evidence in support of these theories, including a review of the existing literature on associations between attachment style and behaviours that have been implicated by the LHTs of attachment and reproduction.

1.2.1. Belsky, Steinberg, and Draper Model (Psychosocial Acceleration Theory)

In the first systematic attempt to integrate attachment theory with the life history perspective, Belsky et al. (1991) proposed that in stressful conditions, characterized by

Table 1

		Relevant	~ ·	~	Role of
Theory	Attachment	Environmental	Somatic	Reproductive	Romantic
	Classification	Antecedents	Development	Strategy	Attachment
Belsky et	Insecure	 High stress Marital discord 	accelerated	Short-term Opportunistic	Dismissing-
	avoidant)	 Maintai discolu Single parenthood 	timing of	• Opportunistic	feature of
	u (of uulit)	 Insensitive parenting 	pubertal	activity	reproductive
		Unstable resources	maturation	Low investment	strategy
al. (1991)/	Secure	Low stress	On time/ rel.	Long-term	Secure
Belsky	~ ~ ~ ~ ~ ~ ~	Marital harmony	delayed timing	Later sexual	attachment in
(1997)		Supportive parenting	of pubertal	activity	adulthood
		Adequate resources	maturation	 Investment- 	
				oriented	
	Insecure-			• "Helper-at nest"	Preoccupied
	anxious			 Delayed 	attachment in
				strategy	adulthood
				• Relationship	
	T			dependency	
	Insecure	High parental stress		• In men: Young	Not
		High local mortality rate		Increased	Implicated
		Parental absence		sexuality risk-	
		Unstable resources		taking.	
				impulsivity,	
Chisholm				aggression	
(1999)				• "Uncommitted"	
				style	
	Secure	 Low parental stress 		• In men: lower	Not
		• Low local mortality		levels of Young	implicated
		rate		foaturos	
		Parentar presence Stable recourses		 Increased 	
		• Stable resources		relationship	
				commitment	
	Insecure \rightarrow	• High stress	Relatively	• Short-term	Avoidant-
	avoidance	• Insensitive parenting	accelerated	 Uncommitted 	based
	(boys)	• Parental	adrenarche	mating	reproductive
		unwillingness to		 Opportunistic 	strategies in
		invest		sexual	adulthood
Del				behaviour	
(2009)	Insecure \rightarrow	Moderate stress	Relatively	• Investment-	Anxiety-
(200))	anxiety (girls)	• Parental inability to	accelerated	eliciting	reproductive
		invest	aurenarene	Closeness with partners	strategies in
				Short-term	adulthood
				orientation in	
				females	
	Secure	Low stress	Normal timing	Long-term	Secure
		 Sensitive parenting 	of adrenarche	• High	attachment to
				relationship	partners in
				commitment	adulthood
				• Future	
				reproduction	

Summary of LHTs of Attachment

Table 1 continued

	Č Č	D 1			D 1 0
		Relevant			Role of
Theory	Attachment	Environmental	Somatic	Reproductive	Romantic
	Classification	Antecedents	Development	Strategy	Attachment
Miller	Not discussed	 Child's perception of low family affection Child's perception of low fear 	Relatively accelerated pubertal timing	 Early and frequent mating No added push for childbearing 	Not discussed
and Pasta (2000)	Not discussed	Low family affectionHigh perception of fear	Relatively accelerated pubertal timing	Opportunistic childbearing	Not discussed
	Not discussed	 High family affection Low perception of fear 	Relatively delayed pubertal timing	High parental investment	Not discussed

Summary of LHTs of Attachment

marital discord, single parenthood, and unstable employment, among other features, parenting style tends to become harsher, less sensitive, and/or more unpredictable, leading a child to adopt a more insecure attachment pattern. Attachment insecurity is then used as a working model through which the child interprets his/her environment: learning that resources are scarce and unpredictable, that people are unreliable, and that interprets and bonds are fleeting and undependable. The child's subjective experience of the rearing environment and parental relationship behaviour will result in the development of an understanding of what the future will hold for his/her *own* relationships, with more exposure to unstable relationship behaviour associated with the belief that mating relationships tend to be short and uncommitted.

Belsky and colleagues (1991) suggest that the outcome of this psychological development is accelerated somatic development, specifically accelerated timing of pubertal maturation. The authors argue that in contexts of early relational stress, and by extension insecure attachment, it may be adaptive for individuals to reliably increase reproductive investment in order to increase the probability of having their genes passed on to the next generation. One way this could be facilitated is through accelerated timing of pubertal maturation, within a biologically acceptable range, enabling for sexual activity and reproduction to begin at a relatively earlier age. Consequently, earlier maturation should be associated with a tendency to adopt a reproductive strategy based on a short-term, opportunistic relationship orientation, early reproduction, and low parental investment, as proposed by Belsky et al. (1991). The exact biological mechanism by which early relational factors influence pubertal timing has not been determined (Mustanski et al., 2004; Graber & Brooks-Gunn, 1996), but could be related to increased eating in high-risk environments and associated rapid weight gain and growth that lead to earlier pubertal onset (e.g., Kuzawa et al., 2010; Bogaert, 2005).

Several studies have demonstrated that childhood exposure to psychosocial stressors accelerates pubertal timing in boys (Arim et al., 2011; Mustanski et al., 2004; Meschke et al., 2003; Chasiotis et al., 1998; Kim & Smith, 1998) and girls (e.g., Meckelmann, Pfeifer, & Rauh, 2013; James et al., 2012; Belsky et al., 2010; Ellis & Essex, 2007; Moffitt et al., 1992), and earlier maturation has an impact on subsequent sexual behaviour (e.g., James et al., 2012; Belsky et al., 2010; Kim & Smith, 1998; Smith, Udry, & Morris, 1985), as well as non-sexual risky behaviour (e.g., Wiesner & Ittel, 2002; Williams & Dunlop, 1999; Duncan et al., 1985). In contrast, the development of secure attachment is hypothesized to lead to relatively on time, or even delayed (within a biologically acceptable range), pubertal maturation, in order to allow for the child to reap the benefits (e.g., sensitive parenting, felt security) of a relatively benign and/or supportive family environment. This should then result in the adoption of a long-term, investing orientation towards mating relationships, delayed mating, and high parental investment (i.e., focus on quality). Thus from an evolutionary standpoint, the trade-off made by those who are insecurely attached is suggested to be towards current reproduction, higher offspring quantity, and a greater focus on mating effort.

Belsky (1997) further argued that the low-investment, short-term reproductive strategy predicted by insecure childhood attachment was specific to the insecure-avoidant style. Within an environment of evolutionary adaptation, children who developed avoidant attachment styles were hypothesized to continue to display dismissing-avoidant attachment behaviours in adulthood, which Belsky conceptualized as "a central feature of an opportunistic and facultative reproductive strategy" (Belsky, 1997, p. 372). In contrast, it was hypothesized that childhood insecure-anxious attachment leads to the development of so-called "helper-at-nest" behaviour, which in adulthood results in the display of behaviours that promote dependency in significant relationships. Belsky speculated that this life history interpretation of anxious/preoccupied attachment might occur more frequently among firstborns and particularly female firstborns, although this speculation has yet to receive empirical support.

The hypothesized relationship between avoidance and short-term mating orientation has been well supported in the adult attachment literature. Individuals classified as avoidant on self-report measures of romantic attachment or who display higher levels of self-reported romantic attachment avoidance on dimensional measures (e.g., Experiences in Close Relationships (ECR) Questionnaire) have been found to be more willing to engage in casual sex or sex without love (Brennan & Shaver, 1991; 1995; Feeney, Noller, & Patty, 1993; Gentzler & Kerns, 2004), more likely to have sexual "hook-ups", sex with strangers, or one-night stands than individuals who self-reported as either anxious-ambivalent or as securely attached (Cooper, Shaver, & Collins, 1998; Paul, McManus, & Hayes, 2000), to show less commitment in romantic relationships (Simpson, 1990; Gentzler & Kerns, 2004; Schmitt, 2005), and to be more likely to have dated more than one person (Kirkpatrick & Hazan, 1994). Furthermore, Jackson & Kirkpatrick (2007), who developed a multi-item self-report measure of mating orientation, found that degree of avoidance, as assessed by the ECR, was positively correlated with scores on the short-term mating orientation subscale of their measure and negatively correlated with scores on the long-term mating orientation subscale in undergraduate students. By contrast, higher levels of ECR-rated attachment anxiety were associated with *less* interest in short-term mating.

1.2.2. Chisholm Model

Building upon the model proposed by Belsky et al. (1991), Chisholm (1993, 1996, 1999) further speculated about the associations between early family experience, the formation of attachment bonds, and the subsequent development of reproductive strategies in adulthood. Although both Belsky et al. and Chisholm propose that parental behaviour and associated experiences in the rearing environment will influence attachment style formation in the infant and child, a point of divergence is how Chisholm interprets the cause of individual differences in parental behaviour and the role of parentchild attachment in this context. According to Chisholm (1993), levels of parental stress are driven by the causes and correlates of the *mortality rate* within the local environment in which children are reared, including experiences of poverty, exploitation, and disease, and the feelings of fear and hopelessness they engender. Children, in turn, indirectly and implicitly react to the outcomes of the local mortality rate by developing attachment behaviours that will maximize their reproductive success. Common individual differences in attachment can thus be interpreted as facultative adaptations to parental behaviour, which itself is a function of the death rate in the local environment. Environments with higher mortality rates are likely to be of higher risk and uncertainty for the child, setting the pathway towards a more insecure vs. secure attachment pattern, and in turn the development of an optimal reproductive strategy based on high effort

devoted toward mating, maximizing of current reproductive opportunities, and low investment in each offspring (Chisholm 1996, 1999).

Chisholm's life-history model discusses the significance of father absence and its potential negative impact on the socioeconomic status (SES) of the child, which increases risk and uncertainty and by extension environmental stress (Chisholm, 1999). The insecure attachment pattern that is proposed to develop subsequent to parental stress, economic or otherwise, is hypothesized to vary depending on the types of relational cues the child perceives to be receiving from his/her parent (Chisholm, 1996). Specifically, avoidant attachment represents a facultative adaptation to parental unwillingness to invest in the child, while anxious-ambivalent attachment represents a facultative adaptation to parental *inability* to invest. Both patterns are speculated to result in the development of short-term, opportunistic reproductive strategies, again a divergence from Belsky's (1997) theorizing, who suggests that it is avoidant attachment that specifically predicts a low-investment, short-term reproductive strategy. Empirical studies have found that early environmental stress (e.g., father absence, marital conflict, low SES) is associated with attachment insecurity (e.g., Nair & Murray, 2005; Clarke-Stewart et al., 2000; Owen & Cox, 1997; Hill, Young, & Nord, 1994a; Mickelson, Kessler, & Shaver, 1997; Riggs & Kaminski, 2010), and both have been related to the development of a short-term relationship orientation (Hill et al., 1994a).

According to Chisholm (1999), the opportunistic reproductive strategy predicted by insecure childhood attachment includes specific behavioural characteristics. Males raised in high-risk environments are proposed to adopt an alternative reproductive strategy that is based on hypermasculine behaviour, increased sexuality, aggression, impulsivity, and risk-taking, the so-called "Young Male Syndrome" (Wilson & Daly, 1985). Chisholm (1999) notes that the Young Male Syndrome (and its behavioural correlates) is itself the optimal reproductive strategy for males reared in environments of high uncertainty, which is understood as the outcome of insecure attachment.

Empirical support for the postulated environmental antecedents of the Young Male Syndrome exists. Investigations by Hill et al. (1994b) and Griskevicius et al.

(2011) have demonstrated that adults who self-report relatively few resources during childhood (i.e., higher levels of childhood socio-economic stress) engage in more risky behaviour in adulthood, such as increased alcohol consumption and/or dependence and making riskier financial decisions. Hill, Jenkins, and Farmer (2008) found that retrospective self-reported levels of environmental instability across the domains of discipline, nurturance, meals, and finances positively predicted risky behaviour and impulsivity (i.e., frequency of engagement in spur-of-the-moment behaviours reflecting loss of control). Furthermore, using structural equation modelling the authors found that the relationship between early instability and risk-taking was mediated by one's conception of the future (example item measuring this construct: "Do you think that there will always be people whom you'll be able to count on in the future?"). How one thinks about the future may reflect a cognitive schema (or internal working model) of unpredictability, which may relate to attachment processes (Ross & Hill, 2002). As will be presented in Section 1.3.2, several lines of research demonstrate that attachment insecurity in both childhood and adulthood are associated with higher levels of externalizing behaviours characteristic of the Young Male Syndrome (i.e., aggression, risk-taking, impulsivity). Given that there exists evidence of developmental continuity (from childhood to adulthood) of externalizing behaviours such as physical aggression in males (e.g., Temcheff et al., 2008; Broidy et al., 2003), it is conceivable that levels of attachment insecurity in childhood could predict features of the Young Male Syndrome in adulthood. According to Chisholm (1999), attachment insecurity in childhood may organize evocative life history traits in adulthood as a way to cope with the lack of "protection" received from caregivers early in life.

In contrast, insecure women are suggested to display the "Young Female Syndrome" in similar environments, resulting in a strategy characterized by impulsive mate choice, early and frequent childbearing, and single motherhood. Thus in Chisholm's model, attachment sets the course for adult reproductive choices, but unlike Belsky's (1997) proposition, it is not further implicated once the individual reaches reproductive maturity, instead the emphasis is placed on sex-specific behavioural presentations.

1.2.3. Del Giudice Model

More recently, Del Giudice (2009) has proposed an updated life history model of attachment and reproductive strategies. Similar to the previous models outlined, Del Giudice has hypothesized that in childhood, attachment security is used as a "socioassay" of the current (and expected) local ecology. Attachment patterns in infancy and early childhood are predicted by parenting and caregiver behaviour. At this level, Del Giudice only makes a distinction between secure *vs.* insecure attachment, such that secure attachment is the outcome of sensitive and responsive parenting towards the child, while insecure attachment is the outcome of a risky environment characterized by insensitive parenting and the child's inference that he/she should expect to receive low parental investment in the future.

The early environmental experience of the child should then affect the timing of the child's transition to juvenility, which according to Del Giudice is marked by adrenarche. Beginning at approximately 6 years of age in both males and females, adrenarche represents the onset of androgen production by the adrenal glands and marks the start of the developmental phase known as adrenal puberty (Auchus, 2011; Auchus & Rainey, 2004). Adrenarche does not involve the secretion of testosterone from the gonads, which begins later, at puberty. Instead adrenal precursors of testosterone and estrogen, specifically dehydroepiandrosterone (DHEA), dehydroepiandrosterone sulphate (DHEAS), and androstenedione (A4) are the products secreted by the adrenal glands at this time (Ducharme et al., 1976; Smith et al., 1975). Del Giudice's emphasis on adrenarche instead of puberty, as explored by Belsky et al. (1991) and others (e.g., Miller & Pasta, 2000), is driven by the fact that middle childhood (i.e., ages 7-12) represents a period of reorganization for the attachment system. Specifically, a sex difference in patterns of insecure attachment emerges at this time. Data show that insecure boys are more often classified as avoidant than anxious-ambivalent and vice versa for girls (e.g. Finnegan, Hodges, & Perry, 1996; Granot & Mayseless, 2001; Karavasilis, Doyle, & Markiewicz, 2003; Del Giudice, 2008). In contrast, studies of children as old as six years typically find comparable portions of avoidant and anxious children in both sexes (e.g., van Ijzendoorn et al., 2000; Moss et al., 1998). There is evidence that early

environmental stress stemming from lack of parental supportiveness, including levels of parental negativity and restrictiveness of parenting style, can, in fact, accelerate the relative timing of adrenarche in both boys and girls (Ellis & Essex, 2007), which then sets the stage for the development of sex-specific insecure attachment styles. In contrast, sensitive parenting and the development of a secure attachment in early childhood will lead to normal timing of adrenarche and the maintenance of attachment security in middle childhood.

Del Giudice hypothesizes that attachment patterns in middle childhood (secure vs. avoidant/ambivalent) will predict attachment-dependent reproductive strategies in adulthood. Insecure men are more likely to display avoidant adult romantic attachment and reproductive strategies that are in line with this attachment pattern: short-term, uncommitted mating relationships and a focus on opportunistic sexual behaviour. Insecure women, according to Del Giudice, should adopt anxious, investment-eliciting strategies when environmental risk is moderate (i.e., when environmental cues present in childhood suggested that parents/caregivers were unable to invest in the individual), while avoidant strategies should be selected when faced with greater environmental risk (i.e., when environmental cues suggested that parents/caregivers were unwilling to invest in the individual, higher mortality rate, less resource availability). Finally, secure individuals of both sexes should continue to display security in attachment relationships in adulthood and demonstrate parenting-oriented reproductive strategies focused on future reproduction and high commitment in intimate relationships. Del Giudice does, however, note that in addition to early psychosocial experiences, other environmental cues (e.g., age, important experiences later in life) and/or biological factors (e.g., genetic predisposition, hormonal factors) can also influence reproductive strategies later in development. Thus plasticity may not be confined to childhood, but in fact may extend into adulthood.

1.2.4. Miller and Pasta Model

Although not life history theorists *per se*, Miller and Pasta (2000) have also proposed a model associating early family environmental experience with somatic

development and later reproductive strategies and contraceptive behaviour. Miller and Pasta's model does, however, diverge in one notable way from the other three models outlined above—the authors do not explicitly implicate attachment patterns in their framework. Instead they postulate that all humans possess four types of neural-based bonding systems dedicated to the forms of social bonding that occur during the reproductive life course. They outline the *succorant* system (bonding the infant and child to parents), the *affiliative* system (bonding the child, adolescent, and adult to the sibling and peer), the *sexual* system (bonding the adolescent and adult to the child). The authors hypothesize that individual differences in approaches to social bonding across the lifespan are a function of both one's biological predisposition (i.e., the neural circuitry that has been laid down, which is suggested to be influenced by gonadal hormones and a number of specific neurotransmitters), as well as the learned affective and security cues, feelings of warmth or affection and feelings of insecurity or fear, one experiences within the local environment. The result is the formation of individually adaptive social bonds.

Miller and Pasta suggest that affection- and security-related experiences within the early family environment (i.e., levels of parental affection and support, types of abuse, parental absence, economic factors including father's occupation) should predict somatic development via its impact on the bonding suprasystem. Thus, the child's perception of affection and security within the family environment should lead to the development of corresponding succorant behaviour (which may be akin to individual differences in childhood attachment proposed by the other models), which in turn will influence pubertal timing and the subsequent development of more adult forms of bonding (e.g., sexual and affiliative bonding). The authors speculate that when affection is high and fear is low, puberty is relatively delayed and a reproductive strategy based on high parental investment is preferred. In contrast, when affection is low and fear is high, puberty is relatively accelerated and the subsequent reproductive strategy is one where childbearing begins as soon as possible, while if affection is low and fear is low, puberty is accelerated and the strategy is based on early and frequent mating but without the added push for childbearing. The authors present empirical data that show that retrospective self-reports of greater family affection predict later age at menarche. Thus it is possible that the low affection-high fear configuration they discuss may be more representative of females than males, similar to how Del Giudice (2009) has conceptualized anxious romantic attachment-based reproductive strategies in women (i.e., impulsive mating with an associated heightened desire for long-term relationships, intimacy, and romance). In men, anxious romantic attachment may instead be associated with a more cautious sexual approach (e.g., Gentzler & Kerns, 2004; Kirkpatrick & Davis, 1994; Schmitt, 2005).

To round out their model, the authors speculate about the mechanism by which early environmental experience influences pubertal timing. They implicate the hypothalamic-pituitary-gonadal (HPG) axis, suggesting that perception of the amount of parental nurturance in the home perhaps affects a genetic switch that alters levels of circulating gonadal hormones. Changes in the hormonal environment should then lead to the advancement or retardation of the onset of puberty, including its physical and behavioural correlates.

1.3. Behavioural Correlates of Attachment Style

1.3.1. Sexual/Reproductive Behaviours

Empirical studies have supported the behavioural profile associated with the reproductive strategies predicted by secure *vs.* insecure attachment. Secure attachment, as judged by both categorical (e.g., Hazan & Shaver (1987) measure) and continuous self-report measures of attachment style, has been related to stable, investment-oriented romantic relationships (e.g., Hazan & Shaver, 1987; Simpson, 1990; Kirkpatrick & Davis, 1994). In contrast, a large body of research (see Section 1.2.1) has demonstrated that insecure attachment patterns, and avoidance more particularly, are correlated with adult romantic relationship behaviours consistent with a short-term mating orientation, such as more unrestricted sociosexuality, increased promiscuity, greater acceptance of casual sex, decreased likelihood of starting a committed relationship, and permissive attitudes towards infidelity (e.g., Simpson, 1990; Brennan & Shaver, 1991;1995; Feeney

et al., 1993; Kirkpatrick & Hazan, 1994; Cooper et al., 1998; Paul et al., 2000; Gentzler & Kerns, 2004; Schmitt, 2005; Jackson & Kirkpatrick, 2007, Schindler, Fagundes, & Murdock, 2010; DeWall et al., 2011). Furthermore, university students who described an avoidant romantic attachment style were not only more likely to experience a relationship breakup, but also felt significantly less upset and more relief by this outcome compared to secure or anxiously attached individuals (Feeney & Noller, 1992). Similarly, Simpson (1990) found that university-aged men who self-reported higher levels of avoidance on a Likert-type rating scale derived from Hazan and Shaver's (1987) measure showed significantly less emotional distress following the dissolution of a romantic relationship. Level of emotional distress was not found to relate to the secure or anxious attachment styles in men, or across all attachment classifications in women. Sex-specific associations have also been observed by Gentzler and Kerns (2004) who found that higher ECR-rated anxiety in men, but not in women, was significantly correlated with fewer lifetime sexual partners. In addition, Bogaert and Sadava (2002) demonstrated that attachment anxiety correlated with earlier age of first intercourse and more lifetime partners in women only. These findings are consistent with Del Giudice's (2009) theory, which suggests that opportunistic mating in women can be linked with anxious strategies but is not for men.

Some studies do not, however, corroborate the general pattern of results presented above. Within a sample of stable, heterosexual couples (married or cohabitating for at least six months) Brassard et al. (2007) found that higher levels of attachment avoidance in men, as assessed by the ECR, were associated with greater self-reported avoidance of sex. Two studies conducted in adolescents found that avoidant individuals were less likely than those who were securely or anxiously attached to have ever had sexual intercourse (Cooper et al., 1998; Tracy et al., 2003).

These contrasting findings highlight the apparent paradox between the discomfort with intimacy characteristic of avoidant attachment and the opportunistic reproductive approach predicted by this same attachment pattern. These divergent aspects of the avoidant style are not, however, necessarily incompatible when the motives behind engagement in sexual behaviour are taken into consideration. Schachner and

Shaver (2004) demonstrated in a sample of university students that attachment avoidance, as rated by the ECR, was significantly and negatively correlated with sexual motives related to intimacy and expressing emotional value for one's partner. Avoidance was also positively associated with having sex for non-romantic goals such as to avoid long-term relationships and to increase one's status among peers. This could help to reconcile the findings of Brassard and colleagues (2007), who found that attachment avoidance and interest in sexual contact was *negatively* correlated in men, but within the context of being in committed, stable relationships, where not having sex may be reflective of the goal of avoiding intimacy. In contrast, Schachner and Shaver (2004) found that attachment anxiety was significantly and positively correlated with having sex to reduce feelings of insecurity and to feel valued by one's partner. Similar results were found by Davis, Shaver, and Vernon (2004) as well as Tracy and colleagues (2003) in a sample of adolescents, where the anxious pattern of sexual behaviour was especially evident among girls. These findings are consistent with the theory set out by Belsky (1997) outlining differing sexual/reproductive goals for avoidant *vs.* anxious strategists.

1.3.2. Externalizing Behaviours

Another body of research has explored associations between attachment style and a range of externalizing behaviours. Studies conducted in children have found that avoidant attachment is related to higher levels of aggression with peers (e.g., Card & Hodges, 2003; Erickson, Sroufe, & Egeland, 1985; Finnegan, Hodges, & Perry, 1996; Granot & Mayseless, 2001; Renken et al., 1989), as well as diagnoses of conduct disorder (Rosenstein & Horowitz, 1996). Anxious-ambivalent attachment has, instead, been found to predict withdrawal and passive behaviour with peers (e.g., Card & Hodges, 2003; Finnegan et al., 1996; Cassidy & Berlin, 1994; Erickson et al., 1985), as well as increased susceptibility to affective disorders (Rosenstein & Horowitz, 1996). In addition, avoidant children tend to more frequently take on the role of bullies, possibly to assert dominance over their peers (Troy & Sroufe, 1987). Consistent with the theoretical predictions of Del Giudice (2009), the results from some of these studies point to sexspecific associations. Renken et al. (1989) demonstrated that avoidance was significantly correlated with higher levels of physically aggressive behaviour in boys, but not in girls. Furthermore, Finnegan and colleagues (1996) found that avoidant coping was not only correlated with more externalizing problems (i.e., physical and verbal aggression towards peers, argumentativeness, disruptive behaviour, dishonesty, and dominance over peers), but was also correlated with fewer internalizing problems (i.e., withdrawal, anxiety/depression) but only in boys; preoccupied (anxious) coping was instead significantly associated with higher internalizing problems in males but not in females.

The Young Male Syndrome, as described in Chisholm's (1999) model, refers to the behaviours of aggression, risk-taking, and impulsivity that characterize the alternative reproductive strategy in adult men. Compared to studies conducted in children, in adults the relationship between attachment style and overt behaviours produces a somewhat less clear pattern of results. The general consensus is that attachment insecurity is related to aggressive/abusive behaviours. Most such investigations have been conducted in the context of intimate relationships and/or with men who have a history of violent behaviour. Across these studies, physical aggression perpetrated on female relationship partners has been significantly associated with an avoidant attachment style (e.g., Holtzworth-Munroe, Stuart, & Hutchinson, 1997; Lafontaine & Lussier, 2005; Lawson, 2008; Waltz et al., 2000), anxious attachment (e.g., Bookwala & Zdaniuk, 1998; Fournier, Brassard, & Shaver, 2011; Dutton et al., 1994; Holtzworth-Munroe et al., 1997; Roberts & Noller, 1998), as well as fearful attachment (e.g., Dutton et al., 1994; Mikulincer & Shaver, 2007). It is important to note, however, that because physically aggressive behaviours expressed in adulthood are socially undesirable and potentially illegal, they may be more difficult to study in the general population compared to in children where physical aggression does not yet take a deviant status. This might help to partly explain why relationships are less clear when studied in adult men.

The apparent indiscriminate nature of these results could also be reconciled by the fact that avoidance and anxiety predict physically aggressive behaviours in different contexts. More specifically, dismissing-avoidant attachment in men predicts physical aggression that is motivated by instrumentality (i.e., the deliberate use of violence as an instrument for social influence or to assert authority or control over others), while attachment anxiety (as well as fearful-avoidance) suggest physically aggressive/abusive

behaviours motivated by impulsivity or affective difficulties (Tweed & Dutton, 1998; Babcock et al., 2000). This interpretation of the relationship between avoidance and aggression is consistent with reports demonstrating a significant, positive correlation between men's avoidance scores and their self-reported levels of interpersonal dominance (e.g., Mauricio & Gormley, 2001; Gormley & Lopez, 2003; Hawley, Shorey, & Alderman, 2009; Gormley & Lopez, 2010). The study by Gormley and Lopez (2010) not only found a positive relationship between levels of avoidance and dominance, but also found that attachment anxiety negatively predicted men's dominance scores.

Insecure attachment patterns have also been related to higher levels of verbal aggression and an associated externalizing behaviour, expression of anger. Lafontaine & Lussier (2005) found within a community sample of heterosexual couples that men's avoidance of intimacy (as judged by scores on the ECR) was positively correlated with psychologically aggressive behaviour including insulting and swearing at one's partner. Similarly, Lawson (2008) found that comfort with closeness in relationships (a dimension of attachment avoidance) was negatively correlated with partner reported levels of men's verbal aggression, as rated by the psychological aggression subscale of the Conflict Tactics Scale. In contrast, investigations by Dutton et al. (1994) and Fournier et al. (2011) found that attachment anxiety, not avoidance, was positively correlated with male perpetration of verbal abuse on female partners. With respect to anger, some studies demonstrate that attachment avoidance is associated with greater anger expression (Hudson & Ward, 1997; Troisi & D'Argenio, 2004) and less suppression or inhibition of anger (Hudson & Ward, 1997; cf., Mikulincer, 1998; Creasey & Hesson-McInnis, 2001).

The literature on attachment and risk-taking, another component of the alternative male reproductive strategy proposed by Chisholm (1999), has explored both sexual and non-sexual risky behaviours. These studies have utilized both self-report and behavioural measures of risk-taking and again point to a dichotomy between secure *vs.* insecure patterns of adult romantic attachment; security predicts less engagement in risky behaviours, while insecurity predicts greater involvement. Attachment avoidance has been positively related to high risk drinking behaviour such as heavy alcohol consumption and intoxication, primarily assessed through self-report measures, in a

number of studies conducted in adolescents and adults (e.g., Tracy et al., 2003; Senchak & Leonard, 1992; Laberge, Lussier, & Godbout, 2012; Brennan & Shaver, 1995; Doumas, Turrisi, & Wright, 2006; cf., Letcher & Slesnick, 2014). Furthermore, in a nationally representative sample of approximately 8000 men and women, Mickelson and colleagues (1997) found that interview-rated alcohol abuse and drug abuse were significantly and positively correlated with avoidant attachment scores but not with anxious attachment scores. A significant negative correlation was found between secure attachment and these same variables. It should be noted, however, that given the tendency for people to underreport socially undesirable behaviour including risky lifestyle behaviours (e.g., heavy drinking, drug use), the results of studies employing self-report measures may need to be interpreted cautiously. A study by Schindler et al. (2005) unexpectedly found a significant negative correlation between severity of drug use (reflecting frequency and quantity of substance use, danger of the substances, and methods of consumption among other factors, and validated by urinalysis) and dismissing-avoidant attachment scores in a sample of substance-dependent adults. It is possible that within a more "disordered" sample the relationship between attachment avoidance and drug-taking behaviour may be different than in the general population.

Some studies have also found that dismissing-avoidant attachment is positively correlated with risky sexual behaviours including less reliable contraceptive use among university students (Zlokovich & Snell, 1997) and college men's engagement in coercive sexual behaviour directed at female partners (Smallbone & Dadds, 2000; 2001). Other studies have not, however, supported this association, instead finding that attachment anxiety, not avoidance, was significantly associated with less reliable condom use (Strachman & Impett, 2009; Feeney et al., 1999) and more general sexual risk taking in men (Laberge et al., 2012).

Research exploring the association between yet another dimension of Chisholm's theory, impulsivity, and attachment styles is rather sparse. One study conducted in a nonclinical sample of young adults found that both attachment avoidance and anxiety positively related to levels of impulsivity assessed by the revised NEO personality inventory, but that associations were stronger with anxiety (Scott, Levy, & Pincus, 2009). In contrast, Johnston (1999) failed to find any association between attachment style (secure, anxious, or avoidant) and an 8-item measure of impulsivity in a largely male sample. This study was, however, not limited to a young adult population, instead subjects ranged in age from 19 to 72 years, which could have moderated the findings. Finally, in a study of children, poorer quality attachment to mother (assessed by a crude self-report measure) significantly predicted higher levels of impulsivity in boys, but not in girls (Chapple & Johnson, 2007).

1.3.3. Internalizing Behaviours

Consistent with studies conducted in the area of attachment and externalizing behaviours, secure attachment has been associated empirically with positive psychological adjustment, including lower levels of depression and anxiety (e.g., Marganska, Gallagher, & Miranda, 2013). In contrast, insecure adult attachment has been linked to both depression and anxiety symptoms. Some studies have found that both anxious and avoidant attachment are concurrently related to depressive symptoms (e.g., Mickelson et al., 1997; Hankin, Kassel, & Abela, 2005), however, closer inspection of the literature suggests that anxious attachment may be more strongly associated to depression than is avoidant attachment (e.g., Marganska et al., 2013; Murphy & Bates, 1997; Riggs & Kaminski, 2010; Wei et al., 2005). Simonelli, Ray, and Pincus (2004) and Williams and Riskind (2004) found that individuals who self-reported secure and dismissive (avoidant) romantic attachment styles reported significantly fewer depressive symptoms, as assessed by the Beck Depression Inventory, than those with preoccupied or fearful styles.

A similar pattern of results has been found in studies investigating the relationship between romantic attachment style and symptoms of anxiety. Although some studies find both anxious and avoidant attachment to be positively associated with anxiety symptoms (e.g., Mickelson et al., 1997; Cooper et al., 2009), overall, anxious not avoidant attachment scores, as measured by self-report questionnaires of romantic attachment, have been found to better predict anxiety symptoms (e.g., Hankin et al., 2005; Marganska et al., 2013). Furthermore, two studies have shown that university students classified on the basis of their self-report as dismissing-avoidant score significantly lower on self-report measures of state and trait anxiety, as well as pathological worry compared to preoccupied or fearful types (Simonelli et al., 2004; Williams & Riskind, 2004). In children, internalizing behaviours have also more strongly been related to anxious-ambivalent attachment, and negatively related to avoidant attachment in some studies (e.g., Finnegan et al., 1996; please see beginning of Section 1.3.2. for review of these findings).

Self-esteem has been consistently found to be higher in securely attached individuals than insecurely attached individuals (e.g., Collins & Read, 1990; Feeney & Noller, 1990). That being said, the fact that the insecure-avoidant pattern has been empirically associated with positive working models of the self and negative working models of others (Bartholomew, 1990), has led to studies finding that avoidance is positively correlated with levels of self-esteem in both children (e.g., Cassidy, 1988) and adults (e.g., Bartholomew, 1994; cf., Mickelson et al., 1997; Roberts, Gotlib, & Kassel, 1996). Consistent with the reproductive profile hypothesized by life history theorists to be associated with avoidant attachment, Schmitt (2005) demonstrated in a large crosscultural study that possessing a short-term mating orientation was significantly associated with higher levels of self-esteem, and that this relationship was specific to men and not women.

1.4. The Biological Basis of Attachment

Although attachment is generally conceptualized in the literature as a learned construct, the fact that it is viewed as an innate human tendency (Bowlby 1969/1982), with a confirmed evolutionary significance, suggests that biological substrates should also contribute to the development of attachment bonds and individual differences in attachment style. Behavioural genetic studies have demonstrated significant genetic contributions to attachment classifications in children (e.g., Finkel & Matheny, 2000; Minnis et al., 2007) and in adults (e.g. Brussoni et al., 2000; Torgersen et al., 2007; Donnellan et al., 2008). In addition, molecular genetic studies have identified specific

candidate genes associated with attachment disorganization in infancy and childhood (e.g., Lakatos et al., 2000; Gervai et al., 2005; Spangler et al., 2009) and with romantic attachment avoidance and anxiety in adulthood (Gillath et al., 2008). Across these studies, polymorphisms in dopaminergic and serotonergic genes have been overwhelmingly implicated. Mechanistically, empirical evidence supports the idea that genetic variation influences susceptibility to environmental experiences, producing a socalled biological sensitivity to context, leading to individual differences in attachment style (e.g., van IJzendoorn & Bakermans-Kranenburg, 2006; Gervai et al., 2007; Reiner & Spangler, 2010; Salo et al., 2011).

The life history models outlined above similarly speculate that both social experience and biological factors contribute to the development of varying reproductive strategies. Belsky et al. (1991) and Miller and Pasta (2000) hypothesize that pubertal maturation represents a bridge between early environmental experience and later relationship behaviour. Puberty is a time of gross hormonal change, whose onset is marked by the dramatic increase in the production of steroid hormones by the gonads via the hypothalamic-pituitary-gonadal (HPG) axis (Swerdloff & Odell, 1975). For males, androgen concentrations rise significantly, with a greater than 20-fold increase in the levels of the androgen testosterone over the course of puberty, while females show only a two-fold increase (Elmlinger et al., 2005, Forest, 1979). In contrast, girls show a substantial increase in estradiol levels compared to boys across the same time frame (Sizonenko, 1978), as a result of increased ovarian production of this major form of estrogen. Accelerated pubertal timing is hypothesized by these theorists to be associated with more opportunistic, mating-oriented reproductive approaches. Earlier transitioning into puberty necessitates that the sex-specific hormonal increases occur earlier compared to same-aged peers. The resulting elevation differences appear to persist into young adulthood. Gesquiere and colleagues (2005) have demonstrated in non-human primates that earlier maturing males consistently had higher testosterone concentrations compared to late maturers at every assessment time point within their study (puberty through to young adulthood). Thus, the sex hormonal correlates of early vs. late maturation may
play a central role in men's reproductive behaviour profiles, as predicted by these theories.

Del Giudice (2009) has broadly speculated that *prenatal* exposure to sex hormones, which organize early brain development, could contribute to sex-specific differences in attachment behaviours. Although the exact mechanism by which hormonal exposure during the pre- and perinatal developmental phases can lead to diverging life history trajectories is unclear, individual differences in sex hormone exposure during early development may function similar to genetic variation described above, leading to a biological sensitivity to context and phenotypic differences in attachment and reproductive strategies. Maner and colleagues (2014) have suggested that the masculinizing effects of prenatal androgens in the central nervous system, in concert with environmental instability, are consistent with the initiation of a faster (i.e., more opportunistic) life history strategy both within and across the sexes. Del Giudice (2009) also speculates about the possibility that sex hormone exposure in *adulthood* may represent a "switch" activating sex-specific and genotype-dependent attachment-related behaviours, including reproductive strategies. Similarly, Chisholm (1999) has explicitly suggested that the Young Male Syndrome, his interpretation of the alternative male reproductive strategy, is caused, in part, by testosterone, which he views as a "drug" for engaging in risky activities that maximize current reproduction.

To date, in those few studies where hormones have been investigated in the attachment literature, the majority of the research has focused on the adrenocortical hormone, cortisol, not testosterone. Theoretical interest in cortisol revolves around its established role in the stress response; cortisol is secreted from the adrenal cortex in response to the presence of either physical or psychological stressors. It has been suggested that childhood exposure to non-optimal caregiving and associated attachment adversity impairs the basic functioning of the hypothalamic-pituitary-adrenal (HPA) axis, the human body's primary stress response system (e.g., Repetti, Taylor, & Seeman, 2002; Wismer Fries, Shirtcliff, & Pollak, 2008). Consequently, basal levels of cortisol may systematically differ across the different attachment styles. Studies in adolescents and adults, of which the majority of the work has been conducted in women, have discovered

significant associations between attachment anxiety and basal cortisol. Gordon et al. (2008) and Jaremka et al. (2013) have found that anxiously attached individuals have higher resting cortisol than those who are less anxiously attached, and that levels of attachment avoidance were generally unrelated to daily cortisol levels. Anxiously attached individuals have also been found to show significantly heightened cortisol levels upon awakening compared to those who are securely attached (Oskis et al., 2011), but show a more attenuated cortisol awakening response (i.e., the natural rise to peak cortisol levels that occurs 30-40 minutes after awakening) (Quirin et al., 2008; Oskis et al., 2011). In the only study conducted exclusively in men to date, Rifkin-Graboi (2008) found that attachment insecurity was positively associated with mid-afternoon cortisol levels and attachment preoccupation (anxiety) was positively correlated with later day (17h30) and night-time (21h00) basal cortisol values, but no significant associations were found with waking cortisol.

Further support, however, for the role of sex hormones in the attachment system stems from the literature exploring sex differences in attachment. A sex difference in insecure attachment emerges in middle childhood, where boys are more often avoidant than anxious-ambivalent and vice versa for girls (e.g. Finnegan et al., 1996; Granot & Mayseless, 2001; Karavasilis et al., 2003; Del Giudice, 2008, Chen & Chang, 2012). This pattern appears to continue, at least into young adulthood. Meta-analytic study has demonstrated that men show higher romantic attachment avoidance and lower romantic attachment anxiety compared to women across a number of geographical regions (Del Giudice, 2011). Furthermore, the degree to which men and women endorse avoidant/dismissive statements is also sexually differentiated; on average men's romantic attachment avoidance scores, assessed by either interview or self-report measures, are significantly higher than women's (e.g., Bartholomew & Horowitz, 1991; Scharfe & Bartholomew, 1994; Schmitt et al., 2003; Brassard et al., 2007, cf. Gentzler & Kerns, 2004). Some evidence suggests that the sex difference in the avoidant (Schmitt et al., 2003) and anxious dimensions of attachment (Picardi et al., 2002; see also Chopik, Edelstein, & Fraley, 2013) is largest during the reproductive years (approximately ages 18-35).

Any theory of the biological basis of attachment must be able to account for the sex differences that are consistently observed in romantic attachment and attachment-related behaviours. Although psychosocial or learned differences between the sexes are a possible source of many gender differences, the universality of sex differences in attachment across different cultures (e.g., Schmitt et al., 2003) suggests that biology could also play a role. A proximate mechanism underlying the sex difference could possibly be differential levels of sex hormones, including androgens. That being said, it is possible that large cultural differences governing the acceptability of expressing sexual/reproductive behaviours are also important. The sexual/romantic attitudes and mores associated with certain cultures, such as East Asian cultures, could obscure the role of biological factors on romantic attachment and related sexual behaviour, and thus may need to be taken into consideration. For example, in the aforementioned cross-cultural study by Schmitt et al. (2003), the authors found that sex differences in romantic attachment were smaller in East Asia compared to other regions of the world including America, Europe, the Middle East, and South/Southeast Asia.

I will now describe the basic mechanisms by which androgens exert their influence on the brain (Section 1.5), followed by a review of relevant literature in both humans and non-human animal species that explores the associations between prenatal (Section 1.6) and adult levels of testosterone (Section 1.7) and the expression of attachment and related behaviours.

1.5. Testosterone: Mechanisms of Action

The androgens, which include testosterone (T), are steroid hormones synthesized from cholesterol and secreted by the gonads and the adrenal glands in response to stimulation from the hypothalamus and pituitary (Brown, 1994). T exerts its physiological effects on the brain by binding to androgen receptors (ARs) located in neurons throughout the nervous system. The distribution of neural ARs is quite specific, with receptors found in brain regions critical for learning, memory, and emotion such as the hippocampus and its pyramidal neurons (Beyenburg et al., 2000; Kritzer, 2004), temporal cortex (Puy et al., 1995), prefrontal cortex (Finley & Kritzer, 1999), and amygdala (Abdelgadir et al., 1999). By contrast, ARs are found in lower densities or not at all in other cortical regions (Brown, 1994; Kritzer, 2004). Once bound to AR, T forms a hormone-receptor complex which is internalized into cell nuclei, and in turn binds to androgen-responsive genes at specific binding domains in these genes known as androgen response elements (Choong & Wilson, 1998). Binding can then either activate or suppress gene transcription, which, in turn, can impact protein synthesis and modify cell function (Rubinow & Schmidt, 1996).

The neuronal effects of such binding in the central nervous system occur both during fetal or neonatal development or during adult life (Rubinow & Schmidt, 1996; Swerdloff et al., 1992). Some recent theories suggest androgens could also be active in the pubertal brain (e.g., Schulz, Molenda-Figueira, & Sisk, 2009; Sisk & Zehr, 2005), although this possibility is still speculative. During pre- or perinatal development, androgens have *organizational* effects on brain development in many species. Organizational effects refer to the actions of T during specific defined periods of sensitivity in early brain development that *permanently* alter the structure or functional potential of the nervous system (Phoenix et al., 1959; Breedlove & Hampson, 2002; Wallen, 2009). In humans, gestational weeks 8-24 are believed to constitute such a sensitive period, as the fetal testes transiently secrete high levels of T during this interval (Collaer & Hines, 1995). In other species, AR binding during the "sensitive" window has been found to impact the structure of the brain, including the size of brain nuclei, synapse formation, and axonal and dendritic branching (Gorski et al., 1978; Williams & Meck, 1991; Breedlove, 1992), with different characteristics influenced at somewhat different times during the critical period in development (Smith & Hines, 2000).

Activational effects of T, in contrast, refer to *reversible* changes in brain function brought about by current levels of circulating T in the post-pubertal or adult organism acting upon neural substrates that may or may not also be organized early in life (Eckel et al., 2008). Compared to organizational effects, activational effects occur later in the lifespan, during or after adolescence (Phoenix et al., 1959). In adulthood, AR binding has been found to have neuroprotective effects against oxidative stress (Ahlborn, Prins, & Ceccatelli, 2001) and excitotoxicity in hippocampal neurons (Pouliot, Handa, & Beck, 1996), as well as neuroregulatory effects on neurotransmitter metabolism (Bitar et al., 1991), neurotransmitter (e.g., serotonin) receptor activity (Mendelson & McEwen, 1990; Kendall, Stancel, & Enna, 1982; Hernandez et al., 1994), and neurotransmitter receptor concentrations (Nakamura, Fujita, & Kawata, 2002; Adler et al., 1999). T levels in adulthood have also been found to influence the functional anatomy of brain regions that include the amygdala, hippocampus, and frontal cortex, as revealed by functional neuroimaging studies (Moffat & Resnick, 2007; Maki et al., 2007).

1.6. Is There an Effect of Prenatal Testosterone on Attachment and Related Behaviours?

LHTs of attachment suggest that attachment patterns developed in early life can have long-term effects, including effects on adult reproductive strategies. Although it is conceivable that learning alone could account for the developmental continuity, it is alternatively possible, as speculated by Del Giudice (2009), that T exposure in the prenatal or early postnatal period could contribute to the organization of a neural predisposition toward specific attachment patterns and their persistence into adulthood.

1.6.1. Prenatal Androgen Exposure and Attachment Behaviours

The association between early androgen exposure and attachment per se has not been formally investigated in humans. Some indirect evidence pertinent to this question comes from a study by Helleday and colleagues (1993) who found that adult women with congenital adrenal hyperplasia (CAH), a condition where the adrenal cortex secretes high levels of androgens during gestation, scored significantly higher than age-matched healthy control women on the "social detachment" subscale of the Karolinska Scales of Personality. Higher scores indicate higher levels of detachment in social interactions and lower levels of attachment towards other people (e.g., "I feel best when I keep people at a certain distance"). It should be noted, however, that CAH is a clinical condition, and some women with CAH experience associated physical abnormalities (e.g., genital anomalies), making it possible that there are other condition-related reasons for their higher levels of social detachment than differences in prenatal T exposure alone.

Additional evidence of a possible association between levels of prenatal testosterone and early attachment stems from research conducted in nonhuman animals. Although evidence from animal studies is necessarily indirect, such studies do allow for T levels to be experimentally manipulated, which is not feasible in humans. Wallen, Maestripieri, and Mann (1995) found that infant male rhesus monkeys randomly assigned to receive neonatal treatment with a GnRH antagonist (i.e., T suppressant) initiated and maintained significantly more time in close proximity to their mothers (similar to female monkeys), compared with males who received the antagonist combined with exogenous replacement T, when assessed at one year of age. Males in the T replacement condition (who had supranormal T levels) spent a significantly smaller percentage of their time in proximity to their mothers than either the suppressed T males or females (but not compared to control males). The authors' reasoned that their findings of an inverse relationship between neonatal T exposure and proximity seeking/maintenance reflect a Tdependent change in the character of the infant's interaction with the mother, rather than changes in maternal response to the infants. In studies conducted in two other species, prenatal T administration has been found to dramatically reduce separation-induced distress vocalizations in the young male quail (Bernroider, Holtzrattner, & Rottner, 1996) or guinea pigs (Panksepp, 1998).

1.6.2. Prenatal Androgen Exposure and Attachment-Related Personality Profile: Evidence from Non-Human Animal Studies

A larger body of work has examined associations between early androgen exposure and a variety of personality variables, including those hypothesized by life history theorists to be related to insecure attachment, and particularly the avoidant style in boys and men. The empirical basis for studying these associations in humans stems from research conducted in non-human mammals.

Nonhuman animal studies have consistently provided evidence that early androgen exposure, occurring during the prenatal or neonatal period, *organizes* aggressive behaviour, and that these influences are expressed later in life as the levels of aggressive behaviour displayed spontaneously in experimental contexts. For example, Epple, Alveario, and St. André (1987) reported that neonatally castrated male saddleback tamarins displayed significantly less aggressive behaviour compared to control males when pair tested against either intact males or females. Similarly, Dixson (1993) found that neonatally castrated male marmosets displayed significantly fewer aggressive behaviours towards intact males (but not females) compared to males castrated in adulthood. In rats, Brain (1979) found that males castrated within 6 days postnatally displayed little aggression when treated with T as adults, but those castrated and treated with neonatal androgen replacement displayed normal levels of aggressive behaviour as adults.

Rodent models have also showcased that prenatal alterations in androgen exposure affect internalizing behaviour displays. Exposure *in utero* to endocrine disruptors with anti-androgenic properties or genetic mutations causing androgen insensitivity has been associated with increased anxiety-like behaviour (e.g., time spent in open arms of the elevated plus maze) (Carbone et al., 2013; Hamson et al., 2014; Zuloaga et al., 2011; cf., Lucion et al., 1996) and increased depression-like behaviour (e.g., increased immobility in the forced swimming test) (Fujimoto, Kubo, & Aou, 2006) in males.

1.6.3. Prenatal Androgen Exposure and Attachment-Related Personality Profile: Evidence from Human Studies

In humans, studying the effects of prenatal androgen exposure on behaviour is difficult because exogenous manipulation of sex hormones is not permissible for ethical reasons. Therefore, experimentation is ruled out. Some studies conducted in special populations have, however, provided preliminary support for an organizational effect of testosterone exposure during early development on human aggressive behaviour. Females with CAH, who are exposed to elevated androgen levels prenatally, have been found to display significantly higher physical aggression than control females (Berenbaum & Resnick, 1997; Pasterski et al., 2007). Another study demonstrated that both boys and girls exposed to androgenizing progestins (another class of sex hormone) *in utero* were significantly more likely to report using physical aggression in conflict

situations than controls (Reinisch, 1981). Cohen-Bendahan et al. (2005) found that girls from opposite-sex twin pairs, who are hypothesized to be exposed to higher levels of prenatal T due to diffusion of small amounts of T from the male to the female fetus *in utero*, showed higher levels of verbal, but not physical, aggression compared to girls from same-sex dizygotic twin pairs. All of these studies implicate testosterone exposure during prenatal development as the critical timepoint for the induction of the effect on aggressive behaviour.

The above-mentioned studies, although revealing, are difficult to conduct and thus have been limited in their investigative scope. In addition, these studies do not provide information on how individual differences in prenatal T exposure in normal populations affect behaviour. Although somewhat controversial, one trait that has been employed as a retrospective proxy measure of individual differences in fetal androgen exposure is the ratio of the lengths of the second to fourth digits of the human hand-the 2D:4D digit ratio (Manning et al., 1998; Manning et al., 2003). The 2D:4D ratio is a sexually dimorphic trait, and men, on average, have a lower ratio of the second to the fourth digit than women, a sex difference originally discovered by 19th century anatomists. Some evidence suggests the sex difference in 2D:4D is determined by actions of T in utero and thus may represent a retrospective marker of the level of androgen exposure during the critical period when the brain becomes sexually differentiated (Manning et al., 1998; Zheng & Cohn, 2011). A smaller ratio is suggestive of greater fetal androgen exposure. In support of this, women with CAH show significantly masculinized digit ratios, similar to the ratios of control men (Brown et al., 2002; Ökten, Kalyoncu, & Yaris, 2002; Rivas et al., 2014; cf., Buck et al., 2003). In addition, Lutchmaya and colleagues (2004) found that higher levels of fetal T, as measured by amniocentesis, were negatively associated with the 2D:4D ratio assessed at 2 years of age, in both sexes. The strongest support for the idea that digit ratio is influenced by androgen exposure stems from a study by Berenbaum and colleagues (2009), which found that individuals with a mutated androgen receptor resulting in complete androgen insensitivity have digit ratios that are feminized; the 2D:4D finger length ratio is higher than those of typical men and similar to those of typical women. Although nearly 500 studies have investigated links between the 2D:4D

ratio and variation in traits believed to be androgen-dependent, the validity of the measure as an accurate reflection of individual differences in prenatal androgen exposure in healthy humans has been questioned (Puts et al., 2004; Berenbaum et al., 2009, Hampson & Sankar, 2012b). Nevertheless, it remains the only practical, albeit indirect, measure available, at present, to estimate individual differences in prenatal T in adulthood.

Several studies have reported a significant correlation between the 2D:4D digit ratio and variables related to sexual behaviours. Smaller 2D:4D ratio, which may reflect greater fetal T exposure, has been associated with higher lifetime number of sexual partners reported by men (Hönekopp, Voracek, & Manning, 2006; cf., Puts et al., 2004; Rahman, Korhonen, & Aslam, 2005), and with stronger short-term relationship orientation in men but not women (Schwarz et al., 2011), preference towards casual sex (Clark, 2004; DeLecce, Polheber, & Matchock, 2014; cf., Puts et al., 2004; Charles & Alexander, 2011, Manning & Fink, 2008), increased levels of courtship behaviours expressed by men during a brief conversation with women (Roney & Maestripieri, 2004), and variables related to reproductive success (e.g., number of offspring, frequency of sex, sex drive) in men but not women (Manning & Fink, 2008).

Consistent with the studies of special populations demonstrating that higher prenatal T exposure is associated with higher levels of aggression in children, research employing finger length ratios as a marker of fetal T exposure has consistently found a negative relationship between 2D:4D and physical aggression (or conduct problems) in children (e.g., Butovskaya et al., 2013; Butovskaya, Burkova, & Mabulla, 2010; Fink et al., 2007; Liu, Portnoy, & Raine, 2012; cf., Williams, Greenhalgh, & Manning, 2003). Most of these studies have observed the negative correlation in boys and not in girls, with one study even demonstrating a positive correlation between 2D:4D and prosocial behaviour in girls only (Fink et al., 2007).

In adults, numerous studies have found that smaller 2D:4D digit ratios are correlated with higher self-reported levels of aggression, including physical aggression, verbal aggression, anger, and hostility (e.g., Bailey & Hurd, 2005a; Hurd, Vaillancourt, &

Dinsdale, 2011; Hampson, Ellis, & Tenk, 2008; Hönekopp, 2011; Shaw et al., 2012; but see Butovskaya et al., 2012; Voracek & Stieger, 2009), as well as physical aggression assessed behaviourally (e.g., McIntyre et al., 2007; Kuepper & Hennig, 2007). Again, the associations appear to predominate in men rather than women, with a recent large meta-analysis concluding that a small, but significant, negative relationship exists between 2D:4D ratios and aggression in men only (Hönekopp & Watson, 2011). A study by Benderlioglu and Nelson (2004) did, however, find a negative correlation between 2D:4D and reactive aggression in women. Reactive aggression refers to anger-like responses (which can be physical) to frustration or provocation. Finding an association between this type of aggression and digit ratios in women specifically is consistent with literature suggesting that impulsive aggression is more commonly seen in individuals with attachment anxiety, which is an attachment style more prevalent among women than men (Del Giudice, 2011). In contrast, Cousins, Fugère, and Franklin (2009) found that low 2D:4D in men, but not women, was associated with higher self-reported levels of intrasexual threats and more threatening and physically aggressive behaviour directed at female partners in order to dominate over others for the purpose of mate guarding. Similar results were found by Maner et al. (2014). These more *instrumental* aggressive tactics have also been found to correlate more strongly with attachment avoidance in men (see Section 1.3.2. for references).

With respect to other externalizing behaviours, low 2D:4D digit ratios (indicative of potentially greater prenatal androgen exposure) have been correlated with higher self-reported levels of dominance (e.g., van der Meij et al., 2012; Manning & Fink, 2008; but see Puts et al., 2004 who reported null findings) and higher levels of perceived male dominance as rated by women (Neave et al., 2003). Furthermore, two studies conducted in non-human primates found significant negative relationships between 2D:4D and dominance rank among females (Howlett, Marshall, & Hughes, 2012; Nelson et al., 2010).

More masculinized 2D:4D ratios (i.e., lower ratios) have also been associated with higher levels of risk-taking assessed primarily in the financial domain (e.g., Brañas-Garza & Rustichini, 2011; Coates, Gurnell, & Rustichini, 2009; Garbarino, Slonim, & Syndor, 2011; cf., Apicella et al., 2008; Sapienza, Zingales, & Maestripieri, 2009). Lower ratios have also been associated with riskier driving in men (Schwerdtfeger, Heims, & Heer, 2010). Among the very few studies in adults employing self-report measures of risk-taking, two have demonstrated significant negative correlations between 2D:4D and global risk taking assessed across several spheres including financial, social, recreational, ethical, and health domains (Hönekopp, 2011; Stenstrom et al., 2011), while a recent investigation by Evans and Hampson (2014) found that the financial component of the Risk Taking Scale from the Jackson Personality Inventory-Revised was inversely correlated with 2D:4D ratio in university men.

Of the limited work conducted on markers of prenatal T exposure and impulsivity, Hanoch, Gummerum, and Rolison (2012) found a negative correlation between 2D:4D ratios and self-reported levels of impulsivity among male offenders but not among male non-offenders, while Wacker, Mueller, and Stemmler (2013) found in a sample of undergraduate men that lower digit ratios correlated with higher scores on an impulsive sensation seeking scale, which the authors argued assesses the impulsivity component of the "Young Male Syndrome".

Studies investigating the relationship between prenatal T and internalizing behaviours in adults have found that higher 2D:4D ratios are related to higher levels of trait anxiety, anxiety sensitivity, and specific phobia related to computer use (Brosnan et al., 2011; Evardone & Alexander, 2009). Furthermore, more feminized 2D:4D ratios have been found to significantly correlate with higher levels of neuroticism, a personality feature considered to be a dimensional precursor to anxiety (Austin et al., 2002; Fink, Manning, & Neave, 2004). A positive association between 2D:4D and self-reported depression in men has been supported Bailey and Hurd (2005b), but studies by Martin, Manning, and Dowrick (1999) and Austin et al. (2002) failed to find any significant relationship. The only study conducted on the topic of digit ratios and self-esteem produced null findings in a sample of young to middle-aged women (Wade, Shanley, & Imm, 2004).

The limited work conducted in children has suggested that boys with anxiety disorders may be exposed to lower prenatal T levels than other boys, as indexed by their higher digit ratios (de Bruin et al., 2006). Similarly, a study by Williams et al. (2003) found a positive correlation between 2D:4D and parent-reported scores on the emotional difficulties subscale of the Strengths and Difficulties Questionnaire in boys aged 2-5 years, but not in girls. The subscale is comprised of five items assessing primarily symptoms of anxiety (e.g., "Many worries, often seems worried") but also depression (e.g., "Often unhappy, down-hearted, or tearful"). Using the same questionnaire in an older youth sample, Fink et al. (2007), however, failed to find any significant association between 2D:4D and emotional difficulties in either boys or girls.

In sum, the overall pattern of results suggests that greater exposure to T early in life may be associated with higher levels of sexual and externalizing behaviours, as well as lower levels of internalizing behaviours, particularly anxiety. Measures believed to index higher prenatal androgen exposure, in general, appear to relate to the sexual and personality profile predicted by life history theories of attachment to be associated with insecure attachment (i.e., shorter-term mating orientation and increased sexual behaviour, as well as the aggression, risk-taking, and impulsivity of the "Young Male Syndrome"), and possibly the avoidant style more specifically in men.

1.7. Is There an Effect of Adult Testosterone on Attachment and Related Behaviours?

The fact that prenatal T exposure might influence attachment-related behaviours does not preclude the possibility that circulating levels of T in adulthood could also influence attachment. Studies of other species have shown that some behaviours organized by exposure to T early in life only become *activated* when the brain is further acted upon by circulating T in adulthood. In line with this proposition, Del Giudice (2009) theorizes that sex hormonal exposure in adulthood may act as a "switch" activating attachment-related reproductive behaviours organized earlier in life (see Section 1.4). Thus it is possible that attachment predispositions may be laid down in the central nervous system (CNS) early in development, but that androgen-related individual

differences in attachment-related behaviours may not be expressed overtly until appropriate hormonal stimulation is present in adulthood. Alternatively, for other behaviours, research has also shown that activational effects can occur completely independently of organizational effects. From this perspective, effects of androgen on attachment-related behaviours may potentially occur only once an individual is at adolescence or beyond when the testes secrete adult-like levels of hormone, and when reproduction and mating become relevant concerns. Belsky et al. (1991) and Miller and Pasta's (2000) hypotheses about the downstream reproductive consequences of pubertal timing are consistent with this interpretation of activational effects. Relatively earlier pubertal maturation in boys, which is associated with higher endogenous T levels at adolescence and in young adulthood (Gesquiere et al., 2005), represents the catalyst for more opportunistic, mating-oriented reproductive behaviour in these models. Similarly, Chisholm (1999) specifically implicates circulating levels of T as a causal factor leading to his conceptualization of the alternative male reproductive strategy, the Young Male Syndrome. Reversible effects of T on neurochemistry are the usual mechanism that underlies activational effects at the behavioural level and could be a biological basis for T's activation of attachment-related behavioural patterns.

1.7.1. Sexual/Relationship Behaviour

Research exploring the link between circulating levels of basal T in adulthood and sexual behaviour in men has generally pointed to a positive association between these two variables. The strongest support for adult T concentrations exerting a causal effect on sexual behaviour in men stems from experimental hormone-manipulation studies conducted in healthy humans. In a double-blind, placebo controlled study, Bagatell et al. (1994) demonstrated that men (ages 20-40) exogenously treated with a gonadotropin releasing-hormone antagonist, which suppresses endogenous T levels, had a significant decrease in frequency of intercourse, kissing and fondling behaviour, as well as sexual fantasies during the treatment period compared to men in a placebo condition. Moreover, men who received both the T antagonist and exogenous, replacement T maintained their sexual function, suggesting that T was the relevant causal factor in maintaining sexual behaviour in normal men. Similarly, in a sample of healthy young men

pharmacologically induced to be in a hypogonadal state and then randomly assigned to receive either placebo (maintaining the low T state) or replacement T (returning T to baseline levels), Schmidt and colleagues (2004) found that men in the hypogonadal condition reported a significant decrease in sexual interest, as measured by daily ratings of symptoms, compared to baseline and compared to men in the T replacement condition. Other studies administering T to supraphysiologic levels in young men have, however, failed to find any significant effect of changes in circulating T on sexual function (e.g., Bhasin et al., 2001; O'Connor, Archer, & Wu, 2004). These patterns of findings suggest that it may be *physiological* levels of T, falling in a biologically acceptable range, that are relevant in maintaining sexual behaviour in normal men.

The vast majority of the work on this topic is, however, correlational in nature, but nonetheless continues to support the possibility of an activational effect of T on male sexual and relationship behaviours, as suggested by the experimental studies. Across a number of observational studies, single men typically have been found to have higher T levels than men in committed, monoamorous relationships (e.g., Booth & Dabbs, 1993; Mazur & Michalek, 1998; Burnham et al., 2003; Gray et al., 2004; van Anders & Watson, 2006). Higher levels of T have also been correlated with greater numbers of sexual partners in younger (Bogaert & Fisher, 1995; Halpern et al., 2002; Peters, Simmons, & Rhodes, 2008) and older men (Pollet et al., 2011). Furthermore, two studies conducted in African communities found that men in polygynous relationships have higher T levels than their monogamously partnered counterparts (Gray et al., 2003; Alvergne, Faurie, & Raymond, 2009). In a North American sample, van Anders, Hamilton, and Watson (2007) similarly found that men who were monoamorously partnered had significantly lower T levels than either single men or men in polyamorous relationships (i.e. non-religious love relationships with more than one partner). Additionally, the authors found that men who believed in a polyamorous lifestyle, but were not currently multipartnered, had significantly higher T levels than monoamorous men, suggesting that T might be linked with a specific type of relationship orientation; specifically a shorter-term, mating-focused orientation.

Consistent with this possibility, van Anders and Goldey (2010) found that not only single men, but also men self-identifying as being in *casual* relationships had significantly higher T levels than men in long-term committed relationships. Moreover, overall interest in more/new partners was a significant, positive predictor of T levels regardless of partnering status. McIntyre and colleagues (2006) found that young adult men in relationships, but who nonetheless remained interested in new sexual encounters (as assessed by their scores on the sociosexual orientation inventory (SOI)), maintained high T levels despite being paired; there was no association between sociosexuality and T among single men. Edelstein, Chopik, and Kean (2011) similarly found that partnered men with higher scores on the desire component of the SOI (i.e., greater desire for uncommitted sexual activity) had T levels that were comparable to those of single men. The association between sociosexuality and T as a function of partnering status in men has not, however, been supported in one other investigation (van Anders et al., 2007). Finally, in a recent study of heterosexual couples, partnered men's T was negatively correlated with self-reported level of investment within their current relationships (Edelstein et al., 2014).

1.7.2. Aggressive Behaviour

A relationship between circulating T in adolescents or adults and human aggression has long been hypothesized, however, evidence in favour of this association from human studies is mixed. In contrast, research conducted in non-human animals more consistently find that T is related to aggressive behaviour in males (e.g., Lumia, Thorner, & McGinnis, 1994; Kalin, 1999).

Among human adults, the consensus from a number of reviews and meta-analytic studies is that there is an inconsistent and weak but positive correlation between endogenous T and both self-report and behavioural measures of aggression (Archer, 1991; Archer, Birring, & Wu, 1998; Albert, Walsh, & Jonik, 1993). This relationship may be stronger among offender/physically violent subsamples than among normal adults (Archer, Graham-Kevan, & Davies, 2005) but has not been supported in other meta-analytic studies (e.g., Archer et al., 1998; Book, Starzyk, & Quinsey, 2001). Two studies

that employed a double-blind, randomized, cross-over design in a sample of healthy adult men found that exogenous administration of T to supraphysiologic levels resulted in significantly more aggressive responding on a behavioural measure of psychological aggression compared to placebo or baseline (Kouri et al., 1995; Pope, Kouri, & Hudson, 2000). Kouri and colleagues (1995) also found that total scores on a widely used traitbased measure of aggression (the Aggression Questionnaire) were significantly higher than baseline following testosterone administration. Closer inspection of these results indicated that this association was primarily attributable to an increase in the *physical aggression* subscale of the questionnaire; all other subscales did not differ from baseline. Other exogenous T administration studies in healthy, male volunteers have, however, produced null findings with a variety of self-report and behavioural measures (e.g., O'Connor et al., 2002, 2004).

One factor that may help to unify the generally disparate findings is a consideration of participants' ages. According to Archer and colleagues (2005), the associations between T and aggression appear to be most pronounced when people are in their twenties and early thirties. Another factor that merits exploration is the possible mediation of the relationship between T and aggression by dominance. A concept related to aggression, dominance is defined as "limitations imposed on the behaviour of another through the implied threat of aggressive acts" (Archer, 1991). A review conducted by Archer (2006) concluded that across 13 studies (see Archer, 2006 for the list of studies) the majority were consistent with a positive correlation between T levels and various measures of dominance, including leadership, toughness, personalized power, and aggressive dominance.

Consistent with this interpretation, experimental study in non-human primates has found that the exogenous administration of T to cynomolgus monkeys caused a significant increase in contact and non-contact aggressive behaviours compared to controls, but only among those monkeys who held a dominant social status (Rejeski et al., 1988). A recent study by Slatcher, Mehta, and Josephs (2011) demonstrated that among young adult men, higher T was positively correlated with the frequency of dominance behaviours displayed during a laboratory intrasexual competition scenario. Furthermore, the strong positive association between T and dominance behaviour was only evident among men high in self-reported dominance, as assessed by the Personality Research Form (PRF). Sellers, Mehl, and Josephs (2007) have also found that higher scores on the dominance subscale of the PRF are correlated with higher T levels in university students (T levels in male and female participants were standardized separately to control for sex differences and correlations were collapsed across sex). Other correlational studies conducted in adult men measuring trait dominance or perceived dominance have failed to produce significant associations with adult T (e.g., Johnson, Burk, & Kirkpatrick, 2007; Neave et al., 2003).

1.7.3. Other Externalizing Behaviours

Although physically aggressive and dominant behaviours as well as more unrestricted sexuality can be conceptualized as "risky behaviour", and is in fact conceptualized this way in much of the risk-taking literature, there do remain other types of behaviours that fall into the risk-taking category and which correlate with circulating T levels in adulthood. In men, higher basal T levels have been associated with engagement in risky *health* behaviours such as cigarette smoking (Booth, Johnson, & Granger, 1999; Knussmann & Christiansen, 1989; Fisher et al., 1997), alcohol use and abuse (La Grange et al., 1995; Mazur & Booth, 2014), and drug use (Booth, Johnson, & Granger, 1999; Mazur & Booth, 2014), as assessed by self-report. Furthermore, within a large-scale study of American Veterans (mean age = 37 years), Booth, Johnson, and Granger (1999) found that high T men (after controlling statistically for age), compared to low T men, had a greater probability of having sexually transmitted diseases and physical injuries. A recent investigation by van Anders and colleagues (2012) found that higher salivary T in young adult men predicted safer sex practices, including higher scores on a safer sex composite measure and safer sex resilience (i.e., more likely to follow through on condom use in the face of barriers to do so). These findings remained even after controlling (via multiple regression) for variables that may affect associations between T and sexual risk-taking such as sexual frequency and relationship status. Although it would appear that these findings suggest that higher T protects against sexual risk-taking, the authors reason that engaging in safer sex practices may paradoxically represent a

bolder choice for men as it might be more socially risky for their relationships or their reputation, in general, to be sexually responsible.

A number of laboratory studies have found that higher levels of basal T are correlated with more risky financial decision-making in men (e.g., Apicella et al., 2008; Evans & Hampson, 2014; Stanton, Liening, & Schultheiss, 2011a; Reavis & Overman, 2001). Using a task involving certain outcomes and risky gambles, Stanton and colleagues (2011b) suggested a U-shaped relationship between T and economic risk preferences. Finally, Goudriaan et al. (2010) found that young men who underwent an experimental manipulation that produced high-normal levels of T and low-normal levels of estradiol (the major type of estrogen found in humans) showed a significant increase in risk-taking, compared to baseline. No such increase was found for men randomized to have low-normal T and high-normal estradiol levels.

Although not conducted in an adult sample, a study by Booth et al. (2003) is worth noting. The authors assessed the associations between T, risk-taking, symptoms of depression, and parent-child relationship quality in a sample of boys and girls aged 6-18 years. Results showed that endogenous T levels were independently related to the quality of relationships that sons formed with their mothers, as rated by child reports of intimacy with the parent and parents' reports of their acceptance of their child, among other factors. In boys only, higher T was related to poorer relationship quality with mothers. In addition, poor parent-son relationship quality moderated the association between T and risk-taking, as assessed by youths' self-reports of the frequency of engagement in 18 risky activities over the past year (e.g., "skip a day of school"; "stay out all night without parents' permission). Given the nature of these questions, however, children between the ages of 6-9 years did not complete the risk-taking measure. T-related risk-taking behaviour was more evident as parent-son relationship quality decreased and was less evident as parent-son relationship quality increased. T was positively correlated with risk-taking only when parent-son relationship quality was rated to be low. This was true regardless of stage of development. These findings support the idea that individual differences in the quality of the relationship a son forms with his parent, which may be

indicative of the nature of the attachment relationship, is associated with T, and that the relationship may have implications for adjustment problems, such as risk-taking.

The literature on adult T and impulsivity has generally considered studies investigating the relationship between T and overt behaviours (e.g., criminality and aggression, sensation seeking) as evidence of a T-impulsivity association. Among the few studies that investigated impulsivity specifically, or have employed measures of trait impulsiveness, the findings are mixed. O'Connor et al. (2002) found that hypogonadal young men reported significantly lower levels of impulsiveness (i.e., a lacking of "futuring" or forethought), as assessed by the Barratt Impulsivity Scale-11 (BIS-11), compared to healthy young men. No significant differences were found for the cognitive or motor components of impulsivity. Similarly, Schmidt and colleagues (2004) found that healthy young men at baseline had significantly higher levels of sensory impulsiveness (but not motor or interpersonal impulsivity), assessed using an earlier version of the BIS, compared to their scores following pharmacologically-induced hypogonadism. Two studies employing a behavioural measure of impulsivity, the delaydiscounting task, have, however, failed to produce consistent associations between T concentrations and impulsivity in young adult men (Ortner et al., 2013; Takahashi et al., 2006). Some work in non-human animals has demonstrated that gonadectomy in adult male rats enhances behavioural inhibition to footshock, indicating improved impulse control (Svensson, Söderpalm, & Engel, 2000), while testosterone treatment induces behavioural disinhibition, reflecting impulsive-like behaviour (Svensson et al., 2003), lending support to an activational effect of T on impulsiveness.

1.7.4. Internalizing Behaviours

Higher levels of T appear to be associated with higher levels of externalizing behaviours, and conversely may also be associated with lower levels of internalizing symptoms. Studies of young men who are hypogonadal have found that self-reported levels of depressive symptoms are significantly higher among the hypogonadal group compared to healthy controls, and that exogenous T administration over a set treatment period results in a significant reduction in depressive symptoms from baseline (Aydogan et al., 2012; O'Connor et al., 2002). Similarly, Schmidt and colleagues (2004) found that pharmacologically-induced hypogonadism (low T levels) precipitated depressive symptoms in a subset (~10%) of healthy young men with no previous history of psychiatric illness, supporting the possibility that reduced T may predict the onset of depressive symptoms. Studies of non-human animals in which T levels have been manipulated in adulthood provide further support for a causal relationship (e.g., Bernardi et al., 1989; Buddenberg et al., 2009; Carrier & Kabbaj, 2012). Assessment of basal T in a sample of undergraduate men also found that T was negatively related to depression, and specifically was associated with self-reported sleep symptoms of depression (Sankar & Hampson, 2012; but see van Honk et al., 1999 who found no significant association between T and depression in healthy young men). In contrast, administration of exogenous T to men without a psychiatric history does not appear to have any effect on self-reported levels of affective symptoms (O'Connor et al., 2004).

Higher T concentrations have also been suggested to have an anxiolytic effect. In young men, a hypogonadal state has been associated with higher levels of anxiety in several studies (e.g., Ayodogan et al., 2012; O'Connor et al., 2002; cf., Schmidt et al., 2004). Studies measuring basal T in healthy males have similarly pointed to a negative relationship between T and self-reported anxiety symptoms (e.g., Berglund et al., 2011; Granger et al., 2003; cf., Maner et al., 2008; van Honk et al., 1999). Again, experimental T-manipulation studies conducted in non-human animals support the possibility of a causal effect of T on anxiety symptom reduction, although the mechanism behind this association is not yet known (e.g., Aikey et al., 2002; Carrier & Kabbaj, 2012; for further references see review by McHenry et al., 2014).

Among the limited work available on the topic of androgenic effects on selfesteem, the data point towards a positive relationship between levels of self-esteem and T concentrations in adulthood. Johnson and colleagues (2007) demonstrated that T levels were a positive predictor of scores on a global self-esteem scale in a sample of undergraduate men. Vermeersch et al. (2010) found similar results, albeit in an adolescent male sample (mean age = 14.4 years). In a placebo-controlled study, a positive effect of exogenous T treatment on self-esteem was found in a group of hypogonadal young men (O'Connor et al., 2002). Significant increases in men's levels of self-esteem were seen between the second and third assessment timepoints in the study (i.e., week 4 and week 8). Consequently, it cannot be ruled out that the observed changes in self-esteem may have been due to associated physical changes secondary to T administration, including concomitant restoration of sexual function. In another study, pharmacologically-induced hypogonadism in healthy men for a period of two months failed to significantly reduce self-esteem scores compared to baseline (Schmidt et al., 2004). Because these men were in a hypogonadal state only transiently, it is possible that the physiological and sexual changes associated with acute hypogonadism may not have been substantive enough to cause negative self-interpretations in participants.

1.7.5. Attachment Behaviours

Finally, some indirect evidence does currently exist of a possible association between circulating T and attachment. Among fathers, those that are more involved in childcare have been found to have significantly lower T levels than those who do not participate in care (e.g., Alvergne et al., 2009; Gettler et al., 2011; Kuzawa et al., 2009; Muller et al., 2009). These findings support the idea that men with lower T are more likely to make a long-term investment in their offspring's care compared to men with higher circulating T, and that interest in forming close, emotional bonds through investment in parenting *vs.* mating effort may be negatively associated with T.

Within a sample of children, Strong and Dabbs (2000) investigated the relationship between children's salivary T levels and their everyday behaviours, including parent-child attachment, broadly. Parents' ratings of children on the "Attached" subscale of the Parenting Stress Index, comprised of two items: "My child wants to be close to his/her mother" and "My child likes to be cuddled or touched", were negatively correlated with T levels in the whole sample, as well as within the prepubertal (3-8 years) and early pubertal (9-12) subgroups separately. Correlations were conducted with boys and girls combined, because at these ages no significant difference in mean T levels between the sexes was evident. As discussed in Section 1.7.3, a study by Booth et al. (2003) demonstrated a significant association between T and parent-child relationship

quality. Among boys, higher T levels were significantly correlated with poorer quality relationships with mothers (i.e., levels of intimacy between child and mother, as well as mother's level of understanding and acceptance toward their child). No such associations were found for girls.

In a recent study examining the interpersonal styles associated with basal T levels in men, Turan and colleagues (2014) hypothesized that high T individuals should show more agency and dominate over others, while also showing lower levels of communion (i.e., unfriendliness, low interpersonal warmth). As part of their assessment of the communal dimension, the authors administered a self-report measure of romantic attachment. Basal T was found to positively correlate with romantic attachment avoidance in their sample, but not with attachment anxiety. The findings from this investigation corroborate the discussion in Section 1.4, providing the first direct evidence of a sex hormonal effect on psychological attachment. The authors did not, however, explore the relationship between adult T and attachment within the broader context of reproductive strategies/orientations, as suggested by life history models.

1.7.6. Summary

Review of the literature appears to suggest that basal levels of T in adult men may be positively correlated with behaviours consistent with greater investment in mating effort (e.g., greater number of sexual partners, more likely to be single) and a shorterterm mating orientation. In keeping with the personality profile predicted by life history theorists to be associated with attachment insecurity, and potentially attachment avoidance in men, some evidence suggests that men's circulating levels of T are positively related to aggression and aggressive dominance, as well as a range of risky behaviours, but are negatively related to behaviours falling into the internalizing category, such as depression and anxiety. Preliminary work in the area of T and attachment is in support of a positive relationship between basal T levels and behaviours more closely aligned with avoidant attachment in men, consolidating the possibility of androgenic influences on psychological attachment in men. The manner in which androgens, including T, influence attachment patterns and related reproductive strategies still remains to be understood, however.

1.8. A Consideration of Androgen Receptor (AR) Functionality

In humans, T exerts its predominant influences on the brain by binding to ARs in various brain structures. However, ARs are known to vary in their level of functionality following T-binding. Differences in AR functionality are mediated in part by genetic factors including the length of a polymorphic polyglutamine stretch, made up of a variable number of trinucleotide (CAG) repeats, in exon 1 of the AR gene, which is located on the X chromosome (Zitzmann, 2009). Normal CAG repeat sequences range between 9 and 35 repeats, with an average of 20 to 22 (Edwards et al., 1992; Krithivas et al., 1999; Hsing et al., 2000). The relative length of the CAG repeat stretch confers differences in transactivational activity (Hsiao et al., 1999; Irvine et al., 2000) and AR expression (Choong et al., 1996; Ding et al., 2004). Short CAG repeat length is associated with increased transcription of androgen-responsive target genes, while long repeat length is associated with lower levels of AR induced gene transcription (see Zitzmann, 2009). In principle, therefore, it is possible that not only differences in circulating T levels, but also individual differences in AR gene CAG length, by influencing the capacity of the CNS to respond to T, may be associated with differential reproductive strategies/orientations in men.

The length of the CAG polymorphism in the AR gene has been associated empirically with conditions that are T-related, including prostate cancer, infertility, malepattern baldness, cardiovascular disease, and bone density loss (see Zitzmann & Nieschlag, 2003). In the case of prostate cancer, for example, where higher T levels influence the development of the disease, reports suggest that short CAG repeats increase the risk of cancer development and/or age of onset of the disease. Although the findings remain equivocal (e.g., Jönsson et al., 2001), shorter CAG repeat length has been related, in some studies, to greater levels of androgen-dependent psychological traits in men, including physically aggressive behaviour (e.g., Cheng et al., 2006; Rajender et al., 2008; cf., Hurd et al., 2011; Butovskaya et al., 2012), self-reported dominance (Simmons & Roney, 2011), and trait impulsiveness (Aluja et al., 2011; Westberg et al., 2009). In contrast, longer CAG repeat length (conferring weaker AR functionality) has been associated with greater symptoms of depression (e.g., Sankar & Hampson, 2012; Härkönen et al., 2003; Schneider et al., 2011) and anxiety (e.g., Schneider et al., 2010; Härkönen et al., 2003; cf., Su et al., 2007; Westberg et al., 2009).

Only two studies, to date, have investigated the relationship between polymorphism in the AR gene and socio-sexual behaviours. Simmons and Roney (2011) demonstrated a statistical trend (p = .063) towards higher scores on the attitudinal component of the Sociosexual Orientation Inventory among men with shorter CAG repeat lengths, while Roney, Simmons, and Lukaszewski (2010) found that following a brief conversation with a woman, men who showed a larger increase in T levels had shorter compared to longer CAG repeat lengths. Based on their findings, the authors speculated that shorter CAG lengths might predict greater physiological and behavioural investments in heterosexual mating. No studies, at present, have looked to see if the CAG repeat polymorphism predicts individual differences in the degree to which men are calibrated towards specific attachment styles, although such a relationship is plausible if androgens do indeed influence attachment orientation in men.

1.9. Model Presentation

The life history theories of attachment and reproduction that were described in Section 1.2 speculate about a series of causal pathways in the development of an adult's attachment patterns towards romantic/sexual partners that have yet to be empirically tested in a sample of young men. Figure 1 illustrates the life history based model that was tested in the present study, and in the discussion that follows I will outline the predictive pathways that were tested.

In the present dissertation, the causal pathways were tested using structural equation modelling (SEM). SEM is a statistical technique that uses a hypothesis testing approach to test the pattern of inter-variable relations within a theory that is specified *a priori*. In this way, SEM can determine whether a hypothesized theoretical model (e.g.,



Figure 1. Integrated life history based model. An integrated model of early environmental experience, childhood attachment, and reproductive strategies in men, and their hypothesized inter-relationships with androgens.

LHT-based model) is consistent with empirical data collected to reflect or test the theory. Another advantage of SEM, that made it an appropriate analytic tool for this dissertation, is that it allows for the study of the relationship among latent constructs that are indicated by multiple measures or indicators (e.g., mating orientation, early developmental environment).

The three LHT models that have been discussed above share certain core elements, but also differ in important respects (see Table 1 for a summary of the various LHTs presented). They hypothesize that a boy's early family environment should be a determinant of the type of attachment pattern he displays in childhood. At this level, Belsky et al. (1991) and Chisholm (1999) conceptualize childhood attachment along the dimension of security *vs.* insecurity, as does Del Giudice (2009) within the context of early childhood. Although Belsky (1997) further discusses the differential (and sexspecific) reproductive consequences of avoidant *vs.* anxious-ambivalent attachment in childhood, as does Del Giudice (2009) from middle childhood onwards, Chisholm (1999) predicts that both avoidant and ambivalent (i.e., insecure) children will adopt life history strategies maximizing short-term reproductive effort, and that the two attachment strategies are simply responses to different safety threats. Thus, as outlined in Figure 1, early family environment was hypothesized to be a latent variable that predicts levels of attachment security/insecurity to parents in childhood.

Figure 1 goes on to show that childhood attachment security was hypothesized to predict men's reproductive strategy (both sexual and as defined by non-sexual evocative traits). This is consistent with Chisholm's (1999) theory which speculates that greater childhood attachment insecurity should lead to the display of the constellation of "Young Male Syndrome" traits, characterized by increased aggression, impulsivity, risk-taking, as well as increased sexuality. Belsky (1997) and Del Giudice (2009) hypothesize that childhood attachment will predict adult romantic attachment-specific reproductive strategies characterized by sexual behaviours such as mating orientation (short- *vs.* long-term), number of sexual partners, and onset of sexual activity (earlier *vs.* later). Of note, because adult romantic attachment style has been conceptualized by some theorists to be *a part of* the sexual reproductive strategy (e.g., Belsky, 1997, Kirkpatrick, 1998; Del

Giudice, Angeleri, Manera, 2009), various measures of romantic attachment style were tentatively included as indicators of the sexual reproductive strategy. That being said, other work has suggested that romantic attachment style is a *correlated trait that precedes* reproductive strategy development (e.g., Del Giudice, 2009; Hill et al., 1994a). Consequently, based on the results, the modelling of romantic attachment was modified accordingly (as Section 3.2.5 will later show).

A more avoidance-based reproductive strategy should hypothetically be associated with reduced commitment and short-term mating, while a more anxiety-based reproductive strategy should be associated with a delayed, indirect mating approach designed to maximize closeness with kin and partners (Belsky, 1997; Del Giudice, 2009), although anxiety in women may involve an opportunistic relational mating style (e.g., Del Giudice, 2009). While not explicitly outlined in the LHTs reviewed, a direct causal pathway between early family relationships and behaviours associated with more opportunistic reproduction (e.g., increased sexuality and hypermasculine behaviour) has been hypothesized in some literature (e.g., Draper & Harpending, 1982) and empirical study has provided some tentative support for this connection (e.g., James et al., 2012; Alvergne, Faurie, & Raymond, 2008). Consequently, the connections between early family environment and the sexual and non-sexual reproductive strategies were also modelled, as illustrated in Figure 1.

The potential effects of androgens, which are implicitly but not explicitly outlined in the life history based models, can theoretically occur at one of two time points based on evidence from the neuroendocrine literature reviewed. One possibility is the prenatal period. As shown in Figure 1, prenatal T levels could directly impact levels of childhood attachment insecurity or alternatively could directly predict the sexual or non-sexual reproductive behavioural profile seen in adulthood (including behaviours such as aggressiveness or impulsiveness, as reviewed above). Significant associations at this level would provide support for the organizational effects of androgens on attachment and related reproductive strategies in men. T's effect within a LHT-based model could alternatively be seen as activational in nature with circulating levels of T in adulthood directly predicting attachment-related sexual and non-sexual reproductive strategies that may or may not have been organized earlier in life (see Figure 1). An effect of adult circulating androgens is thus the second time point where an effect of androgens on attachment bonds or attachment-related behaviours might theoretically be exerted. Recall that the theories of Belsky et al. (1991) and Del Giudice (2009) (as well as Miller & Pasta, 2000) postulate that timing of somatic maturation, secondary to early environmental experience and psychological development, will support the reproductive strategy adopted by men. Given that there is evidence that accelerated pubertal development is associated with having higher endogenous adult T (Gesquiere et al., 2005), it is possible that levels of attachment insecurity in childhood or the early family context could predict basal T levels found in adult men. Thus, these predictive associations were modelled, as shown in Figure 1.

A final androgen-related factor to consider within the model is the role of the CAG repeat polymorphism in the gene that codes for the androgen receptor. Individual variation in the length of this polymorphic stretch can be quantified by using DNA genotyping. As illustrated in Figure 1, measured CAG repeat lengths, which inversely affect androgen receptor *activity* following T-binding, could independently predict sexual and non-sexual reproductive strategies by altering the 'effective levels' of androgen to which the nervous system is exposed. Androgen receptor polymorphism could also be an independent predictor of individual differences in childhood attachment style. Importantly, only characteristics that are influenced through AR binding may be expected to be associated with the AR CAG polymorphism; if T is important in the organizational development of a particular trait but acts through a metabolite of T such as estradiol that does not bind to AR, then we would not expect to find an association. In contrast, the genetic polymorphism itself cannot plausibly cause or be the cause of individual differences in the actual concentration of adult or prenatal T^1 . Similarly, neither early

¹ Although Crabbe et al. (2007) has hypothesized that adult T synthesis may be controlled by feedback mediated by the AR (and thus the CAG genotype) in the hypothalamus and pituitary, most empirical evidence suggests that circulating levels of T are not correlated with CAG polymorphism (e.g., Hampson & Sankar, 2012a; Goutou et al., 2009; Krithivas et al., 1999).

family experience nor childhood attachment can cause individual variability in the length of the CAG repeat sequence.

The following hypotheses were thus proposed: (1) support for Chisholm's (1999) theory would be demonstrated if a test of the model reveals that a less stable early family environment positively predicts childhood attachment insecurity, and attachment insecurity then positively predicts either the non-sexual or sexual reproductive behaviours, or both, (2) support for Belsky et al. (1991) and Del Giudice's (2009) theories would be demonstrated if a less stable early family environment positively predicts childhood attachment insecurity, and in turn attachment insecurity positively predicts the sexual reproductive strategy characterized by more avoidant and less anxious romantic attachment style, (3) the relationship between androgens (prenatal T, adult T, and/or CAG RL) and attachment and related behaviours would be supported based on the model results.

2. Method

2.1. Sample and Recruitment

The final sample size included in all analyses was N =195, out of an original 240 who were enrolled, with participants ranging in age from 18-35 years (M = 21.06, SD = 3.43). All participants were physically healthy heterosexual men recruited from the University of Western Ontario. Participants were recruited through posters placed throughout the university campus, as well as through an introductory psychology participation pool (see Appendix A for a copy of the recruitment poster). Reimbursement was in the form of monetary compensation (\$15 CAD) if recruited through posters, or course credit, if recruited through the participation pool.

One hundred and fifty-five participants (79.5% of the sample) were undergraduate students, 38 (19.5%) were graduate students, and 2 (1%) were staff and/or working fulltime. Although a university sample may be more restricted than the general population on certain variables pertinent to attachment (e.g., SES), university populations are commonly used in studies of romantic attachment (e.g., Edelstein et al., 2010; Gormley & Lopez, 2010; Gillath et al., 2008; Gentzler & Kerns, 2004; Schachner & Shaver, 2004). Thus a student sample was considered to be both acceptable and conventional for this type of research. Table 2 summarizes the major demographic variables for the study sample.

In order to be eligible, participants were required to be between the ages of 18-35 years. This age range was selected because many men between these ages are likely to be looking for partners while others may be in stable relationships, enabling greater variability in relationship patterns to be seen. In support of this, Ranta, Dietrich, and Salmela-Aro (2014) found within a sample of over 1000 "emerging adults" (defined as being between the ages of 20-23 years) a nearly even split between the number of participants who were single (41% of sample at age 20; 34% at age 23) vs. dating or seeing someone (38% at age 20; 27% at age 23). Cohabitation with a partner was also relatively common (20% of sample at age 20) and became more common in the older age group (34% at age 23). Similarly, De Vaus, Qu, and Weston (2003) found that over the course of their 20s, the percentage of men who were unpartnered went from nearly 90% to 57%, and this number continued to drop with increasing age. Over age 35, the proportion of individuals who were married was quite high and stable (60-70%), while levels of unmarried cohabitation or being single were quite low (less than 10% and 30%, respectively). These data again point to greater variability in partnering status during young adulthood compared to middle age. A further reason for limiting the testing to the 18-35 year age range was because of the belief that short-term sexual strategies may be more frequently adopted in young adulthood, and with increasing age there may be a shift away from shorter-term to longer-term strategies (Buss & Schmitt, 1993).

Participants were required to have had at least one romantic or sexual partner, either currently or over the past 1 year. This criterion was selected to ensure that participants were capable of rating their *typical* experience in a romantic or sexual situation. Only heterosexual men were recruited because the LHT predictions of how attachment style may influence reproductive strategies, are specific to opposite-sex relationships. Table 2

Demographics of Study Participants

Variable	Mean (SD)	Range (min-max)	% of sample
Age (N = 195)	21.06 (3.43)	18-35	
Total SES of parental household (based on parental occupation and education; N = 195)	13.24 (2.19)	6.50-16.00	
Financial SES of parental household (based on parental occupation alone; N = 195)	11.93 (3.75)	4.00-18.00	
Education (N = 195)			
First year undergraduate students			43.59
Upper year undergraduate students (U2-U4)			35.89
Graduate students			19.49
Staff/working full time			1.02
Ethnicity (N = 195)			
Caucasian			71.79
South Asian (e.g., Indian, Pakistani, Sri Lankan)			12.31
Other (Black, First Nations, Middle Eastern, Hispanic)			15.90

To avoid the possibility that T levels may be altered as a result of medication use, participants were excluded if their responses to two questions on the demographics questionnaire (see Section 2.2 for description of demographics questionnaire) reflected that they had a medical condition that could influence T metabolism or that they used medications (e.g. antidepressants) that can artificially alter T levels. Participants who reported a history of endocrine pathology were also excluded.

A decision was made to exclude 45 volunteers, who self-identified as East Asian, from the final data analyses (resulting in a reduced sample size of 195 out of an original 240), because of the complexities introduced by significant cultural differences that could have influenced the structural relationship seen between outcome variables of interest (see Section 1.4) and because of linguistic concerns related to English proficiency in this subsample. There is evidence in the literature that young adults of East Asian heritage show differences compared with other ethnic groups on major aspects of sexuality, including more restricted sociosexuality and conservative sexual attitudes, as well as a lower proportion of young adults ever having had intercourse and an older age at first intercourse (e.g., Brotto, Woo, & Ryder, 2007; Meston, Trapnell, & Gorzalka, 1996; for review see Okazaki, 2002), irrespective of the length of time participants in these studies spent in the dominant (i.e., Western) culture. In fact, Asian volunteers in the current study were found to differ significantly on these (and other) variables. Their more conservative sexual mores and behaviours made the Asian subsample difficult to compare with the rest of the sample, and in fact their inclusion altered correlational relationships otherwise seen between biological variables and key behavioural variables. For example, the correlation between Adult T and the primary measure of romantic attachment avoidance (ECR avoidance) was in the opposite direction for the Asian subsample compared to Non-Asians (r = -.16 vs. r = .22, respectively). By contrast, within the next largest non-Caucasian ethnic group in the study, the South Asians (N = 24), the pattern of associations between androgenic variables and key behavioural variables was in keeping with the rest of the Non-Asian subgroup and differed from the East Asians (e.g., r ECR avoidance & Adult T = .43).

There was also a validity issue related to linguistic proficiency. The current study required the completion of a large number of self-report questionnaires, totalling over 300 items. Some of the questionnaires included questions that were semantically complex and thus validity of the responses could only be assured in participants who were adequately fluent in English. Two-thirds (64%) of the Asians indicated that English was not their primary language. Although not in itself indicative of comprehension of questionnaire content, these data coupled with the observation that numerous Asian participants appeared to have difficulty understanding test instructions provided by the experimenter and often took very substantially longer to complete the questionnaires, suggested that their comprehension of the test material could not be assumed.

2.2. Procedure

Participants were scheduled for a single one-hour visit to the laboratory, where they completed a set of confidential paper-and-pencil questionnaires and where biological specimens were collected. Each participant was tested individually in a quiet testing room between 1300 and 1800 hrs, to control for circadian variation in T. Circulating T levels are most steady in the afternoon and early evening (Gupta, Lindemulder, & Sathyan, 2000), a time of day that is thus recommended for studies where individual differences in T are the focus of investigation (Yang et al., 2007). On arrival, DNA sampling was performed as described below, to determine each individual's AR genotype (CAG repeat length). Two saliva specimens were then collected for the measurement of T, one immediately after the DNA sampling and a second approximately 35 minutes later. T concentrations from the two specimens were assayed independently then averaged to yield single mean values. When collected under well-controlled conditions, there is an excellent correlation (r = .85) between a single timepoint measure of T and the mean of seven samples taken over one year (Vermeulen & Verdonck, 1992) or even longer time periods (Mazur & Michalek, 1998). In addition to saliva collection, a digital photocopy of the underside of each participant's right and left hand was taken to measure the 2D:4D digit ratio. Specimen collection and hand measurement are described below.

Participants completed a series of paper-and-pencil self-report questionnaires

(described in detail below), in person. Because of the large number of questionnaires administered, the individual questionnaires were printed on different coloured paper to help maintain participants' attention. Questionnaires were organized into two separate booklets, which were administered independently during the test session (see Appendix B for order of questionnaire presentation). In an effort to maintain perceived confidentiality of responses, following the completion of each questionnaire booklet participants placed the completed booklet into a sealed envelope before returning it to the experimenter.

In addition to the completion of the questionnaire booklets, participants also completed a demographics questionnaire (see Appendix C for copy of questionnaire), which was administered at the start of the test session while participants collected the first of the three saliva samples. The demographics questionnaire inquired about health and environmental factors that can affect T levels (e.g., medications, hormonal conditions, waking time, smoking status, recent alcohol use), hand measurements (i.e., having sustained injuries that could affect the growth of the fingers), or attachment patterns (e.g., parents' occupations and education levels used to compute family socioeconomic status, family structure, sibling order). In addition, other demographic variables assessed were self-reported ethnicity and participants' first language.

2.3. Instruments and Measures

2.3.1. Attachment Scales

Participants completed a series of self-report questionnaires to assess the degree to which they conformed to particular patterns of adult romantic attachment. Several researchers have contended that questionnaire-based measures of romantic attachment are more predictive of mating outcomes such as couple stability and sexual behaviour and more accurately represent one's orientation towards short-term vs. long-term mating than interview-based measures such as the Adult Attachment Interview (AAI; George, Kaplan, & Main; 1984, 1985, 1996), which may be primarily measuring one's level of caregiving/parenting outcomes (e.g., Del Giudice, 2009; Bernier & Dozier, 2002; Kirkpatrick, 1998). Consistent with this, the correlation between AAI classification and self-reported romantic attachment has been found to be low (r = .09) (e.g., Roisman et al., 2007). That being said, questionnaire-based methods do demonstrate a moderate correlation with interview measures assessing the *same* content domain (e.g., romantic relationships), with correlation coefficients ranging in size from r = .27 to r = .50 (Griffin & Bartholomew, 1994; Bartholomew & Shaver, 1998). These findings suggest modest convergent validity between the two main research traditions in the attachment literature, and support the use of self-report questionnaires to assess romantic attachment.

In the present study, two attachment questionnaires were administered. In order to assess individual differences in typical attachment to a romantic partner, participants completed the Experiences in Close Relationships (ECR) Questionnaire (Brennan, Clark, & Shaver, 1998). The ECR is considered to be a preferable measure of romantic attachment compared to other commonly used scales (Fraley, Waller, & Brennan, 2000). It is comprised of two subscales: an 18-item anxiety subscale, which assesses an individual's concern about abandonment in a romantic context, and an 18-item avoidance subscale, which assesses an individual's discomfort with closeness in a romantic context. Sample items include: "I worry a fair amount about losing my partner" (anxiety), and "I don't feel comfortable opening up to romantic partners" (avoidance). Participants are asked to rate the extent to which they agree with each statement using a 7-point Likert scale, ranging from 1 (strongly disagree) to 7 (strongly agree). Ratings are based on how an individual generally experiences relationships and not just in what is happening in a current relationship. Following the usual scoring procedure, items on each of the subscales were summed and averaged. In my sample, the alpha coefficient for the avoidance subscale was $\alpha = .91$ and $\alpha = .92$ for the anxiety subscale. These statistics are highly similar to those obtained by other researchers using the ECR (e.g., Brennan et al., 1998; Edelstein et al., 2010).

The ECR provides a continuous rating of participants' levels of attachment avoidance and anxiety with a typical romantic partner. Research suggests that attachment is a quantitatively distributed variable that may be better modelled with graded dimensions rather than categories, especially in the realm of adult attachment (e.g., Fraley & Waller, 1998). Consequently, the primary measure of interest was participants' continuous scores along the two dimensions (see Section 3.2.1 for how measures of romantic attachment, including continuous measures from the Relationship Questionnaire described below, were integrated into the LHT-based model tested).

Although a revised version of the ECR has been developed (ECR-R; Fraley et al., 2000), empirical examination of the ECR-R's construct validity has found that several of the new items comprising the Anxiety subscale have large amounts of unexplained variance (Fairchild & Finney, 2006). The authors suggest that some of these new items may not be addressing issues specific to attachment anxiety (e.g., "My partner only seems to notice me when I'm angry") and thus may not be representing the latent construct for which they were written. Furthermore, it has been suggested that many of the ECR-R items are conceptually redundant, thereby reducing scale parsimony (i.e., Fairchild & Finney, 2006; Fraley et al., 2000). For these reasons, I decided to use the original ECR as the primary measure of romantic attachment.

Because of the relevance of romantic attachment within the current study, a second measure of adult romantic attachment was administered—The *Relationship* Questionnaire (RQ; Bartholomew & Horowitz, 1991). The RQ is a widely used and well-validated measure of romantic attachment (e.g., Schmitt, 2005; Sibley, Fischer, & Liu, 2005). The RQ consists of four short paragraphs describing the four attachment styles. For example, the dismissing prototype reads as follows: "I am comfortable without close emotional relationships. It is very important to me to feel independent and self-sufficient, and I prefer not to depend on others or have others depend on me." Respondents were asked to read each description and rate the *degree* to which each description "corresponds to your general relationship style" on a 7-point Likert scale, ranging from 1 (not at all like me) to 7 (very much like me). Responses to the four questions provided a continuous measure of the degree to which each participant endorsed avoidant, anxious, secure, and fearful attachment behaviours. The RQ also contains a forced-choice question where participants are asked to choose which of the four paragraphs is *most* characteristic of their personality. Because of the relative brevity of both the continuous and categorical components of the RQ, its psychometric properties are considerably weaker than the ECR. Nevertheless, the RQ remains a commonly used measure of romantic attachment within the attachment literature. In addition, it is the
only measure, among popular measures of attachment, to demonstrate independence from self-deceptive biases (Leak & Parsons, 2001). Reliability estimates for the RQ categorical classifications over an 8-month interval is .35, with approximately 60% of individuals self-reporting the same attachment pattern from time 1 to time 2. Test-retest reliability for the RQ continuous ratings is higher, ranging from .70 over a 3-week interval to .50 over 8-months (Scharfe & Bartholomew, 1994; Sibley et al., 2005).

Although LHTs of attachment (e.g., Belsky, 1997; Del Giudice, 2009) predict continuity between attachment patterns in childhood and adulthood, discontinuity in attachment across development may sometimes exist (for review see Fraley, 2002). Changes in attachment categories have been found in response to environmental (including emotional) experiences that diverge from existing experience. For example, supportive spousal relationships may help to moderate the effects of difficult early attachment relationships (Brown & Harris, 1978; Crockenberg, 1987; Quinton, Rutter, & Little, 1984). In addition, major life transitions that involve the adoption of new social roles, such as leaving for university or getting married may be opportune times for people to reorganize their internal working models of attachment (Scharfe & Bartholomew, 1994). Trivers (1985) reasoned that offspring behaviour is only guided by parental behaviour as long as the offspring is *dependent* on the parent. Thus childhood personality traits, including attachment, influenced by parents are adaptive in the limited context of parental care, but can be modified or replaced in the transition to adulthood.

The *Retrospective Attachment Questionnaire* (RAQ; Parkes, 2006) was administered to assess participants' attachment to parental figures in childhood. The RAQ is a self-report questionnaire that asks adults about their childhood relationships with parents, using 30 yes-no questions rated for each parent independently (e.g., "Was either parent inclined to tease you or make you feel small?"), and about their childhood emotional experiences, using 33 additional yes-no questions (e.g., "Did you find it hard to accept cuddles, or other demonstrations of affection?"). Questions on the first part of the RAQ referred to the parents who raised the participant during childhood. As such, participants raised in single-parent homes were unable to answer the 30 questions about their childhood relationship with their absentee parent. No established procedure exists, however, for handling missing data due to parental absence, which likely has important consequences for an individual's attachment history (C. Parkes, Personal Communication, September 11, 2013). To handle this issue, participants who failed to provide responses for one parent on Part I of the questionnaire due to single parenthood were assigned a "yes" response for five specific items. These items all inquired about separation from or absence of a parent (e.g., "Were you separated from either parent for more than a month before the age of 6 years?"; "Was your parent often away or not available?"). The remaining 25 items were coded as "no" because the absentee parent was not available for the participant to have had any of the experiences described with him/her during their childhood. By contrast, an established procedure does exist for handling random individual missing responses on Part I and Part II of the RAQ (see Parkes, 2006).

Based on responses to the RAQ, continuous scores of attachment security/insecurity, anxiety, avoidance, and disorganization were derived following the standard computational criteria outlined by Parkes, 2006. Higher scores reflected greater levels of insecurity and/or its subtypes. Scores reflected style of attachment to the parental unit, rather that to mothers and fathers individually. The categorization of attachment patterns using the RAQ demonstrates high test-retest reliability, ranging from .81 to .91 across the different attachment styles (Parkes, 2006). Internal consistency estimates for the four attachment patterns have been reported to be strong, ranging from α = .80 (avoidant attachment) to α = .94 (secure/insecure attachment score) (Parkes, 2006). Within the current study, alphas ranged from moderate to high: $\alpha_{insecure}$ = .85; $\alpha_{anxious/ambivalent}$ = .85; $\alpha_{avoidant}$ = .58; $\alpha_{disorganized}$ = .66. Although the RAQ is relatively new in the literature, Crowell et al. (2008) endorse the RAQ as a potentially valuable measure of attachment history.

2.3.2. Personality and Motivational Scales

In order to measure individual differences in willingness to engage in casual, uncommitted sexual relationships, or as Buss and Schmitt (1993) frame it, the orientation towards or away from a short-term mating strategy, participants completed the *Sociosexual Orientation Inventory* (SOI; Simpson & Gangestad, 1991). The SOI is a 7-

62

item scale that assesses attitudinal (e.g., "Sex without love is OK") and behavioural markers (e.g., "With how many partners have you had sex on one and only one occasion, in your lifetime?") of sociosexual orientation. One question from the revised version of the SOI (SOI-R; Penke & Asendorpf, 2008), which inquires about frequency of sexual fantasies outside of a committed relationship ("How often do you have fantasies about having sex with someone you are not in a committed romantic relationship with?"), was included in the present questionnaire to further assess attitudes towards engagement in uncommitted sex. The first 3 items are assessed using an open-ended response format, while the remaining 5 items are rated on a Likert scale, with item 4 rated on a scale from 1 (never) to 8 (at least once a day), items 5-7 rated on a scale from 1 (strong disagree) to 9 (strongly agree), and item 8 rated on a scale from 1 (never) to 5 (at least once a day). A higher score on the SOI is indicative of a more *unrestricted* sociosexual orientation. The SOI demonstrates an adequate level of internal consistency ($\alpha = .73$ as found by Simpson & Gangestad, 1991; $\alpha = .75$ in the current study) and a high temporal reliability (r = .94) over a two-month test-retest period (as cited in Schmitt, 2005). Discriminant validation evidence reveals that sociosexuality does not correlate highly with sex drive, sexual satisfaction, sex-related anxiety, or sex-related guilt, but predicts if someone has engaged in sex with more than one partner in a given time period (Simpson & Gangestad, 1991).

The primary distinction between the original SOI and the revised version, the SOI-R, is the inclusion of sociosexual desire questions on the SOI-R (Penke & Asendorpf, 2008). Given that I also administered the Sexual Desire Inventory (see below; Spector et al., 1996), which directly measured men's motivation for sex in the abstract, there was deemed to be no added advantage to using the SOI-R. Moreover, the SOI-R and its facets have been found to be unrelated to romantic attachment styles (Penke & Asendorpf, 2008), which contrasts with the SOI, which has been found to significantly correlate with romantic attachment patterns across several cultures (e.g., Schmitt, 2005, Brennan & Shaver, 1995; Jackson & Kirkpatrick, 2007).

Because the literature suggests that the adoption of a short-term, opportunistic mating approach involves a low level of commitment in romantic relationships (e.g., Belsky et al., 1991), Lund's (1985) nine-item *Commitment Scale* was administered. The

Commitment Scale assesses participants' commitment in a romantic relationship (e.g., "How likely is it that you and your partner will be together six months from now?"). Responses to each of the nine items are made on a 1 (not at all) to 7 (very much) scale. A higher score is indicative of a greater level of commitment. Because the questions on this scale inquire about level of commitment within a *current* or existing romantic relationship, the measure could only be completed by participants who were in a romantic relationship at the time of testing. Reliability, as measured by Cronbach's alpha, is typically high ($\alpha = .85$ in the current study; $\alpha = .82$ as found by Lund, 1985). Higher scores on the Commitment Scale are seen in individuals who define themselves as more exclusive and pledge to continue their current romantic relationships (Lund, 1985).

Because higher sexual motivation or desire may be required to more successfully engage in a short-term mating strategy (Buss, 1998), the 15-item modified version of the Sexual Desire Inventory (SDI; Spector et al., 1996) was administered. The SDI is a measure of sexual desire in the abstract and thus can be completed independent of partner availability (e.g., "How strong is your desire to engage in sexual activity with a partner?"). The current version of the SDI was adapted by the addition of one question: "During the last month, how often have you had sexual thoughts?" (see also van Anders, 2012). Subjects are asked to indicate the strength and frequency of sexual desire over the past month on an 8-point scale that ranged from "no desire" to "strong desire." The inventory produces total, solitary, and dyadic SDI scores. Internal consistency of the 15item measure used in the current investigation was high at $\alpha = .88$. In Spector et al. (1998), the SDI showed a test-retest reliability of 0.76 over a 1-month period. The SDI has been validated in numerous studies (Spector et al., 1996; Spector & Fremeth, 1996; Galyer et al., 1999; King & Allgeier, 2000).

The Aggression Questionnaire (AQ; Buss & Perry, 1992) was administered, with the addition of 5 supplementary items. The AQ assesses four factor analytically derived components of aggression, including instrumental components (*physical aggression*, *verbal aggression*), affective components (*anger*), and feelings of ill will and injustice, which constitute the cognitive component of aggression (*hostility*). Five additional verbal aggression items taken from the Verbal Aggressiveness Scale (Infante & Wigley, 1986) were added to the original 29-item scale, in order to supplement the verbal aggression subscale of the AQ. This was done because the AQ verbal aggression scale consists of only five items, which is very limited. All statements (e.g., "I have become so mad that I have broken things") were rated on a 5-point Likert scale ranging from 1 (extremely uncharacteristic or men) to 5 (extremely characteristic of me). The score on each subscale is the sum of the item ratings. The total aggression score is the sum of the four subscales. Internal consistency across the four scales has been found to generally range from $\alpha = .70 - .85$ (Buss & Perry, 1992; Harris, 1997; Hampson et al., 2008). Similarly, in the current sample, alpha reliability was found to range between $\alpha = .77 - .83$. For the total modified AQ score, Cronbach's alpha was even stronger at $\alpha = .90$. Test-retest reliabilities for the subscales were found to be high over a nine-week interval (r = .72 - .80; Buss & Perry, 1992). The AQ also demonstrates moderate to high convergent validity with other measures of aggression, including scales designed for clinical use (Harris, 1997).

The *Risk-Taking Scale* from the Jackson Personality Inventory-Revised (JPI-R) (Jackson, 2004) was used to measure risk-taking tendencies. Participants are asked to provide true or false answers for 20 items (e.g., "I think I would enjoy almost any type of gambling") that assess four facets of risk-taking: monetary, physical, social, and ethical risk-taking. Internal consistency reliability values are high, ranging from .81 to .84 using coefficient alpha (Jackson, 1994); a similar estimate was obtained in the current study ($\alpha = .83$). Test-retest reliability for the Risk-Taking Scale has been found to range between .84 and .95 in a sample of college students (Jackson, 1994). In a test for validity, Jackson (1994) reported correlations with the completion of an adjective checklist of r = .75.

Selected subscales of the Personality Research Form—Form E (PRF-E; Jackson, 2003) were used to measure dominance, impulsivity, and autonomy. Participants read each statement such as "I try to control others rather than permit them to control me" (Dominance), "I often stop in the middle of one activity in order to start something else" (Impulsivity), or "I could live alone and enjoy it" (Autonomy) and give a true-false response based on agreement with the statement. Cronbach's alpha reliability coefficients have been found to be high for all three components, with $\alpha = .90$ for

Dominance, $\alpha = .79$ for Impulsivity, and $\alpha = .76$ for Autonomy (Jackson, 1999; Paunonen, 1998). I obtained comparably high internal consistency values with $\alpha = .75$ for Impulsivity, $\alpha = .75$ for Autonomy, and $\alpha = .76$ for Dominance. Evidence of convergent validity for the PRF-E subscales has been demonstrated (Jackson, 1999). For example, Dickman (1990) found high correlations between the PRF Impulsivity scale and other measures of impulsivity, ranging from r = .64 to .83.

In order to quantify participants' typical mood patterns, the Profile of Mood States (POMS; McNair et al., 2003) was administered. The POMS asked participants' to rate each of 65 mood-related adjectives on a 5-point Likert scale, ranging from 0 (not at all) to 4 (extremely), reflecting the extent to which each word accurately described the participants' typical mood. Ratings can be summed to yield scores on six factoranalytically derived subscales: Tension, Depression, Anger, Vigor, Fatigue, and Confusion. Only subscale scores for Tension and Depression were computed for the current study because of the known associations between internalizing problems and attachment insecurity and T, respectively. High levels of internal consistency have been found for each of the subscales in normative samples of adult men, ranging from .83 (Confusion) to .92 (Depression and Anger) (McNair, Lorr, & Droppleman, 1992). Consistent with this, Cronbach's alpha values for the Tension and Depression subscales in the current study were $\alpha = .88$ and $\alpha = .93$, respectively. The POMS total score correlates highly with another general measure of mood state, the Visual Analog Mood Scale (r = .79), while the Tension and Depression subscales have both been found to correlate highly with other well-known measures of anxiety and depression, respectively (Nyenhuis et al., 1999).

Because higher levels of self-esteem may represent a socially competitive trait predictive of a shorter-term mating approach (Buss & Schmitt, 1993), The *Rosenberg Self-Esteem Scale* (RSE; Rosenberg, 1965) was administered. The RSE consists of 10 items assessing global self-esteem (e.g., "On the whole, I am satisfied with myself"), rated on a 4-point scale ranging from strongly disagree to strongly agree. Previous studies have reported high internal consistency estimates, ranging from .72 to .90 (Gray-Little, Williams, & Hancock, 1997; Robins, Hendin, & Trzesniewski, 2001). Similarly, within the current investigation the Cronbach's alpha was high at $\alpha = .88$. The RSE also shows test-retest reliabilities ranging from .82 over a 1-week interval (Fleming & Courtney, 1984) to .63 over a 6-month period (Byrne, 1983) and .50 on a 1-year retest (McCarthy & Hoge, 1984). The RSE converges moderately to highly (r = .62 - .93) with other measures of global self-esteem (Robins et al., 2001).

The *Marlowe-Crowne Social Desirability Scale Form C* (MC-C; Reynolds, 1982) was administered to evaluate whether participants were responding to questionnaires in an accurate and truthful manner. The MC-C is a 13-item short form of the original Marlowe-Crowne Social Desirability Scale (Crowne & Marlowe, 1960). Shorter forms are suggested to be more amenable for use in personality research as they correspond more closely to the length of other unitary trait/state measures of personality (Reynolds, 1982; Fisher & Fick, 1993). The MC-C is composed of 5 items which are keyed "true" (i.e., highly desirable behaviours but low probability of occurrence, e.g., "I'm always willing to admit it when I make a mistake") and 8 items which are keyed "false" (i.e., socially disapproved behaviours but high probability of occurrence, e.g., "There have been times when I was quite jealous of the good fortune of others"). Each item is rated on a 2-point (true-false) scale. High scale scores indicate a strong tendency to respond in a socially desirable fashion. The MC-C demonstrates good psychometric properties with internal consistency estimates ranging from .62 to .76, including in the current study ($\alpha =$.62) (Ballard, 1992; Loo & Thorpe, 2000; Reynolds, 1982; Zook & Sipps, 1985) and a six-week test-retest correlation of .74 (Zook & Sipps, 1985). Scores on this form have been found to correlate highly with the original Marlowe-Crowne Scale with reported values of .91 to .96 (Fischer & Fick, 1993; Loo & Thorpe, 2000; Reynolds, 1982). Because the MC-C uses the same response format as the Risk-Taking scale and PRF-E, the items for all of the scales were randomly ordered and merged to form a composite questionnaire for the purposes of test administration.

In order to obtain information about each participant's relationship/dating history, a Dating History Questionnaire was created and administered (see Appendix C). Developed by the experimenter, the questionnaire was comprised of seven questions inquiring about participants' current partnering status (i.e., single, steady partner but living apart, steady live in partner, more than one steady partner, married, divorced, separated, widowed), the length of participants' most recent relationship (in months), participants' typical relationship length (in months), age at first intercourse (if sexually active), as well as the total cumulative lifetime number of intimate relationships (both sexual and romantic). For comprehensiveness, participants were also asked the number of children they had, as this may be indicative of greater mating effort (e.g., Pollet, Cobey, & van der Meij, 2013). The last question provided participants with an open opportunity to explain to the experimenter if there were any reasons why their answers on the questionnaires would be atypical (e.g., recent relationship dissolution, current stressor).

2.4. Hand-Related Measurements

Digital images of the underside of both the right and left hands were used to measure the ratio of the lengths of the second to fourth digits (2D:4D ratio). Images were obtained by using a digital copier/scanner, and one hand was imaged at a time. Participants were asked to place their fingers in a splayed position on the glass scanning surface and press lightly so as to distribute pressure evenly over the surface of the hand. Finger length was measured from the images produced using a high-precision digital calliper (Digital Measurement Metrology, Inc., Model ABS) with a resolution of 0.005 mm. The physical landmarks used to measure digit length were the most basal crease where the finger joins the palm and the most distal point on the fingertip (see Figure 2 for illustration of hand measurements made). The experimenter was the primary rater. Interrater reliability for the 2D:4D measurement is typically very high, with intra-class correlations of r = .95 or higher (e.g., Hampson & Sankar, 2012b; Berenbaum et al., 2009).

2.5. AR Genotyping

To minimize saliva impurities, participants abstained from eating, drinking fluids other than plain water, smoking, chewing gum, or brushing their teeth for 30 minutes before sample collection. Before the first saliva sample was obtained, participants rinsed their mouths with water. Participants then collected about 2 mL of saliva into a sterile



Figure 2. Illustration of hand-related measurements. Overview of the hand measurements performed on the digital images taken of participants' right and left hands.

Oragene DNA vial (DNA Genotek, Inc., Kanata, Ontario, Canada). Once collected, the whole saliva was mixed with 2 mL of Oragene DNA stabilizing solution and stored prior to analysis. Saliva sampling produces a higher DNA yield than mouthwash or buccal swab methods, and better quality DNA than the buccal swab (Rogers *et al.*, 2007). The median yield using the Oragene method is 110 ug.

Genotyping of the repeat polymorphism in the gene coding the androgen receptor (*AR*) was performed by The Center for Applied Genomics at The Hospital for Sick Children in Toronto (Canada). In brief, 50 ng of DNA was extracted from the saliva samples, and the CAG repeat region of the *AR* was amplified using polymerase chain reaction with one primer labeled with 6-FAM dye for visualization (5'-CTTTCCAGAATCTGTTCCAG-3') and a second unlabeled primer (5'-GAAGGTTGCTGTTCCTCATC-3'). Amplified fragments were run through capillary electrophoresis and were read using an ABI3730XL DNA Analyzer (Applied Biosystems Inc, Foster City, California) to separate the polymerase chain reaction products according to size. Quantification of the length of the CAG repeat region from each sample was accomplished using GeneMapper software (version 3.5; Applied Biosystems). Repeat numbers were confirmed by sequencing a subset of samples with alleles of different lengths.

2.6. Testosterone Measurement

For the T measurements, participants collected 3 mL of whole saliva into a polystyrene culture tube. A separate tube was supplied for the first and second saliva collection. Participants were instructed to use passive drool to collect the saliva, where the chin is tilted downwards to let the saliva naturally and passively accumulate at the edge of the mouth. No saliva stimulation agent (e.g., gum) was used because constituents in commercial gums have a potential to interfere with the accuracy of certain assays (van Anders, 2010). Once a bolus of saliva had accumulated, participants were instructed to spit out the saliva into the tube. This procedure was repeated until they reached the 3 mL mark. Specimens were stored at -20 °C until analysis.

Saliva contains only that fraction of T that is not bound to sex hormone binding globulin (SHBG); salivary T derives from the free and albumin-bound fractions of T in plasma (Quissell, 1993). These are biologically active, or potentially available to tissue for metabolic purposes, while the SHBG-bound fraction is considered to be relatively biologically inert (e.g., Manni et al., 1985; Cummings & Wall, 1985). Therefore salivary T closely approximates the bioavailable fraction of T that is available to interact with the brain (Pardridge & Demers, 1991; Sannikka *et al.*, 1983). High correlations between saliva and serum free T concentrations have been reported, r = 0.97 (Vittek et al., 1985); r = 0.94 (Wang et al., 1981).

The saliva was analysed in duplicate without extraction via radioimmunoassay (RIA). A ¹²⁵I Coat-A-Count kit for testosterone (Siemens Healthcare Diagnostics Inc., Deerfield, Illinois) was modified for saliva according to an established laboratory protocol (Moffat & Hampson, 1996). The antiserum is highly specific for T, showing cross-reactivity with dihydrotestosterone <5% and negligible cross-reactivity with other steroids. The intra-assay coefficient of variation averaged 5%. Assay sensitivity was 6 pg/mL.

2.7. Data Analytic Approach

All data were entered using SPSS software for Windows, version 22.0 (SPSS, Inc., Chicago, IL, USA). Descriptive statistics, alpha coefficients, and zero-order correlations among study variables were computed in SPSS in order to describe the sample, as well as to provide an idea of the relationships among study variables. In order to test the fit of the hypothesized conceptual model, structural equation modelling (SEM) was conducted using SPSS AMOS version 22.0. SEM assesses the overall fit of the data to the constraints imposed by the model through maximum likelihood estimation of the model parameters. A two-step SEM approach (see Kline, 2011) was employed, consisting of an initial analysis of the measurement model using confirmatory factor analysis (CFA), followed by the analysis of the full structural equation model. The chi-square test, comparative fit index, and root mean square error of approximation were the statistics used to evaluate the fit of the model. Because the AMOS interface does not

allow for bootstrap analyses to be conducted when missing values are estimated by the program, follow-up mediation analyses of direct and indirect effects of interest were conducted using Mplus version 7.11 (Muthen & Muthen, 2010).

3. Results

3.1. Data Treatment

Prior to conducting the main analyses of this study, data on all questionnaire items, as well as biological variables (i.e., T levels, hand-related measurement, CAG repeat length), were examined to assess accuracy of data entry and the extent of missing values. Data were double entered to check for any errors in data entry or in reversecoding for specific questionnaire items. In addition, SPSS Frequencies were computed to check for any incorrectly entered data as well as to detect any potential univariate outliers.

Inspection of the extent of missing data on every questionnaire item revealed that the total item level missingness (i.e., when participants omitted one or more items within a multi-item measure) was < 1% both within and across all questionnaires administered. There were no participants in the sample who displayed variable level missingness on a multi-item instrument (i.e., missing all items on a multi-item measure). With respect to variables measured with single-item questions (e.g., age, typical relationship length, age at first intercourse), missing data on individual variables amounted to 2% or less. Across the entire paper-and-pencil portion of the questionnaire battery, no single participant had more than 3% (i.e., 11 items or less) of data points missing. These data indicate a very small degree of missingness within the sample.

Degree of missing data across the various biological measures obtained in the current study was also explored. No participants had missing data on either of the two T measurements. One participant lacked CAG repeat data, due to limited quality or quantity of DNA available from his saliva specimen. Digit ratio data had one missing case (<1% of the sample), due to the experimenter inadvertently making two copies of the right hand for one participant.

To handle the small proportion of missing questionnaire data in the sample, a case mean substitution approach was employed. Case mean substitution assigns the participant's mean score based upon the items that are present on the measure in question to the missing score for that participant. Because this strategy assumes that for any case, the score on any data point in a given measure is closely related to the scores on the remaining data points on that same measure, case mean substitution is particularly effective when the measure or scale in question is unidimensional and when the internal reliability estimate for the measure or scale is high (Fox-Wasylyshyn & El-Masri, 2005; Osborne, 2013 pp. 121-122). When these conditions are satisfied mean substitution is similar to imputation via multiple regression and is considered a defensible practice. Several authors have found that case mean substitution is able to favourably handle itemlevel missingness when data are missing on up to 20% of the items, in both random and systematic patterns (e.g., Downey & King, 1998; Roth, Switzer, & Switzer, 1999; Shrive et al., 2006). Although more sophisticated imputation techniques exist, Tabachnick and Fidell (2007) and Roth and Switzer (1995) state that when the number of missing data points is low (less than 5%), as was the case in the current study (< 1% missing), the choice of technique to handle missing data is of minimal importance.

Because the internal consistency estimates for the unidimensional measures or subscales employed in this study were acceptably high (see Method for Cronbach's alpha data) and the total number of missing data points was very low, case mean substitution was conducted for item-level missingness when the amount of missing data on any one scale was $\leq 20\%$. If the degree of missingness on a scale was $\geq 20\%$, those item data points were left as missing. Any single-item variables with missing data were also left as missing, as case mean substitution could not be performed in these cases. When conducting SEM analyses, any remaining missing data were handled through Maximum Likelihood (ML) estimation, the standard procedure in AMOS. ML imputes missing data by using all other available data points in a database and the multivariate relationship among the study variables (for available cases) to construct the best estimates for the missing values in question. ML is considered to be an appropriate technique to impute missing data when the total amount of missing data in a sample is $\leq 20\%$, regardless of

the pattern of missingness (i.e., systematic or random) (Fox-Wasylyshyn & El-Masri, 2005).

3.2. Analysis of Life History Theory-Based Model

3.2.1. Overview of Indicator Variables

The life history-based model tested in this study (see Figure 1) included a number of higher-order, unobserved variables. Each of these *latent* variables was measured by several observed items (i.e., manifest/indicator variables). The indicator variables were selected based on a consideration of exactly how each latent construct is defined within the various life history theories of attachment and reproduction. In what follows, a brief summary of the indicators associated with each latent variable in the integrated-LHT model will be presented. Please refer to Appendix D for a full and more detailed description of each indicator variable, its associated computation and/or coding, reliabilities of the indicators, and rationale for inclusion.

Early Family Environment. Ten manifest variables were initially used to measure the latent construct of Early Family Environment (see Figure 3 for a schematic representation of the latent variable and associated indicators). Variables listed included those that were explicitly outlined within the life history theories described in Section 1.2 and/or have been empirically related to attachment patterns in childhood, the timing of somatic development, or reproductive strategy choice. All indicators indexed the quality of the early family environment. The 10 indicators were: (1) a "Family affection" index and (2) a "Family abuse" index, derived from items from Part I of the RAQ (Parkes, 2006), (3) Mother's psychiatric health and (4) Father's psychiatric health, also derived from items from the RAQ Part I, (5) SES of the family household (either Total SES, computed only from information about parental occupation. The decision about which of the two SES indicators to include, given the overlapping computational information, was made at the level of the measurement model analyses, based on an empirical consideration of which indicator better optimized the latent construct. This topic will be



Figure 3. Indicators of Early Family Environment latent construct. Indicator (observed) variables initially used to index the Early Family Environment latent construct.

revisited in Section 3.2.5), (6) "Family structure" index for ages 0-4 years, (7) "Family structure" index for ages 5-9 years, and (8) "Family structure" index for ages 10 years and up, each of which quantified the effects of family structure (e.g., two-parent household vs. single parent living alone, etc.) during development on the quality of participants' early family environmental experience. Finally, indicators of (9) the presence or absence of a non-biological stepparent during the first 16 years of life and (10) the death of a parent during the first 16 years of life were included. Each indicator was coded such that a higher score was reflective of a poorer quality environment.

Because the latent variable of Early Family Environment is a weighted composite, where each indicator variable represents an *index* of the quality of the early family environment, the direction of causality runs from the indicators to the latent composite. This differs from the other latent constructs included in the model, which are assumed to cause the indicator variables.

Retrospective Childhood Attachment. Following Parkes' (2006) computational procedures, a total insecure attachment score was calculated from Part I and Part II of the RAQ. Higher scores on this continuous measure corresponded to a more insecure childhood attachment pattern. Scores reflected attachment to parents in general, which is consistent with the life history-based theories reviewed here that do not differentiate between the unique developmental consequences of maternal *vs.* paternal attachment configurations (for review see Del Giudice, 2009). Although individual scores for the 3 insecure attachment patterns (anxious/ambivalent, avoidant, disorganized) can be separately computed from the RAQ, because Belsky et al. (1991) and Chisholm (1999) speculate only about the causes and consequences of overall attachment insecurity (*vs.* security) without differentiating among the specific varieties of insecure attachment in their theoretical models, the decision was made to only compute the total insecure attachment score as a single indicator of the childhood attachment style.

Non-Sexual Reproductive Strategy. The indicator variables thought to be caused by the Non-Sexual Reproductive Strategy latent construct (see Figure 4) included the personality traits usually associated with the Young Male Syndrome (Wilson & Daly,



Figure 4. Indicators of Non-Sexual Reproductive Strategy latent construct. Indicator (observed) variables presumed to be caused by the Non-Sexual Reproductive Strategy latent construct.

1985), which Chisholm (1999) argues is the optimal reproductive strategy for males reared in environments of high uncertainty. Indicators of this personality cluster included total scores on the modified Buss and Perry (1992) Aggression Questionnaire, the Risk Taking Scale of the JPI-R (Jackson, 2004) and the Impulsivity subscale from the PRF-E (Jackson, 2003). Because dominance is a closely related concept to aggression (Archer, 1991), the total score on the Dominance subscale of the PRF-E was also included. In addition, measures of autonomy, self-esteem, and trait anxiety (reverse coded) were included. Total scores on the Autonomy subscale of the PRF-E, the RSE, and the Anxiety subscale of the POMS were the observed variables, respectively. Each indicator variable was coded such that a higher score reflected a higher level of the construct.

Sexual Reproductive Strategy. Both Belsky et al. (1991) and Del Giudice (2009) argue that the reproductive strategies predicted by insecure *vs.* secure attachment fall along a continuum of shorter-term to longer-term mating, earlier to later sexual activity, and unstable to enduring pair bonds. Consequently, the latent factor Sexual Reproductive Strategy (see Figure 5) was assumed to causally influence participants' age at first intercourse, their typical relationship length, as well as their total number of lifetime partners (both romantic and sexual). A current partner status observed variable was also included, which reflected whether or not participants were currently in a relationship, and if so, what type (see Appendix D), coded in a graded fashion according to how closely their current partnering status fit with a shorter-term *vs.* longer-term mating strategy.

Several components of the SOI (Simpson & Gangestad, 1991) were core indicators of the latent Sexual Reproductive Strategy construct and were individually included as separate items. Item 2 ("How many different partners do you foresee yourself having sex with during the next five years?") and Item 3 ("With how many partners have you had sex on *one and only one* occasion, in your lifetime?") were included as independent indicators. The standardized factor scores for the four items comprising the *attitudinal* facet of the SOI (items 4-8, excluding item 7) were used as an additional latent variable.



Figure 5. Indicators of Sexual Reproductive Strategy latent construct. Indicator (observed) variables presumed to be caused by the Sexual Reproductive Strategy latent construct.

As noted during the model presentation in Section 1.9, because life history theorists are not in agreement as to whether romantic attachment itself is part of one's sexual reproductive strategy (Belsky, 1997; Kirkpatrick, 1998; Del Giudice et al., 2009), or whether it is a correlated trait that predicts one's sexual strategy (e.g., Hill et al., 1994a; Del Giudice, 2009), the various romantic attachment subscale scores from the RQ and ECR were tentatively included as separate indicator variables of the Sexual Reproductive Strategy latent factor. Indicators were therefore: the ECR avoidance subscale score, the ECR anxiety subscale score, the RQ dismissing score, and the RQ preoccupied score, the latter as continuous variables. Because a person's overall level of romantic attachment security might also be a characteristic of his reproductive strategy choice, the RQ continuous security score (reverse-coded such that a higher score indicated less attachment security) was included as an indicator. Finally, there is a lack of clarity in LHT-based models about the role of adult fearful attachment (e.g., Del Giudice, 2009). Consequently, participants' RQ continuous fearful score was also included as a possible indicator variable (see Appendix D for further discussion). Each indicator variable of the Sexual Reproductive Strategy was coded (or reverse coded if necessary) such that a higher score was consistent with a shorter-term reproductive strategy.

Biological Variables. For Adult T (Range: 30.53 to 302.98 pg/mL), the internal consistency between first and second measurements was very high (α = .95), thus it was deemed appropriate to average the two measurements and use this as a single-indicator variable in the model. Similarly, the Cronbach's alpha between the right and left hand digit ratio was in an excellent range for biological measures (α = .84) and it was also deemed appropriate to average the right and left hand measurements as a single-indicator of individual differences in prenatal T exposure (2D:4D; Range: .90 to 1.04). The CAG repeat length for each participant was measured in duplicate, however, within our sample none of the duplicates were found to differ. Thus the CAG repeat data displayed perfect correspondence, and as such was represented as a single-indicator in the model (CAG RL Range: 13 to 30 repeats).

3.2.2. Descriptive Statistics

Descriptive statistics for all variables in the model are presented in Table 3. Means and standard deviations for all standardized measures were considered representative, as they were comparable to estimates obtained with samples of male university students in other published literature (Gentzler & Kerns, 2004; Edelstein et al., 2010; Schmitt, 2005; Bogaert & Rushton, 1989; Jackson, 1994; 1999; Buss & Perry, 1992; Hampson et al., 2008; McNair & Heuchert, 2005; Nyenhuis et al., 1999; Sinclair et al., 2010). Similarly, descriptive statistics for all androgen-related variables were comparable to those obtained in other independent studies of young adult men (e.g., For CAG RL: Medland et al., 2005; Sankar & Hampson, 2012b; Skjaerpe et al., 2009; For 2D:4D: Hampson & Sankar, 2012b; Maner et al., 2014; Medland et al., 2010; For salivary T collected at a similar time of year (i.e., Winter to Spring): Stanton, Mullette-Gillman, & Huettel, 2011; Moffat & Hampson, 2000; White, Thornhill, & Hampson, 2006). As reviewed in Sections 2.3.1 and 2.3.2, Cronbach's alpha for the various measures administered were similar to estimates cited in the literature. As Table 3 shows, most of the individual study variables had acceptable internal consistency (α \geq .75). The lowest alpha coefficients were for the derived measures of family affection, family abuse, and parent psychiatric health from the RAQ (α ranging from .48-.64). The lower internal consistency estimates suggest that these scales contained items that assess somewhat inconsistent concepts, which is not unexpected given the nature of the behaviours described by those scales.

3.2.3. Evaluation of Assumptions

Estimation in SEM with ML assumes multivariate normality of continuous outcome variables. Because evaluation of some of the assumptions of multivariate normality is impractical (e.g., ensuring that the joint frequency distributions of all possible combinations of variables are bivariate normal), and statistical tests used to detect multivariate violations can be influenced by small departures from normality in larger samples (Kline, 2011), inspection of univariate distributions is considered to be a

Table 3

Descriptive Statistics for Model Variables

1	0				SE of		SE of
Variable	α	M	SD	Skew	Skew	Kurtosis	Kurtosis
Farly Family							
Environment LV							
Family Structure							
Index (0-4)		1 12	0.58	4 75	0.17	20.94	0.35
Family Structure		1.12	0.00	1.70	0.17	20.91	0.50
Index (5-9)		1 24	0 76	3 00	0 17	7 49	0.35
Family Structure		·· - ·	0170	2.00	0117	,,	0.00
Index (10-16)		1.38	0.89	2.14	0.17	3.04	0.35
Stepparent		0.02	0.16	6.05	0.17	34.95	0.35
Parent Death		0.03	0.17	5.48	0.17	28.28	0.35
Total SES^1		13.24	2.19	-0.76	0.18	0.00	0.35
Financial SES ¹		11.93	3.75	-0.20	0.18	-0.89	0.35
Family Affection	0.58	1.42	1.50	0.80	0.17	-0.19	0.35
Index							
Family Abuse Index	0.48	0.82	1.14	1.44	0.17	1.85	0.35
Mother's Psychiatric							
Health	0.64	0.90	1.10	1.60	0.17	2.50	0.35
Father's Psychiatric							
Health	0.56	0.31	0.71	2.54	0.17	6.02	0.35
RAQ Insecurity	0.85	14.05	7.62	0.62	0.17	0.22	0.35
Non-Sevual							
Reproductive							
Strategy LV							
AO Total	0.90	92 93	19 58	0.27	0.18	-0.51	0.35
Risk-Taking	0.83	12.16	4 57	-0.36	0.18	-0.71	0.35
Impulsivity	0.75	6.81	3 44	0.29	0.18	-0.52	0.35
Dominance	0.76	11.71	3.11	-0.92	0.18	0.66	0.35
Autonomy	0.75	8 04	3 49	0.11	0.18	-0.66	0.35
Self-Esteem	0.88	22.64	5.26	-0.52	0.17	-0.39	0.35
POMS Anxiety ¹	0.88	10.97	6.52	0.64	0.17	0.09	0.35

Note. LV = Latent Variable; RAQ = Retrospective Attachment Questionnaire; AQ = Aggression Questionnaire; POMS Anxiety = Profile of Mood States-Tension Subscale ¹ Means and Standard Deviations are presented for variables without score reversal

Table 3 Continued

Descriptive Statistics for Model Variables

	-				SE of		SE of
Variable	α	M	SD	Skew	Skew	Kurtosis	Kurtosis
Sexual							
Reproductive							
Strategy LV							
SOI Q2		6.23	6.30	1.91	0.18	3.79	0.35
SOI Q3		3.19	5.46	4.53	0.18	30.45	0.35
SOI Attitudinal							
Factor	0.76	0.11	0.95	-0.48	0.17	-0.45	0.35
Total Number							
of Partners		10.85	14.27	4.44	0.17	25.91	0.35
Age at First							
Intercourse		17.11	2.06	1.25	0.18	3.17	0.35
Typical Rel'ship							
Length ^{1, 2}		10.60	17.88	4.12	0.18	19.29	0.35
Partnering Status		0.51	0.53	0.26	0.17	-1.26	0.35
ECR Avoidance	0.91	3.06	1.01	0.61	0.17	0.12	0.35
ECR Anxiety ¹	0.92	3.31	1.11	0.00	0.17	-0.71	0.35
RQ Dismissing		4.45	1.58	-0.28	0.17	-0.75	0.35
RQ Preoccupied ¹		3.02	1.61	0.55	0.17	-0.57	0.35
RQ Security		4.85	1.64	-0.68	0.17	-0.52	0.35
RQ Fearful		3.26	1.89	0.42	0.17	-1.01	0.35
2D:4D ratio	0.84	0.96	0.02	0.14	0.18	-0.03	0.35
Adult T (pg/mL)	0.95	87.13	31.06	2.26	0.17	11.89	0.35
CAG RL (# repeats)		21.80	2.66	-0.07	0.18	0.96	0.35

Note. LV = Latent Variable; SOI Q2/Q3 = Sociosexual Orientation Inventory Question 2/ Question 3; ECR = Experiences in Close Relationships Questionnaire; RQ = Relationship Questionnaire; CAG RL = CAG repeat length

¹ Means and Standard Deviations are presented for variables without score reversal ² Typical Relationship Length coded in number of months

good test of instances of multivariate nonnormality (Kline, 2011; Tabachnick & Fidell, 2007). SPSS Frequencies was used to examine departures from univariate normality. Kline (2011) and Bowen and Guo (2012) suggest that when the absolute value of skewness is > 3.0 and kurtosis is > 10.0, a variable may be problematic and require corrective action. That being said, Tabachnick and Fidell (2007) state that for larger samples inspection of the shape of the distribution for a variable may be more important than using the actual statistic. With respect to kurtosis, the authors suggest that with samples of 100 or more, underestimates of variance associated with positive kurtosis disappear, and with samples of 200 or more underestimation of variance with negative kurtosis disappears. Furthermore, even under conditions of severe nonnormality, ML estimation does not appear to affect standard errors of parameter estimates (Lei & Lomax, 2005), as well as parameter estimates themselves, especially if sample sizes are over 100 (Finch, West, & MacKinnon, 1997; Lei & Lomax, 2005). Consequently, only if variables had indices suggesting both problematic skew *and* kurtosis was corrective action deemed necessary.

As can be seen in Table 3, most variables had skewness and kurtosis values within an acceptable range. The only exceptions were a subset of the indicators of the Early Family Environment latent construct, as well as three of the indicators of the "Sexual Reproductive Strategy" latent construct. SOI Q3, Total Number of Partners, and Typical Relationship Length all showed positive skew and kurtosis. This was consistent with expectation as few participants were expected to have scores at the upper end of the distribution (i.e., very high number of one night stands, total number of partners, or very long relationship lengths). In order to handle non-normality on these variables, the commonly accepted procedure of logarithmic transformation was applied (Kline, 2011). Such transformation brought skewness and kurtosis indices within a more than acceptable range for each of the variables.

With respect to non-normal Early Family Environment indicators, the limited range of obtained scores on these variables likely contributed to the skewness they showed. Because the assumptions of multivariate normality pertain to *outcome* variables, however, correction for exogenous variables, which do not have associated tests of significance, is not required before application of ML estimation (Kline, 2011). More is said about this later. Nevertheless, because the proportion of scores of 0 on the Stepparent and Parent Death variables were so high (over 95%) these variables were deemed too severely non-normal to include in subsequent analyses.

As a further assessment of the assumptions of normality, bivariate correlation coefficients produced through SPSS were inspected to check for the presence of multicollinearity. As shown in Table 4, which presents the correlation coefficients among the model variables, no two variables were so highly correlated that they were essentially conveying the same information. The highest correlations were among the Family Structure Index variables and between Total Number of Partners and SOI Q3 (inquiring about number of partners one has had sex with on only one occasion). Follow-up modifications stemming from these correlations are explored in the confirmatory factor analysis for the model (Section 3.2.5).

Finally, pairwise linearity was inspected using bivariate scatterplots generated through SPSS for all relevant pairs of variables. The scatterplots indicated no significant departures from bivariate linearity.

3.2.4. Correlations among Model Variables

Table 4 shows the Pearson correlation coefficients among all model variables. Several patterns of relationships were noted. As expected, childhood attachment insecurity was significantly correlated with all indicators of early family environment. In all cases, higher attachment insecurity was associated with markers of a poorer quality early family environment (e.g., less stable family structure, lower SES, higher levels of family abuse). Evidence for continuity of attachment styles from childhood to adulthood was provided by the significant positive correlations between RAQ Insecurity scores and both the ECR avoidance and anxiety subscales, and the RQ preoccupied and fearful subscales. In addition, levels of adult romantic attachment security were significantly and negatively correlated with childhood attachment insecurity.

Table 4

Corre	ziuiions	umong	LIII-D	useu M		uriddle	ა 	6	6	4.2	
	1	2	3	4	5	6	7	8	9	10	11
1	-										
2	.66**	-									
3	.46**	.68**	-								
4 ¹	14	- .18 [*]	15*	-							
5 ¹	24**	34**	30**	.68**	-						
6	.14	.11	.14	21**	- .14 [*]	-					
7	.12	.22**	.26**	13	13	42**	-				
8	.13	.15*	.18*	- .16 [*]	13	.22**	.45**	-			
9	.00	.07	.06	01	.00	.19**	.20**	.34**	-		
10	.22**	.34**	.33**	13	- .16 [*]	$.58^{**}$.61**	.55**	.33**	-	
11	.04	.02	10	12	05	.28**	.25**	.20**	.03	.41**	-
12	01	11	- .19 ^{**}	.09	.15*	06	.01	07	09	13	.15*
13	02	.02	06	08	08	.11	$.17^{*}$.15*	02	.22**	.49**
14	.05	08	05	.09	.14	.03	.08	.00	08	.04	.22**
15	.02	11	08	.07	.13	.06	.02	.02	.00	03	01
16	- .21 ^{**}	- .16 [*]	- .18 [*]	.14*	.13	21 ^{**}	20**	24**	11	39**	25**
17 ¹	.19**	.14	.06	- .17 [*]	12	.08	.13	.25**	.16*	.29**	.31**
18	01	04	10	02	06	.12	04	08	.00	07	.19**
19 ²	.06	02	08	- .16 [*]	- .17 [*]	01	06	02	.01	08	.07
20	.11	08	02	10	01	.08	.11	.09	.03	06	.17*
21 ²	.08	02	03	- .17 [*]	11	.04	.03	05	.09	02	.06
22 ¹	.02	.02	.06	.01	09	.06	.04	.10	.06	.09	15 [*]
23 ^{1, 2}	05	05	.04	.01	.08	07	.06	.16*	.10	.00	14
24	05	10	12	.11	04	.02	04	.01	.00	.07	.13
25	.06	01	02	.09	04	.14	01	.02	.05	.20**	.18*
26 ¹	.12	.11	.12	- .17 [*]	- .16 [*]	.20**	.11	.14*	.15*	.35**	.28**
27	11	06	04	.07	$.17^{*}$	01	.06	.07	.09	.06	.01
28 ¹	.03	.08	.02	02	01	.11	.03	.00	08	.19**	.23**
29 ¹	11	09	04	04	.02	15 [*]	09	11	03	23**	21 ^{**}
30	.20**	.21**	.09	11	14	.12	.08	.13	.07	.32**	.19**
31	.02	00	.03	.04	04	10	04	02	11	- .14 [*]	47**
32	05	.05	.06	.00	01	11	.07	.04	.03	02	.00
33	01	.03	03	.07	04	.02	07	03	.08	.06	.17*
34	03	.02	.02	04	03	.02	01	.05	02	.05	01

Correlations among LHT-Based Model Variables

1) Family Structure (0-4), 2) Family Structure (5-9), 3) Family Structure (10-16), 4) Total SES, 5) Financial SES, 6) Family Affection Index, 7) Family Abuse Index, 8) Mother's Psych Health, 9) Father's Psych Health, 10) RAQ Insecure, 11) AQ Total, 12) Risk Taking, 13) Impulsivity, 14) Dominance, 15) Autonomy, 16) Self-Esteem, 17) POMS Anxiety, 18) SOI Q2, 19) SOI Q3, 20) SOI Attitude, 21) Total Partners, 22) Age 1st Intercourse, 23) Typical Relationship Length, 24) Partnering Status, 25) ECR Avoidance, 26) ECR Anxiety, 27) RQ Dismissing, 28) RQ Preoccupied, 29) RQ Security, 30) RQ Fearful, 31) MC-C, 32) 2D:4D, 33) Adult T, 34) CAG RL ¹ Variable presented without score reversal; ² Variable presented following log transformation; * p < .05. ** p < .01.

	12	13	14	15	16	17	18	19	20	21	22
1											
2											
3											
4 ¹											
5 ¹											
6											
7											
8											
9											
10											
11											
12	-										
13	.47**	-									
14	.34**	.11	-								
15	.20**	.08	06	-							
16	.12	08	.15*	12	-						
17 ¹	05	.19**	06	19**	35**	-					
18	.18*	.17*	.20***	.17*	.11	03	-				
19 ²	.11	04	.13	.10	.20	05	.47	-			
20	.18	.09	.14	.22	01	02	.48	.37	-		
212	.20	.04	.18	.24	.07	.02	.51	.68	.34	- **	
22 ¹	22	13	23	13	09	.00	30	34	38	36	-
231, 2	13	14	01	11	.02	.07	39	29	32	35	.17
24	03	.11	19	.28	12	.07	.26	.04	.15	.11	06
25	.04	.10	03	.40	29	.13	.26	.16	.17	.15	02
26'	16	.16	10	26	40	.40	24	18	15	18	.20
27	.06	.09	.01	.41	.03	05	.19	.19	.35	.22	20
28 ²	06	.18	07	15	20 21**	.24	10	12	10	10	.17
29	.06	12	.08	21	.31	20 22**	18	09	15	12	01
30 21	14 20**	.04	09	.00	32	.32	.00	03	.03	05	.11 16*
٥1 22	20	36	14	.09	.12	24	12	.03	16	.02	.16
32 22	04 10	00 10*	05 04	.03	08 17*	.04	08	.00	.04	07	02
33 34	.10	.18	.00	.00	1/	.12	.14	.02	07	.02	.00 15 [*]
34	.01	.03	.02	12	.03	.01	04	09	02	08	.15

Correlations among Model Variables

1) Family Structure (0-4), 2) Family Structure (5-9), 3) Family Structure (10-16), 4) Total SES, 5) Financial SES, 6) Family Affection Index, 7) Family Abuse Index, 8) Mother's Psych Health, 9) Father's Psych Health, 10) RAQ Insecure, 11) AQ Total, 12) Risk Taking, 13) Impulsivity, 14) Dominance, 15) Autonomy, 16) Self-Esteem, 17) POMS Anxiety, 18) SOI Q2, 19) SOI Q3, 20) SOI Attitude, 21) Total Partners, 22) Age 1st Intercourse, 23) Typical Relationship Length, 24) Partnering Status, 25) ECR Avoidance, 26) ECR Anxiety, 27) RQ Dismissing, 28) RQ Preoccupied, 29) RQ Security, 30) RQ Fearful, 31) MC-C, 32) 2D:4D, 33) Adult T, 34) CAG RL ¹ Variable presented without score reversal; ² Variable presented following log transformation; * p < .05. ** p < .01.

Table 4 Continued

	23	24	25	26	27	28	29	30	31	32	33	34
1												
2												
3												
4 ¹												
5 ¹												
6												
7												
8												
9												
10												
11												
12												
13												
14												
15												
16												
17 ¹												
18 10 ²												
19-												
20 21^2												
21 22 ¹												
22 23 ¹ , 2												
23	- 20**											
27	20 - 16 [*]	32**	_									
26 ¹	- 01	10	- 03	_								
27	08	.10	.27**	26**	-							
28 ¹	03	.17*	06	.55**	26**	-						
29 ¹	.07	35**	58**	.14	21**	08	_					
30	.02	.05	.41**	.37**	02	.08	51**	-				
31	.15*	00	06	28**	.04	- .18 [*]	.06	10	-			
32	.04	.01	04	.01	.02	01	01	04	.08	-		
33	08	.14	.22**	.03	.06	.03	14	.14*	10	12	-	
34	06	09	03	.25**	09	.11	.07	.05	07	12	03	_

Correlations among Model Variables

1) Family Structure (0-4), 2) Family Structure (5-9), 3) Family Structure (10-16), 4) Total SES, 5) Financial SES, 6) Family Affection Index, 7) Family Abuse Index, 8) Mother's Psych Health, 9) Father's Psych Health, 10) RAQ Insecure, 11) AQ Total, 12) Risk Taking, 13) Impulsivity, 14) Dominance, 15) Autonomy, 16) Self-Esteem, 17) POMS Anxiety, 18) SOI Q2, 19) SOI Q3, 20) SOI Attitude, 21) Total Partners, 22) Age 1st Intercourse, 23) Typical Relationship Length, 24) Partnering Status, 25) ECR Avoidance, 26) ECR Anxiety, 27) RQ Dismissing, 28) RQ Preoccupied, 29) RQ Security, 30) RQ Fearful, 31) MC-C, 32) 2D:4D, 33) Adult T, 34) CAG RL ¹ Variable presented without score reversal; ² Variable presented following log transformation; * p < .05. ** p < .01.

Also as expected, measures of romantic attachment avoidance (dismissing) were significantly associated with all of the sexual reproductive strategy indicator variables, such that a higher avoidance score was correlated with behaviours indicative of a shorter-term mating approach. Similarly, less endorsement of a secure romantic attachment style was associated with a more unrestricted sociosexual orientation (r SOI Q2 = -.18; r SOI Attitudinal Factor = -.15) and with a significantly lower likelihood of being in a committed partnership (r Partnering Status = -.35). Other correlations between RQ security and sexual behaviours were in the anticipated direction, but were non-significant. In contrast, the measures of romantic attachment *anxiety* (preoccupied) were significantly associated with all the sexual reproductive strategy indicators in such a way that higher anxiety correlated with scores indicative of *less* endorsement of a shorter-term mating approach. This is consistent with how anxiety-based strategies may be expressed in men. No significant associations were found between the fearful attachment pattern and the indicators of the sexual reproductive strategy; however, no *a priori* hypothesis had been made with respect to this particular attachment pattern.

In general, the continuous scores on the adult romantic attachment measures did not correlate very strongly or consistently with the non-sexual personality measures. One notable exception was the Autonomy subscale. Consistent with the conceptualization of avoidance as an attachment style that is characterized by independence and distancing oneself from others, avoidance scores showed a strong, positive correlation with scores on the Autonomy subscale of PRF-E (r = .40 and r = .41), while romantic attachment anxiety, which is characterized by greater dependency in intimate relationships, was significantly and negatively correlated with levels of Autonomy (r = -.26 and r = -.15). Interestingly, and contrary to expectation, a negative correlation was found between levels of self-esteem and participants' continuous dismissing-avoidance scores; higher levels of self-esteem were, instead, associated with higher romantic attachment security scores.

In line with the hypothesized relationship, basal T levels showed a significant, positive correlation with levels of ECR avoidance (r = .22). T failed to show, however, any relationship with ECR anxiety. By contrast, CAG repeat length was significantly and

positively correlated with ECR anxiety (r = .25), suggesting androgenic effects on both dimensions of adult romantic attachment. T concentrations also were correlated, as expected, with some of the non-sexual behaviours (e.g., aggression, impulsivity), but contrary to expectation, did not correlate strongly with the indicators of sexual reproductive strategy. Digit ratios, a proxy marker of prenatal T levels, showed no significant inter-correlations with any other study variables. This finding is consistent with the increasing scepticism in the literature, alluded to in Section 1.6.3, about the validity of the 2D:4D finger length ratio as a marker of fetal T levels in humans (e.g., McIntyre, 2006; Berenbaum et al., 2009; Hampson & Sankar, 2012b).

Correlations coefficients between the various indicators in the LHT-based model and the Marlowe-Crowne Social Desirability Scale Form C (MC-C) were also computed. Some significant inter-correlations were found, primarily with the anxiety-related measures (e.g., ECR anxiety, RQ preoccupied, POMS Anxiety). All significant correlations were either negative in direction for correlations with negative behaviours or positive in direction for positive characteristics, suggesting some degree of socially desirable responding in the study population, which was to be expected. Inspection of histograms in SPSS Frequencies showed that scores on the MC-C were normally distributed (M = 5.50, SD = 2.51). Because socially desirable responding is particularly a problem for those who have objectively high scores on measures of social desirability, only these cases may need to be dealt with (Harris, 1997). Within the current distribution, no participants had scores that fell in an outlier range (i.e., 3 standard deviations above the mean). Using a more conservative cutpoint of 2 standard deviations above the mean, only 7 participants in the sample fell into the "outlier" category at the upper bound. Removing these participants made a trivial difference to the size of the correlation coefficients described above. Furthermore, the full structural model described below was re-run without these 7 participants and it did not change the size or direction of the associations. Finally, partial correlations, controlling for MC-C scores, between variables that were significantly correlated with the MC-C were computed. These analyses also revealed trivial changes in correlation coefficient sizes. Consequently, it was deemed that social desirability was not influencing the study findings.

3.2.5. Measurement Model

The hypothesized model depicted in Figure 1 was tested using SEM analyses with SPSS Amos v. 22.0. Kline (2011), and others (e.g., Anderson & Gerbing, 1988), recommend assessing measurement models before examining complete structural models so as to verify that the latent variable measures are psychometrically sound and that their dimensionality is as expected. Thus, before examining the fit of the complete structural equation model, confirmatory factor analysis (CFA) was conducted to assess the measurement model.

Prior to running the full CFA, independent factor analyses for each latent variable in the model were performed as a preliminary step to identify which indicators were mathematically most robust. Because each latent construct was presumed to be caused by (or indexed by in the case of the Early Family Environment latent variable) a number of indicators and, in general, there was no strong theoretical reason that one indicator would be better than another, running the CFA with this exploratory first step was a good way to identify and eliminate any weak indicators that could affect model parsimony and fit. Furthermore, for complex models, it is recommended to start with simpler models that are subsets of the whole model, in order to be able to uncover potential identification problems more easily (Kline, 2011).

Early Family Environment. A preliminary factor analysis was run with eight indicators underlying the latent construct, removing the two indicators of "Stepparent" and "Parent Death", as stated previously, because of the severely disproportional scores of 0 on these variables (Section 3.2.3). Because of the overlap in the computation of the "Total SES" and "Financial SES" indicators, factor analyses were run with each of these indicators separately. Analysis revealed that the "Financial SES" manifest variable loaded more strongly onto the Early Family Environment latent construct than did "Total SES" (standardized factor loading for Financial SES = .37 vs. Total SES = .20), with no change to the size or direction of the standardized factor loadings for the other indicators. Consequently, "Financial SES" was chosen to be used as the indicator of SES in the model. Table 5 shows the standardized factor loadings for the indicators of the Early

Family Environment latent construct. Kline (2011) recommends that all indicators specified to measure a common factor should have relatively high standardized factor loadings on that factor. Consequently, as is customary in SEM studies, a conservative cutpoint was selected of \geq .45, rather than the .30 cutoff commonly employed in traditional factor analytic studies. As shown in Table 5, only the three Family Structure Indices met this criterion, with the pattern of factor loadings remaining the same as individual indicators not meeting this criterion were removed sequentially, according to the size of their standardized loadings. Consequently, the decision was made to retain only the three "Family Structure Index" measures as the indicators of the Early Family Environment latent construct.

Because there are difficulties identifying measurement models where some factors have only cause (formative) indicators and the composite is latent (such as Early Family Environment), one approach to deal with this issue is to combine the indicators into a single weighted manifest variable (Kline, 2011; MacCallum & Browne, 1993). This was deemed a feasible approach as the three measures of Family Structure across the ages of 0-16 showed a high degree of internal consistency ($\alpha = .80$), and were measured using identical scaling. As a result, the three Family Structure Indices were summed to form a single Family Structure variable, which measured the theoretical stress of participants' family structure across the ages of 0-16. Scores now ranged from 3-12. Furthermore, creating a Total Family Structure Index effectively addressed the skewness and kurtosis issues that were seen in the individual Family Structure Indices, bringing these univariate normality statistics into the acceptable range (skewness = 2.94, SE = .17; kurtosis = 8.47, SE = .35), although as mentioned earlier assumptions of multivariate (and univariate) normality do not apply to exogenous, independent variables. Finally, any potential issues related to multicollinearity, which can influence the results of SEM analyses, were addressed by combining the highly correlated Family Structure Indices (see Table 4 for sizes of correlations) into one.

Sexual Reproductive Strategy. The preliminary factor analysis, with all 13 indicators, log transformed as needed (i.e., SOI Q3, Total Number of Partners, and Typical Relationship Length), revealed the following standardized factor loadings (see

Table 5

Early Environment Indicator	Standardized Factor Loading			
Family Structure Index (0-4)	.69			
Family Structure Index (5-9)	.92			
Family Structure Index (10-16)	.73			
Financial SES ¹	.37			
Family Affection Index	.17			
Family Abuse Index	.27			
Mother's Psychiatric Health	.20			
Father's Psychiatric Health	.08			

Standardized Factor Loadings for Early Family Environment Latent Variable

¹ Reversed score (i.e., higher score = lower SES) used in factor analysis

Table 6a). As can be seen, several measures met the \geq .45 criterion. Interestingly, all the measures of adult, romantic attachment did not meet this cutpoint and showed substantially lower loadings. This was taken as evidence that the attachment measures, although related to the sexual behaviour measures, are not themselves tactics of the sexual reproductive strategy. Importantly, as stated earlier (see Sections 1.9 and 3.2.1), the life history theories are not explicit or in agreement as to whether romantic attachment ought to be considered a component of the sexual reproductive strategy or as a separate but correlated entity. Consequently, on the basis of the CFA results, the romantic attachment variables were extracted for separate consideration (see below).

Table 6b shows the standardized factor loadings following the removal of the 6 adult romantic attachment measures. With the exception of Partnering Status, all indicators loaded above .45. Removing Partnering Status did not change the pattern of factor loadings. Because the correlation between SOI Q3 and Total Number of Partners was large (r = .68), albeit not large enough to suggest issues with multicollinearity, the decision was made to allow the errors associated with these two variables to correlate in the CFA and subsequent structural model. Over a mathematical rationale, there needs to be a theoretical rationale for allowing for such a modification. Given that question 3 on the SOI inquires about the number of partners with whom the participant has had sex on one and only one occasion, the answer to this question should logically be related to the total number of partners (romantic and sexual) a participant has had. No other modifications were made to this latent variable.

Adult Romantic Attachment. Factor analysis, constrained to one factor, was run on the six measures of adult romantic attachment. This was done because although romantic attachment avoidance and anxiety are suggested to be separate dimensions, it was not known how the measures of total attachment insecurity (RQ security) and fearfulness (RQ fearful) would load onto these factors. As shown in Table 7a, the standardized loadings of the two measures of attachment anxiety (ECR Anxiety, RQ Preoccupied) were very low and in the opposite direction to the other variables. This suggested that attachment anxiety was orthogonal to the non-anxious (or more avoidant; please see Section 4.3 for further discussion) attachment styles, and might be better

Table 6a

Sexual Strategy Indicator	Standardized Factor Loading
501.02	71
301 Q2	./1
SOI Q3	.72
SOI Attitudinal Factor	.58
Total Number of Partners	.76
Age at First Intercourse ¹	.47
Typical Relationship Length ¹	.45
Partnering Status	.22
ECR Avoidance	.30
ECR Anxiety ¹	.28
RQ Dismissing	.36
RQ Preoccupied ¹	.19
RQ Security	.21
RQ Fearful	02

Standardized Factor Loadings for Sexual Reproductive Strategy Latent Variable

¹ Reversed score (e.g., higher score = earlier age at first intercourse) used in factor analysis

Table 6b

Standardized Factor Loadings for Modified Sexual Reproductive Strategy Latent Variable

Sexual Strategy Indicator	Standardized Factor Loading
SOI Q2	.68
SOI Q3	.76
SOI Attitudinal Factor	.54
Total Number of Partners	.80
Age at First Intercourse ¹	.46
Typical Relationship Length ¹	.46
Partnering Status	.18

⁻¹ Reversed score (e.g., higher score = earlier age at first intercourse) used in factor analysis

Table 7a

Adult Attachment Indicator	Standardized Factor Loading
ECR Avoidance	.68
ECR Anxiety ¹	18
RQ Dismissing	.22
RQ Preoccupied ¹	07
RQ Security ¹	.86
RQ Fearful	.60

Standardized Factor Loadings for Adult Attachment Latent Variable

¹ Reversed score (i.e., higher score = less anxiety; higher score = less secure) used in factor analysis

Table 7b

81	2
Avoidant Attachment Indicator	Standardized Factor Loading
ECR Avoidance	.70
RQ Dismissing	.24
RQ Security ¹	.84
RQ Fearful	.59

Standardized Factor Loadings for Insecure/Avoidant Attachment Styles

¹ Reversed score (i.e., higher score = less secure) used in factor analysis

Table 7c

Standardized Factor Loadings for Modified Avoidant Attachment Latent Var	riab	le
--------------------------------------------------------------------------	------	----

Avoidant Attachment Indicator	Standardized Factor Loading
ECR Avoidance	74
RO Dismissing + Fearful	.67
RQ Security ¹	.79

¹ Reversed score (i.e., higher score = less secure) used in factor analysis
represented as a separate factor. Removing the anxiety measures and running the analysis again (see Table 7b), revealed high factor loadings for all variables except for RQ Dismissing. Based on the statistical criterion established (a loading \geq .45) the RQ Dismissing indicator was accordingly removed. However, because the dismissing and fearful styles are discussed in the theoretical literature as the two "avoidant" subtypes, a composite RQ avoidance score was created and tested, comprising the sum of participants' dismissing and fearful scores. This was done to avoid losing one of the two indicators of avoidant attachment. Since both indicators originate from the same measure and are scaled identically, summing did not affect the weighting of the items or adversely impact the psychometric properties of the measure (as it might if components of the RQ were combined with the psychometrically stronger ECR subscales). As shown in Table 7c, re-running the factor analysis revealed that the standardized factor loadings were all equally high now. No further modifications were made to the resulting Avoidant Attachment factor.

Regarding the two indicators of attachment anxiety (ECR Anxiety, RQ Preoccupied), issues with non-convergence or improper solutions are known to occur when there are only two indicators per factor (Kline, 2011; Marsh & Hau, 1999). These problems can, however, be circumvented with the use of a single indicator if the measurement error can be accurately estimated (Kline, 2011; Hayduk & Littvay, 2012; Sarkisian, 2010). Consequently, it was deemed preferable to model romantic attachment anxiety using a single indicator. Because the ECR Anxiety subscale has stronger psychometric properties than the one-item RQ, and is preferred over the RQ in analytical situations where a consideration of measurement error is important (Sibley et al., 2005), the decision was made to use the 18-item ECR Anxiety subscale over the single-item RQ preoccupied scale as the observed measure of adult attachment anxiety. Although both scales clearly load onto the anxiety dimension, the single-item from the RQ provides only a global non-specific assessment of participants' perceived level of attachment anxiety, whereas the individual items from the ECR Anxiety scale are more targeted in their focus. Thus, summing the scores to create a single indicator was not considered conceptually appropriate.

Non-Sexual Reproductive Strategy. The preliminary factor analysis was run with all seven indicators. Table 8 shows the standardized factor loadings for the indicators of the Non-Sexual Reproductive Strategy latent construct. As can be seen, "Aggression", "Impulsivity", and "Risk-Taking" were the only three measures that fell above the established cutpoint of .45, with the other indicators displaying considerably weaker factor loadings. Sequential removal of the variables not meeting the \geq .45 criterion did not change the pattern of factor loadings. Consequently only the Aggression, Impulsivity, and Risk-Taking indicators were retained.

Full CFA Model

The full CFA measurement model was run according to the modifications described above. Indicators of all latent variables were specified as continuous. The single indicators of Total Family Structure Index, RAQ Insecurity, ECR Anxiety, 2D:4D, Adult T, and CAG RL were also included in the measurement model. All latent variables were allowed to correlate with the single observed variables and with each other. The specified correlation between the errors of SOI Q3 and Total Number of Partners was also included. The estimation process converged. The measurement model is presented in Table 9.

The model demonstrated a good fit, χ^2 (104) = 169.88, p < .001, Comparative Fit Index (CFI) = .91, Root Mean Square Error of Approximation (RMSEA) = .057 (90% CI = .041 to .072), PCLOSE = .22. The inferential chi-square statistic was significant, which indicates a discrepancy between the population covariances and those predicted by the model. Chi-square is, however, highly sensitive to sample size and can easily become significant even with modest sized samples (Iacobucci, 2010). Consequently, it has been suggested, with some consensus in the literature, that a model demonstrates reasonable fit if the chi-square statistic adjusted by its degrees of freedom does not exceed 3.0 (i.e., $\chi^2 / df \le 3$) (Kline, 2004). In the current model, this criterion equaled 1.6, suggesting an acceptable chi-square fit. With respect to the approximate fit indices of CFI and RMSEA, both statistics met the rule-of-thumb values for inferring good fit (e.g., Bentler & Bonett, 1980; Hu & Bentler, 1999). Furthermore, based on the lower bound of the Table 8

Non-Sexual Strategy Indicator	Standardized Factor Loading
AQ Iotal	.55
Risk Taking	.50
Impulsivity	.91
Dominance	.18
Autonomy	.09
Self Esteem	11
POMS Anxiety	21

Standardized Factor Loadings for Non-Sexual Reproductive Strategy Latent Variable

Table 9

Latent Variable Indicators	Standardized Factor Loadings	Latent Variables	Latent Variable Correlations
ECR Avoidance	.74	Avoidant Attachment and Sexual Strategy	.30
RQ Dismissing + Fearful	.68	Avoidant Attachment and Non-Sexual Strategy	.28
RQ Secure	.77	Sexual Strategy and Non- Sexual Strategy	.28
SOI Q2	.78		
SOI Q3	.59		
SOI Attitudinal Factor	.63		
Total Number of Partners	.64		
Age at First Intercourse	.48		
Typical Relationship Length	.48		
AQ Total	.83		
Risk Taking	.28		
Impulsivity	.62		

CFA with Standardized Factor Loadings and Correlations

RMSEA 90% CI (.041), the close-fit hypothesis was supported, and based on the upper bound of the 90% CI (.072), the poor-fit hypothesis was rejected (Kline, 2011).

In order to further check for sources of model misfit, the sizes of the standardized factor loadings for all indicators specified to measure a common factor were inspected. All factor loadings were significant (p < .001) and were large in size ranging from .48 to .83, which provided evidence in support of the convergent validity of the indicators. Thus, all the latent variables were generally well measured by their respective indicators. The only exception was the risk-taking measure, which had a loading of .28, p < .001. Although this indicator could not be said to explain a large proportion of the variance of the factor it loaded onto (i.e., Non-Sexual Reproductive Strategy), it was kept in the measurement model for a theoretical reason. Risk-taking (along with aggression and impulsivity) is considered a central feature of the Young Male Syndrome (Wilson & Daly, 1985), which Chisholm (1999) believes is the optimal reproductive strategy for males reared in environments of high uncertainty. Consequently, it was deemed important to preserve the risk-taking indicator, in an effort to have a more accurate non-sexual reproductive strategy construct, in spite of its low standardized factor loading.

Misfit was also assessed by looking at the correlations between factors in the measurement model and ensuring than none were excessively high (e.g., > .90 in absolute value). As shown in Table 9, no latent factors were highly correlated with each other. These findings provided evidence of discriminant validity of the factors. In addition, as discussed earlier, there was no evidence of multicollinearity between indicator variables. Thus, this measurement model was used to test the theoretical structural model.

3.2.6. Full Structural Equation Model Analyses

The second step of the two-step SEM analysis was to evaluate the structural components of the conceptual model depicted in Figure 1. Incorporating the changes from the measurement model, the structural model then tested how several androgenic markers (Adult T, CAG RL, 2D:4D) related to attachment and reproductive strategies in men within the context of life history theories.

Latent Variable with a Single Indicator. Prior to running the full structural model, one additional type of modification was made. Recall that the use of single indicator variables within SEM is better when measurement error for that variable can be accurately estimated. Latent variables allow for the isolation of the true score from the inherent unreliability of indicators. Consequently, following an established procedure (Kline, 2011), latent variables for the questionnaire-based single indicators of Total Family Structure Index, RAQ Insecurity, and ECR Anxiety were created. For each observed single indicator, measurement error was estimated by subtracting a measure of the indicator's reliability (i.e., Cronbach's alpha) from 1. In order to determine the proportion of the total variance for each single indicator that was due to measurement error, the computed estimate of measurement error was multiplied by the total variance for that indicator. The error variance for the latent variable was fixed to this value.

Latent variables were not created for the three biological variables in the model. Because the internal consistency estimates were either very high and/or the proportion of error variance was very low for all three biological measures (i.e. Adult T, CAG RL, 2D:4D), it was deemed unnecessary to create latent variables for these single indicators.

Finally, the covariances between certain pairs of exogenous variables were set to zero in order to reduce the number of parameters within the model. In SEM analyses, all exogenous variables are by default allowed to covary with one another. Within the current model this meant that Total Family Structure Index covaried with both 2D:4D and CAG RL, however, it is not physiologically possible for direct correlations between these variables to exist (and they did not exist in the current data), as an individual's biological constitution cannot be systematically related to the composition of their parental family unit. Consequently, only the 2D:4D and CAG RL, were allowed to covary with each other, as this relationship was theoretically plausible.

The resulting structural model, incorporating the above-described modifications, fit the data well, particularly for its size and complexity, as indicated by χ^2 (110) = 186.31, p < .001, CFI = .90, RMSEA = .06 (90% CI = .045 to .074), PCLOSE = .14.

Figure 6 depicts the standardized regression weights for all hypothesized pathways. All significant pathways were significant at p < .05.

Support for the theories of Belsky et al. (1991) and Del Giudice (2009) was found. Consistent with the hypotheses, the Total Family Structure Index (where a higher score is indicative of a less stable family structure) positively predicted Childhood Attachment Insecurity (standardized path coefficient $\beta = .43$), which then positively predicted levels of both Avoidant Attachment ($\beta = .36$) and Anxious Attachment ($\beta = .38$) in adulthood. The adult romantic attachment latent variables each then differentially predicted scores on the Sexual Reproductive Strategy latent variable, in a manner again consistent with Belsky (1997) and Del Giudice's theorizing. Avoidant Attachment positively predicted scores on the Sexual Reproductive Strategy latent variable ($\beta = .35$), while Anxious Attachment negatively predicted scores on the same latent variable ($\beta = .35$).

As hypothesized, the results of the structural model also provided support for androgenic influences on romantic attachment behaviour. Adult T was found to positively predict Avoidant Attachment ($\beta = .20$). This finding is congruent with the possibility that men's circulating levels of T had an activational effect on adult avoidant attachment. In contrast, no relationship was found between Adult T and Anxious Attachment, however, CAG RL positively predicted Anxious Attachment ($\beta = .25$). A positive relationship between these two variables suggests that weaker androgen receptor functionality, as conferred by a longer CAG RL, resulting in lower levels of T activity, may be causally associated with higher levels of adult romantic attachment anxiety.

No support for the hypothesized causal pathways between the Family Structure Index and Adult T ($\beta = .04$, p > .05) or between Childhood Attachment Insecurity and Adult T ($\beta = .08$, p > .05) was found. Thus there was no evidence from the present study that environmental and psychological factors in childhood, albeit assessed retrospectively, influenced individual differences in T levels in adulthood as suggested by the theories of Belsky et al. (1991) and Miller and Pasta (2000). In addition, significant





Figure 6. Structural model. The top part of the figure depicts all significant standardized path coefficients (* p < .05, ** p < .01). The dashed line represents a trend-level pathway (.05 > p < .10). The bottom part of the figure depicts all non-significant pathways within the same model. Ovals represent latent variables. Rectangles represent observed variables. Single headed arrows pointing towards a latent variable represent error.

relationships were not found between 2D:4D and any of the variables it was hypothesized to predict. CAG RL also did not significantly predict individual differences in childhood attachment insecurity ($\beta = .04, p > .05$). The current model thus provided no evidence in support of organizational effects of androgens on either attachment or related reproductive strategies.

The full structural model also provided support for Chisholm's (1999) life history theory of attachment and reproduction. A less stable early family environment, as measured by the Total Family Structure Index positively predicted Childhood Attachment Insecurity ($\beta = .43$), which in turn directly predicted higher scores on the Non-Sexual Reproductive Strategy latent variable ($\beta = .26$). Chisholm does not implicate adult, romantic attachment in his model, thus the failure to find significant relationships between either of the adult attachment latent variables and Non-Sexual Reproductive Strategy was not inconsistent with expectations ($\beta = .06$, p > .05 for Avoidant Attachment; $\beta = .07$, p > .05 for Anxious Attachment). Although no direct relationship between Childhood Attachment Insecurity and the Sexual Reproductive Strategy was found ($\beta = -.05$, p > .05), the "Young Male Syndrome" (Wilson & Daly, 1985) is defined more heavily by the evocative behaviours of aggression, impulsivity, and risk-taking used for the purpose of engaging in opportunistic mating, rather than sexuality alone. Finally, Adult T was a positive predictor of the Non-Sexual Reproductive Strategy ($\beta = .18$), which is in line with Chisholm's theorizing. Chisholm (1999) highlights T as the "mechanism for motivating young men to engage in risky activities that maximize current reproduction". No relationship was found between CAG RL and the Non-Sexual Reproductive Strategy ($\beta = .01, p > .05$).

To investigate further model optimization, all non-significant pathways, save for those in a trend range (i.e., .05 > p < .10), were removed from the model. Such modification made only a very minor improvement to model fit, χ^2 (113) = 183.50, p < .001, CFI = .90, RMSEA = .057 (90% CI = .041 to .071), PCLOSE = .22. All previously significant estimates became significant at *p* < .01 level; trend-level estimates remained in the trend range (see Figure 7).



Figure 7. Structural model run with only significant pathways (* p < .05, ** p < .01). The dashed line represents a trend-level pathway (.05 > p < .10). Ovals represent latent variables. Rectangles represent observed variables. Single headed arrows pointing towards a latent variable represent error.

Indirect Effects. Using the full structural model with all hypothesized connections, specific indirect effects of theoretical interest were evaluated. Because the present study was interested in understanding the role of androgens in causal models of attachment and reproduction, the mediated effects of Adult T on the Sexual Reproductive Strategy and CAG RL on the Sexual Reproductive Strategy were computed. In addition, mediation effects were tested for pathways that supported the various life history theories (i.e., Total Family Structure Index to Non-Sexual Reproductive Strategy and Childhood Attachment Insecurity to Sexual Reproductive Strategy).

The indirect effects were derived through a product of the coefficients for the constituent direct paths from independent variable to intervening variable and from intervening variable to dependent variable (MacKinnon, 2008). Mediation is considered to be present when indirect effects are statistically significant. The significance of the indirect effect was computed using bias-corrected bootstrapping with 1000 resamples (Kline, 2011), an increasingly popular method of testing indirect effects (e.g., Shrout & Bolger, 2002). Bias-corrected bootstrapping is a nonparametric approach that requires fewer assumptions than traditional methods of mediation (e.g., mediated regression analysis) and tends to provide more accurate estimates (Bollen & Stine, 1990; MacKinnon, Fairchild, & Fritz, 2007; Preacher & Hayes, 2008; Preacher & Kelly, 2011; Shrout & Bolger, 2002). In addition to inspecting the *p*-value for a given indirect effect (i.e., p < .05), if the 95% bias-corrected bootstrapped confidence interval (95% CI) does not contain 0 it suggests the presence of mediation (Bollen & Stine, 1990; Preacher & Hayes, 2008; Shrout & Bolger, 2002). Because Amos v. 22.0 does not allow for bootstrapping when a data set includes missing cases that are estimated by the program, mediation analyses were conducted using Mplus v. 7.11. Results of the mediation analyses are summarized in Figure 8.

The bias-corrected indirect effect of Adult T through Avoidant Attachment on Sexual Reproductive Strategy was evaluated using a one-tailed test of significance. The extant literature suggests that the direction of the relationship between basal T and avoidance should be positive, as should the relationship between attachment avoidance and sexual behaviours indicative of a shorter-term mating orientation. Accordingly, a



Estimated indirect effect (ab) = .124 (95% CI = .003 - .246)

Figure 8. Summary of mediation analyses. Mediation analyses of Adult T, CAG RL, and Childhood Attachment Insecurity on Sexual Reproductive Strategy, directly, and indirectly through Romantic Attachment (panel A and B). Panel C depicts the mediation analysis of Total Family Structure Index on Non-Sexual Reproductive Strategy, directly, and indirectly through Childhood Attachment Insecurity.

one-tailed test was considered appropriate. The results of the mediation analysis demonstrated that the constituent paths comprising the indirect effect (i.e., Adult T to Avoidant Attachment and Avoidant Attachment to Sexual Reproductive Strategy) were both in the anticipated positive direction, and each path was significant at p < .01. The indirect effect was significant at p < .05. By contrast, the direct effect of Adult T on the Sexual Reproductive Strategy was non-significant (p > .05), indicating mediation.

For CAG RL, the indirect effect of CAG RL through Anxious Attachment on Sexual Reproductive Strategy had to be evaluated using a two-tailed significance test, as the literature is silent on the importance of the androgen receptor per se. The indirect effect was significant at p = .03, 95% CI [-.130 to -.006], indicating that Anxious Attachment mediated the relationship between CAG RL and Sexual Reproductive Strategy. Again, the direct effect of CAG RL on the Sexual Reproductive Strategy was non-significant (p > .05).

Testing the two specific indirect pathways that supported the theories of Belsky et al. (1991) and Del Giudice (2009) showed that the indirect effect of Childhood Attachment Insecurity through Avoidant Attachment on Sexual Reproductive Strategy was significant at p = .02, 95% CI [.023 to .255], indicating the presence of mediation. The indirect effect of Childhood Attachment Insecurity through Anxious Attachment on Sexual Reproductive Strategy was also significant at p = .01, 95% CI [-.188 to -.022]. The direct effects were non-significant at p > .05.

Finally, the indirect pathway of Total Family Structure Index through Childhood Attachment Insecurity on Non-Sexual Reproductive Strategy, as outlined by Chisholm (1999), was tested for mediation. The indirect effect was significant at p = .04, 95% CI [.003 to .246], indicating that attachment insecurity in childhood mediated the relationship between early family environment and engagement in a reproductive strategy characterized by aggression, impulsivity, and risk-taking.

Subgroup Follow-Up Analysis. Recent evidence in the literature has begun to suggest that the most frequently occurring CAG repeat lengths in the population may

confer stronger androgen receptor functionality compared to short or long repeat lengths (Buchanan et al., 2004; Nenonen et al., 2010a; 2010b). This view differs from the commonly assumed negative linear association between CAG length and androgen receptor function that formed the basis for the present study. To investigate this emerging possibility, exploratory correlations were computed in SPSS.

Review of the frequency distribution of CAG RL in the current sample revealed a mean of 21.80 (SD = 2.66) and a mode of 21. A modal range of 21±1 (20-22) therefore was used, as this encompassed approximately 50% of the sample (N = 94). Pearson correlations were then computed for the modal subsample between Adult T and any variables it was hypothesized to predict. For the latent variables of Avoidant Attachment, Sexual Reproductive Strategy and Non-Sexual Reproductive Strategy, as well as the single indicator Anxious Attachment, factor scores were first computed using the formula: $F_{ji} = \beta_{j1}z_{i1} + \beta_{j2}z_{i2} + ... + \beta_{jk}z_{ik}$, where z = the standard score of each observed variable caused by the latent construct and β = the standardized regression weight of the latent factor on the observed (indicator) variable.

Results showed that the relationship between Adult T and Avoidant Attachment was stronger within the modal CAG group, with r = .31, p < .01. Similarly, the correlation between Adult T and Non-Sexual Reproductive Strategy was larger in size compared to the standardized regression coefficient between these two variables in the SEM analyses, with r = .28, p < .01. By contrast, the correlations between Adult T and Anxious Attachment and Adult T and Sexual Reproductive Strategy remained weak and non-significant (r = .05, p > .05 and r = .07, p > .05, respectively), consistent with the pattern seen in the structural analyses. Overall, these findings support the possibility that a CAG repeat range that covers the lengths most people carry is associated with optimal androgen receptor functionality.

4. General Discussion

4.1. Overview of Research Goals

In the past several decades, researchers have begun to understand individual differences in human reproductive strategies. Life history-based approaches suggest a critical role for attachment processes, in both childhood and adulthood (e.g., Belsky et al., 1991; Chisholm, 1999; Del Giudice, 2009). Attachment insecurity (*vs.* security) has been proposed to lead to early reproductive effort, a short-term mating orientation, and associated characteristics of increased aggression, impulsivity, and risk taking, among adult men. Furthermore, theories by Belsky (1997), as well as Del Giudice (2009), have hypothesized differential reproductive consequences of the different insecure attachment styles. In particular, avoidant patterns have been associated with short-term, opportunistic, uncommitted mating, and are suggested to be characteristic of insecure men (Del Giudice, 2009).

These theorists, as well as Miller and Pasta (2000), have also speculated about the role of sex hormones within LHTs of early environment, attachment, and reproduction, with hormonal effects suggested to occur prenatally and/or in adult life when reproductive behaviours are of direct relevance (Belsky, 1997). The biological plausibility of this suggestion is based on a large body of literature showing that androgens in particular, and especially testosterone, are capable of influencing the structure and/or neurochemistry of the brain at these same time points (Breedlove & Hampson, 2002; Wallen, 2009), with exposure during critical periods in early development having enduring consequences for behavioural predispositions. Chisholm (1999) specifically implicates testosterone as a causal factor for increased male engagement in risky activities that adaptively maximize current reproductive opportunities (i.e., the "Young Male Syndrome"), while Belsky et al. (1991) and Miller and Pasta (2000) discuss the role of pubertal maturation, characterized in males by a significant increase in androgen concentrations, in the development of reproductive strategies. Because puberty ushers in adult-like levels of testosterone, it opens the door to a new class of hormonal action, the so-called activational effects of the hormone (Eckel

et al., 2008). Activational effects are not possible in childhood when testosterone production is negligible, because they involve changes in the CNS that are sustained by the ongoing current presence of a hormone. The fact that the avoidant style is hypothesized to be a "male-biased" strategy is consistent with studies that have found that on average, men have higher attachment avoidance and lower attachment anxiety than women (e.g., Del Giudice, 2011; Bartholomew & Horowitz, 1991; Scharfe & Bartholomew, 1994; Schmitt et al., 2003; Brassard et al., 2007). A possible proximate mechanism underlying the sex difference may be differential levels of sex hormones, including androgens.

The goal of the present research was to formulate and empirically test a LHTbased model of attachment and reproductive strategies in men, integrating the potential effects of androgens. Support for the basic tenets of a LHT-based model would be demonstrated by finding that a less stable early family environment positively predicted childhood attachment insecurity, and that in turn attachment insecurity positively predicted a more opportunistic reproductive strategy in adulthood, characterized by a shorter-term mating orientation and more avoidant and less anxious adult romantic attachment style. The relationship between testosterone and attachment was investigated by incorporating measures of adult testosterone, as well as an indirect estimate of prenatal differences in testosterone (2D:4D ratio), and of androgen receptor sensitivity (indexed by CAG repeat length) into the model in an effort to evaluate whether any association with androgens was supported (and if so, whether effects occurring during very early development or in adulthood were implied). Although the LHTs are neither uniform, nor very specific, about the *timing* of androgenic effects, the testosterone-related indices chosen in the present study had the potential to provide insights into this important question.

The discussion below is organized as follows. I will begin by summarizing the overall findings from the research study. Following this, I will discuss the pattern of results obtained within the context of the existing theory and literature. I will then discuss the role of androgens within the LHT-based model, and what, if any, type of effect androgens have on attachment-related processes in men. The chapter will conclude

112

with a discussion of the limitations of the present research, followed by a review of the study's contributions and some directions for future research.

4.2. Summary of Research Findings

The present research represents the first empirical study to test and support a LHT-based model of attachment and reproductive strategies in young men. To my knowledge no studies before it have investigated the role of attachment processes within evolutionary-developmental models of reproduction. Structural analyses demonstrated that a less stable family structure in childhood positively predicted childhood attachment insecurity (assessed retrospectively), which then positively predicted a more opportunistic sexual reproductive strategy was negatively predicted by levels of adult anxious attachment. These findings highlight the differential significance of adult romantic attachment styles on the development of men's reproductive strategies, and support the theoretical models proposed by Belsky (1997) and Del Giudice (2009), who have hypothesized that avoidant attachment is associated with opportunistic, uncommitted mating behaviours. The findings also provide new evidence that within men, anxious attachment may be associated with more investment-oriented and less opportunistic sexual behaviours.

Another novel contribution of the present study was that it demonstrated that childhood attachment insecurity mediated the relationship between quality of early family structure and engagement in non-sexual evocative behaviours (i.e., Young Male Syndrome) believed to be associated with a more opportunistic male reproductive strategy (Chisholm, 1999). This finding again highlights the centrality of attachment processes in the development of mating strategies in men. Furthermore, it provides the first empirically-based support for Chisholm's model (1996, 1999), which focuses on the role of *childhood* attachment configurations in the expression of reproductive strategies in adulthood. Support for the integration of androgenic factors within a LHT-based model of attachment and reproduction was also demonstrated—the first empirical evidence of its kind. Levels of testosterone in adulthood were found to independently and positively predict levels of avoidant attachment, as well as the non-sexual reproductive strategy. Integration of testosterone into the model was further expressed by the identification of avoidant attachment as a mediator variable in a significant pathway connecting adult testosterone levels to the sexual reproductive strategy. In addition, weaker androgen receptor responsivity to testosterone predicted higher levels of anxious romantic attachment and anxious attachment was found to be a significant negative predictor of a short-term reproductive strategy. Mediation analysis showed that anxious attachment was a significant mediator in a pathway connecting responsivity to adult testosterone levels and the sexual reproductive strategy. Thus, adult romantic attachment was found to be an important mediator of associations observed between levels of androgen activity and the sexual reproductive strategy.

Overall, the findings from this study highlight the pivotal role that attachment processes play in evolutionary-developmental models of mating in men.

4.3. The Integrated Life History Theory-Based Model

I will now turn to a more complete interpretation of the core findings from the structural analyses, within the context of the LHT framework. The core set of relationships, as described in Section 4.2, is summarized schematically in Figure 6. As can be seen in the Figure, childhood attachment insecurity had three separate, direct effects: on adult anxious attachment, adult avoidant attachment, and non-sexual reproductive strategies (as represented by higher levels of aggression, risk-taking, and impulsivity). In other words, men's retrospective self-reports of their level of attachment insecurity to caregivers in childhood seems to predict degree of self-acknowledged attachment insecurity in adulthood, as well as predicting engagement in evocative non-sexual behaviours characteristic of opportunistic mating.

Evidence of temporal continuity of insecure attachment patterns is consistent with certain LHT-based models of attachment and reproduction (i.e., Belsky, 1997; Del Giudice, 2009) that suggest that early attachment prototypes should influence the development of corresponding adult, romantic attachment styles, which are used to guide reproductive strategies. As proposed by Bowlby (1982), attachment relationships formed early in life represent internal working models of individuals and their social worlds that can be stable across significant portions of the lifespan. Empirical study has supported this claim, finding stability in insecure attachment patterns from infancy to young adulthood (e.g., Hamilton, 2000; Fraley, 2002), and that exposure to negative life events over the developmental course seems to increase the stability of attachment insecurity (Hamilton, 2000). Furthermore, and consistent with the current results, retrospective studies have found overlap in levels of attachment security to parents in childhood and to romantic partners in adulthood (e.g., Hazan & Shaver, 1987; Levy, Blatt & Shaver, 1998, J. Steele as cited in Fraley, 2002). These findings support the view that attachment is an enduring trait that shapes the kinds of interactions a person experiences and facilitates personality stability within a given environment (Fraley, 2002).

Consistent with this view, confirmatory factor analysis in the current study did not find that measures of romantic attachment style loaded strongly onto the sexual reproductive strategy latent variable, but instead suggested that they may represent separate, yet correlated entities (i.e., avoidant and anxious attachment) which *predict* the sexual reproductive strategy. As Hill and colleagues (1994a) have suggested, romantic attachment style may act as a mediator between early environmental factors (including childhood attachment) and adult reproductive strategy (which was empirically supported in the current investigation). Identifying that romantic attachment predicts reproductive strategy is of relevance, as it helps to clarify the ambiguities about where romantic attachment fits within LHT-based models (i.e., Belsky, 1997; Del Giudice, 2009). It is important to note, however, that the existence of stability of attachment prototypes into adulthood does not preclude the possibility that men may adjust their reproductive strategies later in development (Del Giudice, 2009). In fact, biological factors, including sex hormones, may contribute to adaptive reproductive plasticity in adulthood (Del Giudice, 2009). More is said about this in Section 4.4.

Within the current model, romantic attachment avoidance was not a single indicator, but rather it was a component of a multi-indicator latent construct. As outlined in Section 3.2.5, CFA found that the indicators of ECR Avoidance, RQ Dismissing-Avoidance, RQ Insecurity, and RQ Fearful loaded most strongly together, while the two measures of anxious attachment only weakly loaded onto the same factor. The finding that overall attachment insecurity (computed by reverse coding the scores on the RQ Secure Attachment scale) was more closely associated with attachment avoidance, than anxiety, has been supported in the literature. In a factor analytic study of several romantic attachment measures, Brennan and Shaver (1995) identified an Insecurity factor, defined by high loadings on self-reliance, frustration with partners, proximityseeking (in a negative direction), and low loadings on jealousy and anxious clinging to partners, and a *Preoccupation with Attachment* factor, defined by high loadings on anxious clinging to partners, jealously, frustration with partners, and low loadings on self-reliance. The insecurity factor was found to correlate strongly and positively with attachment avoidance (.67), as measured by Hazan and Shaver's (1990) rating-scale of attachment style, and not at all with anxiety (.03), while the Preoccupation factor correlated positively with anxious-ambivalence (.59), and not at all with avoidance (.09). Similarly, Simpson (1990) factor analysed the 13 items of Hazan and Shaver's (1987) attachment measure and found that 2 factors emerged: a secure-avoidant dimension (made up of the 5 secure and 4 avoidant items, and reflecting level of insecurity/avoidance), and an anxious-nonanxious dimension (made up of the 4 anxious items, and reflecting level of anxiousness).

As introduced in Section 3.2.1 (see Appendix D), fearful attachment has been described as a special case of dismissing-avoidance (what has been referred to here as avoidance), in which the deactivating defences characteristic of dismissing-avoidant individuals (i.e., lack of proximity-seeking) are not fully operative (Shaver & Mikulincer, 2002). Consequently, fearfulness is considered to be inherently linked with dismissing-avoidance, and fearful individuals remain capable of employing behavioural strategies

116

consistent with deactivation (Shaver & Mikulincer, 2002). As such, it is conceptually appropriate that levels of fearfulness loaded more strongly on the avoidant than the anxious attachment latent variable.

Romantic attachment configurations were found to directly and differentially predict men's sexual reproductive strategy, such that avoidant attachment *positively* predicted men's sexual behaviour and anxious attachment *negatively* predicted sexual behaviour. These results corroborate the existing empirical evidence that suggests that romantic attachment avoidance is associated with more opportunistic sexual behaviour (e.g., Simpson, 1990; Brennan & Shaver, 1991,1995; Feeney & Noller, 1992; Paul et al., 2000; Schmitt, 2005; Jackson & Kirkpatrick, 2007, Schindler et al., 2010, DeWall et al., 2011), while anxious attachment may be associated with behaviours indicative of less interest in short-term mating (e.g., Gentzler & Kerns, 2004; Kirkpatrick & Davis, 1994; Schmitt, 2005, Jackson & Kirkpatrick, 2007). Furthermore, the current findings are in support of lifespan attachment models (i.e., Belsky, 1997; Del Giudice, 2009), which suggest that avoidance in men is associated with short-term, uncommitted mating.

Interestingly, a double-dissociation in how child and adult attachment predicted sexual and non-sexual reproductive strategies was found. Attachment insecurity in childhood directly predicted the non-sexual reproductive strategy, and not the sexual reproductive strategy, while adult romantic attachment configurations directly predicted the sexual reproductive strategy, and not the non-sexual reproductive strategy. Because juveniles cannot reproduce, it has been suggested that the adaptive consequence of insecure attachment in childhood may be to increase male coercive tactics (i.e., increased aggressiveness and impulsiveness) that characterize alternative reproductive strategies (Chisholm, 1996; 1999). By contrast, attachment insecurity early in life only predicted adult sexual behaviour through men's adult romantic attachment configurations. This finding supports the notion proposed by Del Giudice (2009) that it is *adult* attachment that takes on the mature function of regulating relationships and sexual mating strategies. Empirically, a substantial literature base exists to support the association between adult, romantic attachment and sexuality (e.g., Simpson, 1990; Brennan & Shaver, 1991,1995; Feeney & Noller, 1992; Paul et al., 2000; Gentzler & Kerns, 2004; Schmitt, 2005;

Jackson & Kirkpatrick, 2007, Schindler et al., 2010). At present, however, no research has been conducted on the association between childhood attachment patterns and adult sexual behaviour. Although the lack of such study within a rich and long-standing research traditional like child attachment is intriguing, and raises questions about researchers' beliefs about the existence of a direct relationship between these variables, it remains possible that future studies, if conducted, could find a significant association between early attachment and later sexual behaviour. Nevertheless, the current findings reconcile Chisholm's theory (1996; 1999), which discusses the consequences of *childhood* insecurity/security on later reproductive behaviour, with those of Del Giudice (2009) and Belsky (1997), which highlight the relevance of attachment in adulthood in the development of sexual behaviour preferences. Overall, the findings from the model highlight the importance of attachment processes, in both childhood and adulthood, in the development of male reproductive strategies.

LHTs of attachment and reproduction hypothesize that early childrearing experiences should predict the development of corresponding psychological attachment styles in childhood. In support of this, current results showed that less stable, and supposedly more stressful, family structure from birth to age 16 years, predicted greater levels of childhood attachment insecurity. Early family composition itself, however, did not directly predict sexual or non-sexual reproductive strategies. Draper and Harpending (1982) posit that early experiences concerning fathering and marital relationships influence the reproductive strategies that individuals develop in adulthood. While there exists some limited evidence of an association between father absence during early life and increased sexuality in men in young adulthood (e.g., Alvergne et al., 2008; Kim & Smith, 1998), most studies focus only on women, and James and colleagues (2012) have suggested that the unique role of father presence/absence on sexual development (i.e., paternal investment theory, see Ellis (2004)) may be more specific to women than men. Furthermore, the total family structure index included in the current study was not purely a measure of father presence/absence, and instead assessed family composition (and associated stress) from infancy to adolescence, which could help to account for the lack of relationship with mating strategy. A less father-centric perspective is consistent with

the theories proposed by Belsky et al. (1991), Chisholm (1999), and Del Giudice (2009), which suggest that the stress of the early family environment (which may derive in part from family composition) affects reproductive strategies via the attachment system (also see Hill et al., 1994a). In support of this, Miller and Pasta (2000) found that various measures of early family stress (e.g., level of family affection, exposure to abuse, father absence) failed to predict self-reported levels of aggression in young adult men.

4.4. The Role of Androgens Within an Integrated Life History Model

The most novel contribution of this study is that it provides some of the first empirical evidence in support of a predictive association between androgen action and romantic attachment in young adult men. Circulating levels of testosterone (T) in adulthood were found to positively predict an avoidant form of attachment. Higher T predicted a more avoidant approach. In addition, the length of the CAG repeat sequence in the androgen receptor (AR) gene, which has been suggested to be inversely related to the degree of responsivity of the AR to T following binding (e.g., Choong et al., 1996; Ding et al., 2004), predicted anxious attachment. In other words, lower activation of genomic processes by T at the receptor level was found to predict greater anxious attachment.

Although this study is the first to systematically investigate the link between T activity and adult romantic attachment, these results are supported by a number of related observations. Previous studies have shown that romantically partnered men have lower circulating T levels compared to single men (e.g., Burnham et al., 2003; Gray et al., 2004), and that men's T levels may remain stable even with changes in partnering status (i.e., low T men are more likely to become and stay partnered than high T men; van Anders & Watson, 2006). Across several cultures, the level of caregiving that fathers provide their children, as assessed by self-report, partner-report, or interview, has been found to be negatively correlated with T levels (Gettler et al., 2011; Alvergne et al., 2009; Muller et al., 2009; Kuzawa et al., 2009). One investigation has reported a reduction in T levels subsequent to new fatherhood (infants aged 1 month or less) (Gettler et al., 2011). Weisman, Zagoory-Sharon, and Feldman (2014) recently demonstrated that lower basal T

levels in fathers of infants were associated with greater engagement in father-infant bonding behaviours (i.e., affectionate touch, "motherese", positive affect, and gaze towards infant), as coded by the experimenters. Collectively, these findings are consistent with the hypothesis that lower T in men may promote more committed, nurturing behaviour (Gettler et al., 2011). Finally, in an investigation of young men's interpersonal styles, adult T levels were negatively correlated with degree of communion, assessed in part by levels of attachment-related avoidance (Turan et al., 2014).

Low T has been associated in other studies with higher levels of internalizing symptoms including anxiety (e.g., Granger et al., 2003) and depressive symptoms (e.g., Sankar & Hampson, 2012; Schmidt et al., 2004). This converges with studies showing that romantic attachment anxiety is related to higher levels of self-reported depression, as well as state and trait levels of anxiety, compared to the avoidant style (e.g., Simonelli et al., 2004; Williams & Riskind, 2004). Within the current study too positive correlations were found between self-reported trait anxiety and measures of romantic attachment anxiety. The present study, however, is the first to show a link between a genetic marker of T-related activity and level of attachment anxiety.

Although not explicitly tested until now, several life history theorists have speculated about the possible role of T in evolutionary-developmental models of attachment and reproduction. For example, accelerated pubertal timing, a variable that is a proxy for the relative levels of circulating T in males, has been hypothesized to have consequences for sexual behaviour profiles (i.e., Belsky et al., 1991; Miller & Pasta, 2000). In addition, Chisholm (1999) views T as part of the mechanism for motivating young men to engage in behaviours that maximize current reproduction, based on the fact that T is implicated in each of the core features of the Young Male Syndrome. The current findings demonstrated that adult T directly predicted the non-sexual reproductive strategy, in line with Chisholm's (1999) proposition. Furthermore, in the present study androgen variables directly predicted romantic attachment style, and via the mediation of romantic attachment, also predicted the sexual reproductive strategy. This speaks to the centrality of attachment processes in the development of sexual behaviours/reproductive orientations in men. As Del Giudice (2009) suggests, it is attachment in adulthood that takes on the mature function of regulating relationships and sexual mating strategies, and it is possible that increased androgen action may contribute to reproductive plasticity in adulthood by serving to amplify romantic attachment-dependent effects on mating strategies.

Of the limited empirical research conducted on life history models of mating and reproduction, no studies to date have assessed attachment either in childhood or romantically in adulthood. Based on the current findings, it appears that attachment configurations are indeed important in linking early family experience with reproductive behaviour in adulthood, and for accounting for how androgens are integrated within LHT-based models. In fact, the mediating role of romantic attachment may help to explain why existing evolutionary-developmental studies have not found a significant predictive association between pubertal timing, a variable that is a proxy for the relative levels of circulating T, and later sexual behaviour in males (i.e., Meckelmann et al., 2013; James et al., 2012). No evidence was found in this study, however, to support a direct predictive relationship from childhood attachment (or early family structure) to adult T, as might be expected by Belsky et al. (1991) or Miller and Pasta (2000). Although consistency would be expected between an individual's levels of pubertal T and adult T (e.g., Gesquiere et al., 2005), it is conceivable that stronger associations may have been seen presently if we had used a measure of T taken at puberty. Nevertheless it is possible, and perhaps more plausible, that T concentrations in adulthood independently predict attachment-related behaviour, once reproduction and mating become relevant concerns, as the current analyses suggest.

Although romantic attachment mediated the relationship between T or CAG RL and the sexual reproductive strategy, a direct predictive association between the androgen variables and the same latent construct was not found. This is somewhat unexpected given that previous literature has generally found positive associations between adult T, measured in either serum or saliva, and engagement in sexual behaviour, assessed experimentally or in correlational studies (e.g., van Anders & Goldey, 2010; Alvergne et al., 2009; Peters et al., 2008; van Anders et al., 2007; Schmidt et al., 2004). It is doubtful that the reason for the lack of relationship is due to T-related measurement error; the mean and standard deviation for adult T in the present work were comparable to those found in most other studies of young adult men, and the internal consistency estimate showed a high reliability between the two T measurements obtained during testing.

Given that many of the indicators of the sexual reproductive strategy were single items that depended on self-report (e.g., total number of partners, typical relationship length, age at 1st intercourse), it is possible that the inherently weaker reliability of single item or limited item measures could have reduced the size of the associations. Stronger relationships may have been seen if multi-item measures were used as indicators. In addition, the results from recent studies investigating the relationship between T and the SOI (the source of several of our indicators) are mixed. For example, several studies have found that SOI scores are positively correlated with T in partnered but not single men (e.g., Puts et al., 2015; Edelstein et al., 2011; McIntyre et al., 2006). Given that nearly half of the study sample (N = 98) self-identified as single, this may have weakened the associations. Other studies have not found significant associations between T and the SOI (e.g., Charles & Alexander, 2011; van Anders et al., 2007). Although age at first intercourse was not significantly correlated with adult T levels, it was significantly correlated with CAG repeat length, in the anticipated direction (i.e., older age at first intercourse associated with longer CAG repeat sequence). It remains, however, that the results at this level of the model are weakened by the fact that the expected association between T and the sexual reproductive strategy was not found.

Adult T did, however, positively predict the non-sexual reproductive strategy, which is consistent with much existing literature on adult T and externalizing behaviours in men that points to a positive correlation between circulating T levels and aggression and risk-taking (e.g., Archer, 1991; White et al., 2006). The direction of association between adult T and impulsivity is less clear, partly due to the limited number of studies conducted on this topic in human males, although some studies suggest a positive correlation (e.g., O'Connor et al., 2002; Schmidt et al., 2004). Chisholm (1999) proposed that the so-called Young Male Syndrome, defined by the hypermasculine behaviours of increased aggression, risk-taking, and impulsiveness, is caused, in part, by T, which he viewed as the "drug" for engaging in risky activities. Within the current data, the

individual correlations between adult T and aggression, risk-taking, and impulsivity were all in the expected positive direction, although stronger for aggression and impulsivity than for risk-taking. The relationship between adult T and risk-taking, as assessed by paper-and-pencil measures, is not as well established in the literature, with most studies tending to assess risk-taking through laboratory-based gambling tasks. Such measures were not employed in this study because of the interest in assessing risk-taking more globally, rather than restricted to the financial domain. Nevertheless, such differences in the constructs assessed may help to explain the weaker association between risk-taking and adult T seen in the current data. Interestingly, when correlations between adult T and the non-sexual reproductive strategy were conducted within the *modal* CAG subgroup (see Section 3.2.6), comprising the range of repeat lengths that recently have been speculated to confer the strongest androgen receptor functionality (Buchanan et al., 2004; Nenonen et al., 2010a; 2010b), the association became stronger (r = .28), providing additional support for the role of adult T in the display of non-sexual evocative behaviours in men. Finding the expected relationship between adult T and the non-sexual reproductive strategy makes it all the more surprising that no significant association was found between adult T and sexual behaviours. The fact that the measures used to assess aggression, risk-taking, and impulsivity were made up of multiple items (range of 16 to 34 items), again reinforces the possibility that the better reliability of multi-item measures may help to strengthen the size of the associations with adult T.

No direct association was found between CAG repeat length and either the sexual or non-sexual reproductive strategy. To date, only limited research has been conducted on the relationship between CAG and androgen-related psychological traits, and the existing data are mixed (e.g., Jönsson et al., 2001; see Section 1.7). Conceivably, weaker relationships for CAG could reflect the fact that CAG length must be considered in conjunction with individual differences in adult T concentrations, as the receptor does not act on its own. However, in a different context, CAG recently has been found to independently predict symptoms of depression in men (Sankar & Hampson, 2012) or other traits (Hampson & Sankar, 2012a), showing that such relationships are theoretically possible. Nevertheless, further study of the association between CAG repeat length and

externalizing behaviours, as well as sexual behaviours, in men is merited in order to understand the nature of these relationships.

Recall that T is known to exert its effects on the brain by binding to ARs located in neurons throughout the central nervous system, and that the neuronal effects of such binding can occur either during fetal or neonatal development or during adult life (Rubinow & Schmidt, 1996; Swerdloff et al., 1992). These are regarded as two different classes of T-induced effects, because the types of changes induced in the nervous system during early brain development and in the adult brain are quite different. As such, the effects of androgens may be organizational, whereby the actions of T occurring as a result of exposure during specific windows of sensitivity in early brain development *permanently* alter the structural organization or functional potential of the nervous system (Phoenix et al., 1959; Breedlove & Hampson, 2002; Wallen, 2009), and/or activational, referring to reversible changes in brain function brought about by current levels of circulating T acting upon neural substrates that may or may not also be organized by T exposure early in life (Eckel et al., 2008). The activational effects of T often consist of changes at the neurochemical level, such as changes caused by T in the synthesis or metabolism of particular neurotransmitters, or changes in the numbers of neurotransmitter receptors present in defined regions of the brain. The T-related findings in the current model were consistent with the possibility of an activational effect, whereby circulating levels of T in adulthood independently influence romantic attachment avoidance and in turn the sexual reproductive strategy, as well as influencing the expression of non-sexual externalizing behaviours.

It should be noted that the neural mechanisms that underlie attachment are expected to be exceedingly complex, like any higher-level integrated pattern of behaviours, and likely includes affective, motivational, and cognitive dimensions that cannot be pinpointed to a particular part of the brain. Contemporary studies are only just beginning to identify pathways in the brain that are involved in romantic attachment (for review see Coan, 2008). The present study points toward the possibility that adult T can modify one or more processes involved in romantic attachment but do not permit any insights into which particular processes are regulated by T.

The 2D:4D digit ratio, the proxy measure of individual differences in prenatal androgen exposure used in the present study, was not significantly associated with any of the attachment-related variables, nor did it significantly correlate with any of the behavioural characteristics it was hypothesized to relate to in the model. A smaller ratio has been argued to reflect greater T exposure *in utero*, when bone formation is laid down in the digits (Manning et al., 1998). The lack of associations is somewhat unexpected, as previous studies have found significant correlations between 2D:4D and some sex-related variables, as well as externalizing behaviours such as aggression and risk-taking (see Section 1.6.3 for a detailed review). Furthermore, null findings in this study occurred in spite of the high reliability of the 2D:4D measurement and in spite of finding a typical mean ratio very consistent with other published reports (e.g., Maner et al., 2014; Medland et al., 2010). An accumulating body of research, however, has begun to point to the possible fallibility of the 2D:4D finger length ratio as a valid marker of fetal T levels among *healthy* humans, even though an association with T has been found in clinical populations where variations in fetal T exposure are more extreme (e.g., Hampson & Sankar, 2012b; Breedlove, 2010; Berenbaum et al., 2009; McIntyre, 2006). It is possible that the 2D:4D ratio is not sufficiently sensitive in healthy individuals (for example, differences in the level of T action may not always be overtly reflected in the ratio), and correlations observed may be hard to replicate across studies because of the low signalto-noise ratio associated with 2D:4D itself. For these reasons, the present data do not rule out the possibility of an organizational influence of prenatal T on attachment, as broadly speculated by Del Giudice (2009), even though no significant relationships were found in the present study. To investigate this question, future studies should incorporate a more refined and undisputed measure of prenatal T concentration, such as a direct measure of T taken from mid-trimester amniotic fluid.

The model findings revealed that T and CAG differentially predicted avoidant and anxious attachment; T positively predicted avoidant attachment, while CAG length positively predicted anxious attachment, suggesting that testosterone's influences on romantic attachment style in men may reflect either differences in adult T levels or differences at a receptor level. Interestingly, T and CAG were not significantly correlated with one another, compatible with a number of previous reports suggesting they vary independently (Hampson & Sankar, 2012a; Goutou et al., 2009; Krithivas et al., 1999). In humans, most effects caused by T are thought to be mediated by T binding to the AR. Therefore possessing a less sensitive receptor ought to simultaneously be associated with higher anxious attachment, but also might be expected to attenuate the heightening effects of adult T on avoidance. In this respect, the effects of adult T on avoidant attachment could potentially be more potent than what is revealed by the model.

In considering why we did not find a significant path between CAG polymorphism and avoidant attachment, there is another possibility that must be considered. Although the possibility is more hypothetical, it is not altogether impossible that sex hormonal influences on the anxious vs. avoidant dimensions could occur through different receptor mechanisms. Data from laboratory animals has shown that under some circumstances T can also exert its effect through metabolic conversion to its aromatized metabolite 17β -estradiol. While this pathway has proven to be important in some species, evidence is slim that the 'aromatization route' is important in humans. Nevertheless, there is still a possibility that adult T might act to affect avoidant attachment via metabolic conversion to estradiol (with subsequent binding to the estrogen receptor), potentially explaining why no association was seen for CAG RL (polymorphism in the AR). In the present data we did find a significant association between anxious attachment and AR repeat length, which suggests that AR is involved. It is for the other form of adult romantic attachment, avoidant attachment, that our data could be taken to imply that a separate mechanism exists. Nevertheless, the overall importance of estradiol conversion as a mechanism by which T affects behaviour in men remains relatively unexplored, and represents an important direction for future investigation.

4.5. Limitations

Sample. Participants for this study were recruited from a university campus and the sample was primarily comprised of students. As mentioned in Section 2.1, university populations have been used in a multitude of studies assessing romantic attachment (e.g.,

Edelstein et al., 2010; Gormley & Lopez, 2010; Gillath et al., 2008; Gentzler & Kerns, 2004; Schachner & Shaver, 2004), and thus our recruitment sample was conventional for this type of research. One possible limitation of using a post-secondary student sample is its potential to show a restriction in range across certain demographic variables (e.g., SES, parent education). There is, however, evidence that Canadian university populations are demographically diverse (e.g., Clifton et al., 2008; Goonewardena, Rankin, & Weinstock, 2004), and inspection of the descriptive statistics from the current study revealed substantial range in scores across a number of the measures of interest (see Table 2 and Table 3). Nevertheless, it remains true that a university sample may be less diverse than the general population, and therefore could influence the scores seen on relevant parameters, including those pertinent to the development of attachment styles (e.g., SES, family composition). If anything, the use of a student sample may have weakened the correlations seen; that being said, significant correlations were still found in the sample studied here. Conclusions as to how the present findings may generalize to the wider population should be made with caution. Expanding recruitment into the greater community may be a beneficial next step.

Another potential limitation relates to the fact that our East Asian participants had to be excluded from the final analysis. This resulted in the reduction of our total sample size by 45 participants. In may be beneficial for future studies to test the integrated-LHT model within a larger subsample of East Asian men, preferably in their native community. This could also help to address the potential validity issue related to linguistic proficiency by allowing participants to complete self-report questionnaires in their native language.

Measurement Issues. Although the timing of pubertal maturation (advanced or delayed) has been speculated to connect early environmental experience with later relationship behaviour and choices related to mating (i.e., Belsky et al., 1991; Miller & Pasta, 2000), pubertal timing was not directly assessed in the present study. Because pubertal development in males lacks a salient and discrete event comparable to menarche in females, retrospective reports of the timing of puberty onset in men are susceptible to inaccuracy, and are not empirically well-established (Mustanski et al., 2004). In fact,

Dubas, Graber, and Petersen (1991) found only chance levels of stability in adolescents' self-reports of perceived pubertal timing, and cautioned readers that perceived timing should not be interpreted as actual timing. Given that males who mature earlier maintain higher circulating levels of T into adulthood compared with males who enter sexual maturity at a relatively later age (Gesquiere et al., 2005), individual differences in adult T concentrations are likely to reflect relative differences in pubertal timing, and have the advantage of being more objective than retrospective self-reports. As mentioned earlier, however, predictive associations from early family structure or childhood attachment to adult T were not found in the current study. Studies using prospective, *in vivo*, measurement of pubertal timing and associated T concentrations would provide the best and most accurate test of the role of pubertal maturation within LHT-based models of attachment and reproduction.

All participants in the present study were recruited and tested in young adulthood, and therefore measurement of attachment to parents/caregivers in childhood had to be retrospective. Few retrospective measures of attachment exist. The RAQ (Parkes, 2006) was selected because is the only measure that assesses attachment history according to the patterns of attachment described by Ainsworth and colleagues (1978), and is the only measure available to be used in more normative and less pathological samples (Crowell et al., 2008). It is, however, a relatively new measure that has not been widely studied in the literature, and its lack of procedure for how to handle missing data due to parental absence (i.e., single parenthood) needs to be formally addressed (although I did outline a sound method for addressing this issue; see Section 2.3.1). As more psychometric and substantive research is done on the topic of attachment history it is possible that more comprehensive retrospective measures of attachment will be developed, and their validity demonstrated. Nevertheless, one limitation of all retrospective reports, within the attachment domain or not, is that memory for or interpretation of past events has the potential to become distorted with increasing time lag since the occurrence of the event/experience (e.g., Baldwin & Fehr, 1995). Unfortunately, this is a limitation that cannot be circumvented without the use of a longitudinal design that would allow for parent-child attachment and other early life experience variables to be assessed directly in childhood. It bears noting though that following children into young adulthood to study the development of their adult romantic attachment styles would be a rather difficult undertaking given the time spans involved.

Specific items from Part I of the RAQ were used to compute indices of Family Affection and Family Abuse. The nature of the items used to assess these constructs was consistent with those used in other studies to retrospectively assess early family environment in men (particularly for the abuse domain) (e.g., Miller & Pasta, 2000; Kim & Smith, 1998). Nevertheless, the estimates of internal consistency obtained for the Family Affection Index and the Family Abuse Index were low ($\alpha = .58$ and $\alpha = .48$, respectively), suggesting that the items comprising each of these indices may not be very closely related. Although employing more psychometrically robust measures might have been beneficial, the availability of measures, particularly in the affection domain, represents a challenge. It appears that few retrospective measures assessing perceived quality of early family life exist. For example, while Meckelmann et al. (2013) used the Relationship Context Scale (Scheffer et al., 2000) to retrospectively assess quality of parents' marital relationship and quality of child's maternal relationship, this scale has not been validated in English speaking populations.

4.6. Study Contributions and Directions for Future Research

Overall, the present study has begun to bridge the independent research traditions in attachment and neuroendocrinology, and has made novel contributions in each domain. I tested a life history-based model of reproductive strategies in men, and results provided some of the first empirical evidence linking childhood and romantic attachment configurations to differential mating strategies. No previous studies have assessed attachment within evolutionary-developmental models, and the findings highlight that both the level of retrospective attachment insecurity to parents/caregivers in childhood, as well as style of attachment to romantic partners in adulthood influence the degree to which men align with a short-term, opportunistic mating approach. The results confirm the speculated, but to date directly untested, view that romantic attachment behaviours are modulated, in part, by androgens. These findings are theoretically important because they help to re-formulate the general understanding of the etiology of romantic attachment in men from a purely learned construct to one that is also hormonally-based. Adult T itself, and androgen function at the receptor level, predicted avoidant attachment and anxious attachment, respectively, and importantly, romantic attachment configurations mediated the relationship between androgenic variables and sexual behaviour. In addition, and consistent with expectation, the levels of adult T were a positive predictor of the evocative non-sexual behaviours that characterize the Young Male Syndrome, again providing support for a role for androgens in the development of male reproductive strategies/orientations.

These findings may inspire avenues for future research. The ultimate test of the integrated-LHT model would be to assess it through a longitudinal design, following a cohort of males from infancy through young adulthood, and even extending into fatherhood. Such a design would allow for relevant early environmental, psychological (i.e., early attachment style), and somatic (i.e., pubertal timing) factors to be assessed prospectively, thereby obviating the need for retrospective self-reports. Furthermore, T could be measured directly at multiple time points (i.e., perinatally, at various stages of puberty, in adulthood), allowing for a more detailed understanding of androgenic influences within the context of evolutionary-developmental models of early environment, attachment, and reproduction.

Another direction for future research is to test the LHT-based model in a broad community-based sample. Such recruitment may increase the proportion of participants coming from high-risk backgrounds, potentially serving to expand the range in scores across all psychological/behavioural variables of interest. Expanding the sample into the larger community could help to strengthen associations seen with the early family environment latent construct and to increase the number of indicators that could be used to characterize this construct (e.g., increased range in family SES, more variability in early family composition, family affection index, and family abuse index).

4.7. Conclusion

The conceptualization of attachment as an innate human tendency, one of evolutionary significance, lends itself to being characterized as not only a socially learned construct, but also by biological substrates. The role of androgens in the expression of attachment behaviours, specifically in adulthood, has been contextualized within the present study in models that outline the lifespan effects of early environmental experiences on the development of corresponding mating orientations, with attachment representing a figural mediating variable. The current study has empirically identified a conceptually plausible connection between androgenic variables and romantic attachment configurations that influence mating approaches, using structural modelling techniques. Through this work, a precedent is set for future research to continue to explore the biological, and sex hormonal, correlates of psychological attachment.

5. References

- Abdelgadir, S. E., Roselli, C. E., Choate, J. V., & Reski, J. A. (1999). Androgen receptor messenger ribonucleic acid in brains and pituitaries of male rhesus monkeys: Studies on distribution, hormonal control, and relationship to luteinizing hormone secretion. *Biology Reproduction*, 60, 1251-1256.
- Adler, A., Vescovo, P., Robinson, J. K., & Kritzer, M. F. (1999). Gonadectomy in adult life increases tyrosine hydroxylase immunoreactivity in the prefrontal cortex and decreases open field activity in male rats. *Neuroscience*, 89, 939-954.
- Ahlborn, E., Prins, G. S., & Ceccatelli, S. (2001). Testosterone protects cerebellar granule cells from oxidative stress-induced cell death through a receptor mediated mechanism. *Brain Research*, 892, 255-262.
- Aikey, J. L., Nyby, J. G., Anmuth, D. M., & James, P. J. (2002). Testosterone rapidly reduces anxiety in male house mice (*Mus musculus*). *Hormones and Behavior*, 42, 448-460.
- Ainsworth, M. D. S. (1991). Attachment and other affectional bonds across the life cycle. In C. M. Parkes, J. Stevenson-Hinde & P. Marris (Eds.), *Attachment Across the Life Cycle* (pp. 33-51). New York, NY: Tavistock/ Routledge.
- Ainsworth, M. D. S., Blehar, M. C., Waters, E., & Wall, S. (1978). Patterns of Attachment: A Psychological Study of the Strange Situation. Hillsdale, NJ: Lawrence Erlbaum Associates Inc.
- Albert, D. J., Walsh, M. L., & Jonik, R. H. (1993). Aggression in humans: What is its biological foundation? *Neuroscience and Biobehavioral Reviews*, 17, 405-425.
- Allison, P. D., & Furstenberg, F. F. (1989). How marital dissolution affects children: Variations by age and sex. *Developmental Psychology*, 25, 540-549.
- Aluja, A., Garcia, L. F., Blanch, A., & Fibla, J. (2011). Association of androgen receptor gene, CAG and GGN repeat length polymorphism and impulsive-disinhibited personality traits in inmates: the role of short–long haplotype. *Psychiatric Genetics*, 21, 229-239.
- Alvergne, A., Faurie, C., & Raymond, M. (2008). Developmental plasticity of human reproductive development: Effects of early family environment in modern-day France. *Physiology and Behavior*, 95, 625-632.
- Alvergne, A., Faurie, C., & Raymond, M. (2009). Variation in testosterone levels and male reproductive effort: Insight from a polygynous human population. *Hormones and Behavior*, 56, 491-497.
- Amato, P. R., & Keith, B. (1991). Parental divorce and the well-being of children: A meta-analysis. *Psychological Bulletin*, 110, 26-46.
- Anderson, J. C., & Gerbing, D. W. (1988). Structural equation modeling in practice: A review and recommended two-step approach. *Psychological Bulletin*, 103, 411-423.
- Apicella, C. L., Dreber, A., Campbell, B., Gray, P. B., Hoffman, M., & Little, A. C. (2008). Testosterone and financial risk preferences. *Evolution and Human Behavior*, 29, 384-390.
- Archer, J. (2006). Testosterone and human aggression: An evaluation of the challenge hypothesis. *Neuroscience and Biobehavioral Reviews*, *30*, 319-345.
- Archer, J. (1991). The influence of testosterone on human aggression. *British Journal of Psychology*, 82, 1-28.
- Archer, J., Birring, S., & Wu, F. (1998). The association between testosterone and aggression among young men: Empirical findings and a meta-analysis. *Aggressive Behavior*, 24, 411-420.
- Archer, J., Graham-Kevan, N., & Davies, M. (2005). Testosterone and aggression: A reanalysis of Book, Starzyk and Quinsey's (2001) study. *Aggression and Violent Behavior*, 10, 241-261.
- Arim, R. G., Tramonte, L., Shapka, J. D., Dahinten, V. S., & Willms, J. D. (2011). The family antecedents and the subsequent outcomes of early puberty. *Journal of Youth and Adolescence*, 40, 1423-1435.
- Auchus, R. J. (2011). The physiology and biochemistry of adrenarche. *Endocrine Development, 20*, 20-27.
- Auchus, R. J., & Rainey, W. E. (2004). Adrenarche—physiology, biochemistry and human disease. *Clinical Endocrinology*, 60, 288-296.
- Austin, E. J., Manning, J. T., McInroy, K., & Mathews, E. (2002). A preliminary investigation of the associations between personality, cognitive ability and digit ratio. *Personality and Individual Differences*, 33, 1115-1124.
- Aydogan, U., Aydogdu, A., Akbulut, H., Sonmez, A., Yuksel, S., Basaran, Y., et al. (2012). Increased frequency of anxiety, depression, quality of life and sexual life in young hypogonadotropic hypogonadal males and impacts of testosterone replacement therapy on these conditions. *Endocrine Journal*, 59, 1099-1105.
- Babcock, J. C., Jacobson, N. S., Gottman, J. M., & Yerington, T. P. (2000). Attachment, emotional regulation, and the function of marital violence: Differences between

secure, preoccupied, and dismissing violent and nonviolent husbands. *Journal of Family Violence*, 15, 391-409.

- Bagatell, C. J., Heiman, J. R., Rivier, J. E., & Bremner, W. J. (1994). Effects of endogenous testosterone and estradiol on sexual behavior in normal young men. *Journal of Clinical Endocrinology and Metabolism*, 78, 711-716.
- Bailey, A. A. & Hurd, P. L. (2005a). Finger length ratio (2D:4D) correlates with physical aggression in men but not women. *Biological Psychology*, 68, 215-222.
- Bailey, A. A. & Hurd, P. L. (2005b). Depression in men is associated with more feminine finger length ratios. *Personality and Individual Differences*, 39, 829-836.
- Baldwin, M. W., & Fehr, B. (1995). On the instability of attachment style ratings. *Personal Relationships*, 2, 247-261.
- Ballard, R. (1992). Short forms of the Marlowe–Crowne Social Desirability Scale. *Psychological Reports, 71*, 1155–1160.
- Bartholomew, K. (1990). Avoidance of intimacy: An attachment perspective. *Journal of Social and Personal Relationships*, 7, 147-178.
- Bartholomew, K., & Horowitz, L. M. (1991). Attachment style in young adults: A test of a four-category model. *Journal of Personality and Social Psychology*, 61, 226-244.
- Bartholomew, K., & Shaver, P. (1998). Methods of assessing adult attachment: Do they converge? In J. A. Simpson & W. S. Rholes (Eds.), *Attachment Theory and Close Relationships* (pp. 25-45). New York, NY: Guilford Press.
- Belsky, J. (1997). Attachment, mating and parenting: An evolutionary interpretation. *Human Nature*, *8*, 361-381.
- Belsky, J., Steinberg, L., & Draper, P. (1991). Childhood experience, interpersonal development, and reproductive strategy: An evolutionary theory of socialization. *Child Development*, 62, 647-670.
- Belsky, J., Steinberg, L., Houts, R. M., & Halpern-Felsher, B. L. (2010). The development of reproductive strategy in females: Early maternal harshness → earlier menarche → increased sexual risk taking. *Developmental Psychology*, 46, 120–128.
- Benderlioglu, Z. & Nelson, R. J. (2004). Digit length ratios predict reactive aggression in women, but not men. *Hormones and Behavior*, *46*, 558-564.

- Bentler, P. M., & Bonett, D. G. (1980). Significance tests and goodness of fit in the analysis of covariance structures. *Psychological Bulletin, 88*, 588-606.
- Berenbaum, S.A., Bryk, K.K., Nowak, N., Quigley, C.A., & Moffat, S. (2009). Fingers as a marker of prenatal androgen exposure. *Endocrinology*, *150*, 5119-5124.
- Berenbaum, S. A. & Resnick, S. M. (1997). Early androgen effects on aggression in children and adults with congenital adrenal hyperplasia. *Psychoneuroendocrinology*, 22, 505-515.
- Berger, L. M. (2004). Income, family structure, and child maltreatment risk. *Children* and Youth Services Review, 26, 725-748.
- Berglund, L. H., Prytz, H. S., Perski, A., & Svartberg, J. (2011). Testosterone levels and psychological health status in men from a general population: the Tromsø study. *The Aging Male*, *14*, 37-41.
- Bernardi, M., Genedani, S., Tagliavini, S., & Bertolini, A. (1989). Effect of castration and testosterone in experimental models of depression in mice. *Behavioral Neuroscience*, 103, 1148-1150.
- Bernier, A., & Dozier, M. (2002). Assessing adult attachment: Empirical sophistication and conceptual bases. *Attachment & Human Development*, *4*, 171-179.
- Bernroider, G., Holtzrattner, M., & Rottner, K. (1996). Sex steroid-opioid interactions associated with the temporal component of avian calling patterns. *Hormones and Behavior*, *30*, 583-589.
- Beyenburg, S., Watzka, M., Clusmann, H., Blümcke, I., Bidlingmaier, F., Elger, C. E., et al. (2000). Androgen receptor mRNA expression in the human hippocampus. *Neuroscience Letters*, *294*, 25-28.
- Bhasin, S., Woodhouse, L., Casaburi, R., Singh, A. B., Bhasin, D., Berman, N., et al. (2001). Testosterone dose-response relationships in healthy young men. *American Journal of Physiology, Endocrinology, and Metabolism, 281*, 1172-1181.
- Bianchi, S. M. (1995). The changing demographic and socioeconomic characteristics of single parent families. *Marriage and Family Review*, 20, 71-97.
- Bitar, M. S., Ota, M., Linnoila, M., & Shapiro, B. H. (1991). Modification of gonadectomy-induced increases in brain monoamine metabolism by steroid hormones in male and female rats. *Psychoneuroendocrinology*, 16, 547-557.
- Bogaert, A. & Fisher, W. (1995). Predictors of university men's number of sexual partners. *Journal of Sex Research*, *32*, 119-130.

- Bogaert, A. F., & Rushton, J. P. (1989). Sexuality, delinquency and r/K reproductive strategies: Data from a Canadian university sample. *Personality and Individual Differences*, 10, 1071-1077.
- Bogaert, A. F., Sadava, S. (2002). Adult attachment and sexual behavior. *Personal Relationships*, *9*, 191-204.
- Bollen, K. A., & Stine, R. (1990). Direct and indirect effects: Classical and bootstrapped estimates of variability. *Sociological Methodology*, 20, 115-140.
- Book, A. S., Starzyk, K. B., & Quinsey, V. L. (2001). The relationship between testosterone and aggression: A meta-analysis. *Aggression and Violent Behavior*, 6, 579-599.
- Bookwala, J., & Zdaniuk, B. (1998). Adult attachment styles and aggressive behavior within dating relationships. *Journal of Social and Personal Relationships*, 15, 175-190.
- Booth, A. & Dabbs, J. M. (1993). Testosterone and men's marriages. *Social Forces*, 72, 463-477.
- Booth, A., Johnson, D. R., & Granger, D. A. (1999). Testosterone and men's health. *Journal of Behavioral Medicine*, 22, 1-19.
- Booth, A., Johnson, D. R., Granger, D. A., Crouter, A. C., & McHale, S. (2003). Testosterone and child and adolescent adjustment: The moderating role of parentchild relationships. *Developmental Psychology*, 39, 85-98.
- Bowen, N. K., & Guo, S (2012). *Structural Equation Modeling*. New York, NY: Oxford University Press.
- Bowlby, J. (1969/1982). *Attachment and Loss: Vol. 1. Attachment*. New York, NY: Basic Books.
- Bowlby, J. (1973). Attachment and Loss: Vol. 2. Separation: Anxiety and Anger. New York, NY: Basic Books.
- Bowlby, J. (1979). *The Making and Breaking of Affectional Bonds*. London, UK: Tavistock/ Routledge.
- Bowlby, J. (1980). Attachment and Loss: Vol. 3. Loss: Sadness and Depression. New York, NY: Basic Books.
- Brain, P. F. (1979). *Hormones and Aggression* (Annual Research Reviews, Vol. 2). Montreal, QC: Eden Press.

- Brañas-Garza, P., & Rustichini, A. (2011). Organizing effects of testosterone and economic behavior: Not just risk taking. *PLoS ONE*, *6*, 12.
- Brassard, A., Shaver, P. R., & Lussier, Y. (2007). Attachment, sexual experience, and sexual pressure in romantic relationships: A dyadic approach. *Personal Relationships*, 14, 475-493.
- Breedlove, S. M. (2010). Organizational hypothesis: Instances of the fingerpost. *Endocrinology*, 151, 4116-4122.
- Breedlove, S. M. (1992). Sexual dimorphism in the vertebrate nervous system. *Journal of Neuroscience*, *12*, 4133-4142.
- Breedlove, S. M. & Hampson, E. (2002). Sexual differentiation of the brain and behavior. In: J. B. Becker, S. M. Breedlove, D. Crews, & M. M. McCarthy (Eds.), *Behavioral Endocrinology* (2nd ed.) (pp. 75-114). Cambridge, MA: MIT Press.
- Brennan, K. A., Clark, C. L., & Shaver, P. R. (1998). Self-report measurement of adult attachment: An integrative overview. In J. A. Simpson & W. S. Rholes (Eds.), *Attachment Theory and Close Relationships* (pp. 46-76). New York, NY: Guilford Press.
- Brennan, K. A., & Shaver, P. R. (1991). Dimensions of attachment and the dynamics of romantic relationships. Cited in Belsky, J. (1997) Attachment, mating and parenting: An evolutionary interpretation. *Human Nature*, 8, 361-381.
- Brennan, K. A., & Shaver, P. R. (1995). Dimensions of adult attachment, affect regulation, and romantic relationship functioning. *Personality and Social Psychology Bulletin*, 21, 267-283.
- Broidy, L. M., Nagin, D. S., Tremblay, R. E., Bates, J. E., Brame, B., Dodge, K. A., et al. (2003). Developmental trajectories of childhood disruptive behaviors and adolescent delinquency: A six-site, cross-national study. *Developmental Psychology*, 39, 222-245.
- Brosnan, M., Gallop, V., Iftikhar, N., & Keogh, E. (2011). Digit ratio (2D:4D), academic performance in computer science and computer-related anxiety. *Personality and Individual Differences*, 51, 371-375.
- Brotto, L. A., Woo, J. S. T., & Ryder, A. G. (2007). Acculturation and sexual function in Canadian East Asian men. *Journal of Sexual Medicine*, *4*, 72-82.
- Brown, R. E. (1994). Steroid and thyroid hormone receptors. In R. E. Brown (Ed.), *An Introduction to Neuroendocrinology* (pp. 147-182). Cambridge: Cambridge University Press.

- Brown, G. W., & Harris, T. (1978). Social Origins of Depression. London, UK: Tavistock.
- Brown, W. M., Hines, M., Fane, B. A., & Breedlove, S. M. (2002). Masculinized finger length patterns in human males and females with congenital adrenal hyperplasia. *Hormones and Behavior*, *42*, 380-386.
- Brussoni, M. J., Jang, K. L., Livesley, W. J., & MacBeth, T. M. (2000). Genetic and environmental influences on adult attachment styles. *Personal Relationships*, 7, 283-289.
- Buchanan, G., Yang, M., Cheong, A., Harris, J. M., Irvine, R. A., Lambert, P. F., et al. (2004). Structural and functional consequences of glutamine tract variation in the androgen receptor. *Human Molecular Genetics*, 13, 1677-1692.
- Buck, J. J., Williams, R. M., Hughes, I. A., & Acerini, C. L. (2003). In-utero androgen exposure and 2nd to 4th digit length ratio-comparisons between healthy controls and females classical congenital adrenal hyperplasia. *Human Reproduction*, 18, 976-979.
- Buddenberg, T.E., Komorowski, M., Ruocco, L.A., Silva, M.A., & Topic, B. (2009). Attenuating effects of testosterone on depressive-like behavior in the forced swim test in healthy male rats. *Brain Research Bulletin*, 79, 182-186.
- Burnham, T. C., Chapman, J. F., Gray, P. B., McIntyre, M. H., Lipson, S. F., & Ellison, P. T. (2003). Men in committed, romantic relationships have lower testosterone. *Hormones and Behavior*, 44, 119-122.
- Buss, A. H., & Perry, M. (1992). The aggression questionnaire. *Journal of Personality* and Social Psychology, 63, 452-459.
- Buss, D. M. (1998). Sexual strategies theory: Historical origins and current status. *The Journal of Sex Research*, *35*, 19-31.
- Buss, D. M., & Schmitt, D. P. (1993). Sexual strategies theory: An evolutionary perspective on human mating. *Psychological Review*, 100, 204-232.
- Butovskaya, M. L., Burkova, V. N., & Mabulla, A. (2010). Sex difference in 2D:4D ratio, aggression and conflict resolution in African children and adolescents: A cross-cultural study. *Journal of Aggression, Conflict and Peace Research, 2*, 17-31.
- Butovskaya, M., Fedenok, J., Burkova, V., & Manning, J. (2013). Sex differences in 2D:4D and aggression in children and adolescents from five regions of Russia. *American Journal of Physical Anthropology*, *152*, 130-139.

Butovskaya, M. L., Vasilyev, V. A., Lazebny, O. E., Burkova, V. N., Kulikov, A. M.,

Mabulla, A., et al. (2012). Aggression, digit ratio, and variation in the androgen receptor, serotonin transporter, and dopamine D4 receptor genes in African roragers: The Hadza. *Behavior Genetics*, *42*, 647-662.

- Bylsma, W. H., Cozzarelli, C., & Sumer, N. (1997). Relation between attachment styles and global self-esteem. *Basic and Applied Social Psychology*, 19, 1-16.
- Byrne, B. M. (1983). Investigating measures of self-concept. *Measurement and Evaluation in Guidance, 16*, 115-126.
- Carbone, S., Ponzo, O. J., Gobetto, N., Samaniego, Y. A., Reynoso, R., Scacchi, P., et al. (2013). Antiandrogenic effect of perinatal exposure to the endocrine disruptor di-(2-ethylhexyl) phthalate increases anxiety-like behavior in male rats during sexual maturation. *Hormones and Behavior*, 63, 692-699.
- Carnes, P.J. (1991). *Don't Call it Love: Recovering from Sexual Addiction*. New York, NY: Bantam Books.
- Carrier, N., & Kabbaj, M. (2012). Testosterone and imipramine have antidepressant effects in socially isolated male but not female rats. *Hormones and Behavior, 61*, 678-685.
- Cassidy, J. (1988). Child-mother attachment and the self in six-year-olds. *Child Development*, 59,121-134.
- Cassidy, J., & Berlin, L. (1994). The insecure/ambivalent pattern of attachment: Theory and research. *Child Development*, 65, 971-991.
- Chapple, C. L., & Johnson, K. A. (2007). Gender differences in impulsivity. *Youth Violence and Juvenille Justice*, *5*, 221-234.
- Charles, N. E., & Alexander, G.M. (2011). The association between 2D:4D ratios and sociosexuality: A failure to replicate. *Archives of Sexual Behavior*, 40, 587-595.
- Chasiotis, A., Scheffer, D., Restemeier, R., & Keller, H. (1998). Intergenerational context discontinuity affects the onset of puberty. A comparison of parent-child dyads in West and East Germany. *Human Nature*, 9, 321-339.
- Chen, B.-B., & Chang, L. (2012). Adaptive insecure attachment and resource control strategies during middle childhood. *International Journal of Behavioral Development*, 36, 389-397.
- Cheng, D., Hong, C. J., Lia, D. L., & Tsai, S. J. (2006). Association study of androgen receptor CAG repeat polymorphism and male violent criminal activity. *Psychoneuroendocrinology*, 31, 548-552.

- Cherlin, A. J., Chase-Lansdale, P. L., & McRae, C. (1998). Effects of parental divorce on mental health throughout the life course. *American Sociological Review*, 63, 239-249.
- Chisholm, J. S. (1993). Death, hope, and sex. Life-history theory and the development of reproductive strategies. *Current Anthropology*, *34*, 1-24.
- Chisholm, J. S. (1996). The evolutionary ecology of attachment organization. *Human Nature*, *7*, 1-38.
- Chisholm, J. S. (1999). *Death, Hope, and Sex: Steps to an Evolutionary Ecology of Mind and Mortality*. Cambridge: Cambridge University Press.
- Choong, C. S., Kemppainen, J. A., Zhou, Z-X., & Wilson, E. M. (1996). Reduced androgen receptor gene expression with first exon CAG repeat expansion. *Molecular Endocrinology*, 10, 1527-1535.
- Choong, C. S., & Wilson, E. M. (1998). Trinucleotide repeats in the human androgen receptor: A molecular basis for disease. *Journal of Molecular Endocrinology*, 21, 235-257.
- Chopik, W. J., Edelstein, R. S., & Fraley, R. C. (2013). From the cradle to the grave: Age differences in attachment from early adulthood to old age. *Journal of Personality*, 81, 171-183.
- Clark, A. P. (2004). Self-perceived attractiveness and masculinisation predict women's sociosexuality. *Evolution and Human Behavior, 25*, 113-124.
- Clarke-Stewart, K. A., Vandell, D. L., McCartney, K., Owen, M. T., & Booth, C. (2000). Effects of parental separation and divorce on very young children. *Journal of Family Psychology*, 14, 304-326.
- Clifton, R. A., Perry, R. P., Roberts, L. W., & Peter, T. (2008). Gender, psychosocial dispositions and the academic achievement of college students. *Research in Higher Education*, 49, 684-703.
- Coan, J. A. (2008). Towards a neuroscience of attachment. In J. Cassidy & P. R. Shaver (Eds.), *Handbook of Attachment. Theory, Research, and Clinical Applications* (pp. 241-265). New York, NY: Guilford Press.
- Coates, J. M., Gurnell, M., & Rustichini, A. (2009). Second-to-fourth digit ratio predicts success among high-frequency financial traders. *Proceedings of the National Academy of Science*, 106, 623-628.
- Cohen-Bendahan, C. C. C., Buitelaar, J. K., van Goozen, S. H. M., Orlebeke, J. F., & Cohen-Kettenis, P. T. (2005). Is there an effect of prenatal testosterone on

aggression and other behavioral traits? A study comparing same-sex and opposite-sex twin girls. *Hormones and Behavior*, 47, 230-237.

- Collaer, M. L. & Hines, M. (1995). Human behavioral sex differences: A role for gonadal hormones during early development? *Psychological Bulletin*, *11*, 55-107.
- Collins, N. L., & Read, S. J. (1990). Adult attachment, working models, and relationship quality in dating couples. *Journal of Personality and Social Psychology*, *58*, 644-663.
- Cooper, R. M., Rowe, A. C., Penton-Voak, I. S., & Ludwig, C. (2009). No reliable effects of emotional facial expression, adult attachment orientation, or anxiety on the allocation of visual attention in the spatial cueing paradigm. *Journal of Research in Personality, 43*, 643-652.
- Cooper, M. L., Shaver, P. R., & Collins, N. L. (1998). Attachment styles, emotion regulation, and adjustment in adolescence. *Journal of Personality and Social Psychology*, 74, 1380–1397.
- Cousins, A. J., Fugère, M. A., & Franklin, M. (2009). Digit ratio (2D:4D), mate guarding, and physical aggression in dating couples. *Personality and Individual Differences*, 46, 709-713.
- Crabbe, P., Bogaert, V., De Bacquer, D., Goemaere, S., Zmierczak, H., & Kaufman, J. M. (2007). Part of the interindividual variation in serum testosterone levels in healthy men reflects differences in androgen sensitivity and feedback set point: Contribution of the androgen receptor polyglutamine tract polymorphism. *Journal* of Clinical Endocrinology and Metabolism, 92, 3604-3610.
- Creasey, G., & Hesson-McInnis, M. (2001). Affective responses, cognitive appraisals, and conflict tactics in late adolescent romantic relationships: Associations with attachment orientations. *Journal of Counseling Psychology*, *48*, 85-96.
- Crockenberg, S. (1987). Predictors and correlates of anger toward and punitive control of toddlers by adolescent mothers. *Child Development*, 58, 964-975.
- Crowell, J. A., Fraley, R. C., & Shaver, P. R. (2008). Measurement of individual differences in adolescent and adult attachment. In J. Cassidy & P. R. Shaver (Eds.), *Handbook of Attachment. Theory, Research, and Clinical Applications* (pp. 599-634). New York, NY: Guilford Press.
- Crowne, D. P., & Marlowe, D. (1960). A new scale of social desirability independent of psychopathology. *Journal of Consulting Psychology*, *24*, 349-354.

- Cummings, D. C., & Wall, S. R. (1985). Non-sex hormone-binding globulin-bound testosterone as a marker for hyperandrogenism. *Journal of Clinical Endocrinology and Metabolism*, 61, 873-876.
- Davis, D., Shaver, P. R., & Vernon, M. L. (2004). Attachment styles and subjective motivations for sex. *Personality and Social Psychology Bulletin*, 30, 1076-1090.
- de Bruin, E. I., Verheij, F., Wiegman, T., & Ferdinand, R. F. (2006). Differences in finger length ratio between males with autism, pervasive developmental disorder– not otherwise specified, ADHD, and anxiety disorders. *Developmental Medicine* & Child Neurology, 48, 962-965.
- DeLecce, T. L., Polheber, J. P., & Matchock, R. L. (2014). Sociosexual orientation and 2D:4D ratios in women: Relationship to men's desirability ratings as a long-term pair bond. *Archives of Sexual Behavior*, 43, 319-327.
- Del Giudice, M. (2008). Sex-based ratio of avoidant/ambivalent attachment in middle childhood. *British Journal of Developmental Psychology*, *26*, 369-379.
- Del Giudice, M. (2009). Sex, attachment, and the development of reproductive strategies. *Behavioral and Brain Sciences*, *32*, 1-67.
- Del Giudice, M. (2011). Sex differences in romantic attachment: A meta-analysis. *Personality and Social Psychology Bulletin, 37*, 193-214.
- Del Giudice, M., Angeleri, R., & Manera, V. (2009). The juvenile transition: A developmental switch point in human life history. *Developmental Review, 29*, 1-31.
- Del Giudice, M., & Belsky, J. (2010). Sex differences in attachment emerge in middle childhood: An evolutionary hypothesis. *Child Development Perspectives*, *4*, 97-105.
- Deonandan, R., Campbell, K., Ostbye, T., Tummon, I., & Robertson, J. (2000). A comparison of methods for measuring socio-economic status by occupation or postal area. *Chronic Diseases in Canada, 21*, 114-118.
- De Vaus, D., Qu, L., & Weston, R. (2003). Changing patterns of partnering. *Family Matters*, 64, 10-15.
- DeWall, C. N., Lambert, N. M., Slotter, E. B., Pond Jr., R. S., Deckman, T., Finkel, EJ., et al. (2011). So far away from one's partner, yet so close to romantic alternatives: Avoidant attachment, interest in alternatives, and infidelity. *Journal of Personality and Social Psychology*, 101, 1302-1316.

Dickman, S. J. (1990). Functional and dysfunctional impulsivity: Personality and

cognitive correlates. Journal of Personality and Social Psychology, 58, 95-102.

- Ding, D., Xu, L., Menon, M., Reddy, G. P. V., & Barrack, E. R. (2004). Effect of a short CAG (glutamine) repeat on human androgen receptor function. *The Prostate*, 58, 23-32.
- Dixson, A. F. (1993). Sexual and aggressive behaviour of adult male marmosets (*Callithrix jacchus*) castrated neonatally, prepubertally, or in adulthood. *Physiology and Behavior*, 54, 301-307.
- Donnellan, M. B., Burt, S. A., Levendosky, A. A., & Klump, K. L. (2008). Genes, personality, and attachment in adults: A multivariate behavioral genetic analysis. *Personality and Social Psychology Bulletin, 34*, 3-16.
- Doumas, D. M., Turrisi, R., & Wright, D. A. (2006). Risk factors for heavy drinking in college freshmen: Athletic status and adult attachment. *The Sport Psychologist*, 20, 419-434.
- Downey, R. G., & King, C. V. (1998). Missing data in likert ratings: A comparison of replacement methods. *The Journal of General Psychology*, 125, 175-191.
- Draper, P., & Harpending, H. (1982). Father absence and reproductive strategy: An evolutionary perspective. *Journal of Anthropological Research*, *38*, 255-273.
- Dubas, J. S., Graber, J. A., & Petersen, A. C. (1991). A longitudinal investigation of adolescents' changing perceptions of pubertal timing. *Developmental Psychology*, 27, 580-586.
- Ducharme, J. R., Forest, M. G., De Peretti, E., Sempe, M., Collu, R., & Bertrand, J. (1976). Plasma adrenal and gonadal sex steroids in human pubertal development. *Journal of Clinical Endocrinology and Metabolism, 42*, 468-476.
- Duncan, P. D., Ritter, P. L., Dornbusch, S. M., Gross, R. T., & Carlsmith, J. M. (1985). The effects of pubertal timing on body image, school behavior, and deviance. *Journal of Youth and Adolescence*, 14, 227-235.
- Dutton, D. G., Saunders, K., Starzomski, A., & Bartholomew, K. (1994). Intimacy-anger and insecure attachment as precursors of abuse in intimate relationships. *Journal* of Applied Social Psychology, 24, 1367-1386.
- Eckel, L. A., Arnold, A. P., Hampson, E., Becker, J. B., Blaustein, J. D., & Herman, J. P. (2008). Research and methodological issues in the study of sex differences and hormone-behavior relations. In Becker, J. B., Berkley, K. J., Geary, N., Hampson, E., Herman, J. P., & Young, E. A. (Eds.), *Sex Differences in the Brain: From Genes to Behavior* (pp. 35-61). New York, NY: Oxford University Press.

- Edelstein, R. S., Chopik, W. J., & Kean, E. L. (2011). Sociosexuality moderates the association between testosterone and relationship status in men and women. *Hormones and Behavior, 60*, 248-255.
- Edelstein, R. S., Stanton, S. J., Henderson, M. M., & Sanders, M. R. (2010). Endogenous estradiol levels are associated with attachment avoidance and implicit intimacy motivation. *Hormones and Behavior*, 57, 230-236.
- Edelstein, R. S., van Anders, S. M., Chopik, W. J., Goldey, K. L., & Wardecker, B. M. (2014). Dyadic associations between testosterone and relationship quality in couples. *Hormones and Behavior*, 65, 401-407.
- Edwards, A., Hammond, H.A., Jin, L., Caskey, C.T., & Chakraborty, R. (1992). Genetic variation at five trimeric and tetrameric tandem repeat loci in four human population groups. *Genomics*, *12*, 241-253.
- Ellis, B. J. (2004). Timing of pubertal maturation in girls: An integrated life history approach. *Psychological Bulletin, 130*, 920-958.
- Ellis, B. J., & Essex, M. J. (2007). Family environment, adrenarche, and sexual maturation: A longitudinal test of a life history model. *Child Development*, *78*, 1799-1817.
- Elmlinger, M. W., Künhel, W., Wormstall, H., & Döller, P. C. (2005). Reference intervals for testosterone, androstenedione and SHBG levels in healthy females and males from birth until old age. *Clinical Laboratory*, *51*, 625-632.
- Epple, G., Alveario, M. C., & St. André, E. (1987). Sexual and social behavior of adult saddle-back tamarins (*Saguinus fuscicollis*), castrated as neonates. *American Journal of Primatology*, 13, 37-49.
- Erickson, M. F., Sroufe, L. A., & Egeland, B. (1985). The relationship between quality of attachment and behavior problems in preschool in a high-risk sample. *Monographs of the Society for Research in Child Development, 50*, 147-166.
- Evans, K. L., & Hampson, E. (2014). Does risk-taking mediate the relationship between testosterone and decision-making on the Iowa Gambling Task? *Personality and Individual Differences*, 61-62, 57-62.
- Evardone, M., & Alexander, G. M. (2009). Anxiety, sex-linked behaviors, and digit ratios (2D:4D). *Archives of Sexual Behavior*, *38*, 442-455.
- Fairchild, A. J., & Finney, S. J. (2006). Investigating validity evidence for Experiences in Close Relationships-Revised Questionnaire. *Educational and Psychological Measurement*, 66, 116-135.

- Faisandier, K. M., Taylor, J. E., & Salisbury, R. M. (2012). What does attachment have to do with out-of-control sexual behaviour? *New Zealand Journal of Psychology*, 41, 19-29.
- Feeney, J. A. (2008). Adult romantic attachment: Developments in the study of couple relationships. In J. Cassidy & P. R. Shaver (Eds.), *Handbook of Attachment. Theory, Research, and Clinical Applications* (pp. 456-481). New York, NY: Guilford Press.
- Feeney, J. A., Kelly, L., Gallois, C., Peterson, C., & Terry, D. J. (1999). Attachment style, assertive communication, and safer-sex behavior. *Journal of Applied Social Psychology*, 29, 1964-1983.
- Feeney, J. A., & Noller, P. (1992). Attachment style and romantic love: Relationship dissolution. *Australian Journal of Psychology*, 44, 69-74.
- Feeney, J. A., & Noller, P. (1990). Attachment style as a predictor of adult romantic relationships. *Journal of Personality and Social Psychology*, 58, 281-291.
- Feeney, J. A., Noller, P., & Patty, J. (1993). Adolescents' interaction with the opposite sex: Influences of attachment style and gender. *Journal of Adolescence*, 16, 169-186.
- Finch, J. F., West, S. G., & MacKinnon, D. P. (1997). Effects of sample size and nonnormality on the estimation of mediated effects in latent variable models. *Structural Equation Modeling*, 4, 87–105.
- Fink, B., Manning, J. T., & Neave, N. (2004). Second to fourth digit ratio and the 'big five' personality factors. *Personality and Individual Differences*, *37*, 495-503.
- Fink, B., Manning, J. T., Williams, J. H. G., & Podmore-Nappin, C. (2007). The 2nd to 4th digit ratio and developmental psychopathology in school-aged children. *Personality and Individual Differences*, 42, 369-379.
- Finkel, D., & Matheny, A. P. Jr. (2000). Genetic and environmental influences on a measure of infant attachment security. *Twin Research*, *3*, 242-250.
- Finley, S. K., & Kritzer, M. F. (1999). Immunoreactivity for intracellular androgen receptors in identified subpopulations of neurons, astrocytes and oligodendrocytes in primate prefrontal cortex. *Journal of Neurobiology*, 40, 446-457.
- Finnegan, R. A., Hodges, E., & Perry, D. (1996). Preoccupied and avoidant coping during middle childhood. *Child Development*, 67, 1318-1328.

- Fischer, D.G., & Fick, C. (1993). Measuring social desirability: Short forms of the Marlowe–Crowne Social Desirability Scale. *Educational and Psychological Measurement*, 53, 417–424.
- Fisher, C. L., Mannino, D. M., Herman, W. H., & Frumkin, H. (1997). Cigarette smoking and thyroid hormone levels in males. *International Journal of Epidemiology*, 26, 972-977.
- Fleming, J. S., & Courtney, B. E. (1984). The dimensionality of self-esteem: II. Hierarchical facet model for revised measurement scales. *Journal of Personality* and Social Psychology, 46, 404-421.
- Forest, M.G. (1979). Plasma androgens (testosterone and 4-androstenedione) and 17hydroxyprogesterone in the neonatal, prepubertal, and peripubertal periods in the human and the rat: Differences between species. *Journal of Steroid Biochemistry*, *11*, 543-548.
- Fournier, B., Brassard, A., & Shaver, P. R. (2011). Adult attachment and male aggression in couple relationships: the demand-withdraw communication pattern and relationship satisfaction as mediators. *Journal of Interpersonal Violence, 26*, 1982-2003.
- Fox-Wasylyshyn, S. M., & El-Masri, M. M. (2005). Handling missing data in self-report measures. *Research in Nursing & Health*, 28, 488-495.
- Fraley, R. C. (2002). Attachment stability from infancy to adulthood: meta-analysis and dynamic modeling of developmental mechanisms. *Personality and Social Psychology Review*, 6, 123-151.
- Fraley, R. C., & Shaver, P. R. (2000). Adult romantic attachment: Theoretical developments, emerging controversies, and unanswered questions. *Review of General Psychology*, 4, 132-154.
- Fraley, R. C., & Waller, N. G. (1998). Adult attachment patterns. A test of the typological model. In J. A. Simpson & W. S. Rholes (Eds.), *Attachment Theory* and Close Relationships (pp. 77-114). New York, NY: Guilford Press.
- Fraley, R.C., Waller, N.G., & Brennan, K.A. (2000). An item response theory analysis of self-report measures of adult attachment. *Journal of Personality and Social Psychology*, 78, 350-365.
- Fujimoto, T., Kubo, K., & Aou, S. (2006). Prenatal exposure to bisphenol A impairs sexual differentiation of exploratory behavior and increases depression-like behavior in rats. *Brain Research*, 1068, 49-55.

- Furstenberg, F. F., & Teitler, J. O. (1994). Reconsidering the effects of marital disruption. What happens to children of divorce in early adulthood? *Journal of Family Issues*, 15, 173-190.
- Galyer, K. T., Conaglen, H. M., Hare, A., & Conaglen, J. V. (1999). The effects of gynecological surgery on sexual desire. *Journal of Sex & Marital Therapy*, 25, 81–88.
- Garbarino, E., Slonim, R., Sydnor, J. (2011). Digit ratios (2D:4D) as predictors of risky decision making for both sexes. *Journal of Risk and Uncertainty* 42, 1-26.
- Gentzler, A. L., & Kerns, K. A. (2004). Associations between insecure attachment and sexual experiences. *Personal Relationships*, 11, 249-265.
- George, C., Kaplan, N., & Main, M. (1984/1985/1996). *Adult Attachment Interview Protocol.* Unpublished manuscript, University of California at Berkeley.
- Gervai, J., Nemoda, Z., Lakatos, K., Ronai ,Z., Toth, I., Ney, K., et al. (2005). Transmission Disequilibrium Tests confirm the link between DRD4 gene polymorphism and infant attachment. *American Journal of Medical Genetics -Part B (Neuropsychiatric Genetics)*, 132B, 126-130.
- Gervai, J., Novak, A., Lakatos, K., Toth, I., Ronai, Z., Sasvari-Szekely, M., et al. (2007). Infant genotype may moderate sensitivity to maternal affective communication: Attachment disorganization, quality of care, and the DRD4 polymorphism. *Social Neuroscience*, 2, 307–319.
- Gesquiere, L. R., Altmann, J., Khan, M. Z., Couret, J., Yu, J. C., Endres, C. S., et al. (2005). Coming of age: Steroid hormones of wild immature baboons (*Papio cynocephalus*). American Journal of Primatology, 67, 83-100.
- Gettler, L. T., McDade, T. W., Feranil, A. B., & Kuzawa, C. W. (2011). Longitudinal evidence that fatherhood decreases testosterone in human males. *Proceedings of the National Academy of Science, 108*, 16194-16199.
- Gibson-Davis, C. M. (2008). Family structure effects on maternal and paternal parenting in low-income families. *Journal of Marriage and Family*, 70, 452-465.
- Gillath, O., Shaver, P. R., Baek, J-M., & Chun, D. S. (2008). Genetic correlates of adult attachment style. *Personality and Social Psychology Bulletin*, *34*, 1396-1405.
- Gordon, I., Zagoory-Sharon, O., Schneiderman, I., Leckman, J. F., Weller, A., & Feldman, R. (2008). Oxytocin and cortisol in romantically unattached young adults: Associations with bonding and psychological distress. *Psychophysiology*, 45, 349-352.

- Goonewardena, K., Rankin, K. N., & Weinstock, S. (2004). Canadian Journal of Urban Research, 13, 1-26.
- Gormley, B., & Lopez, F. G. (2010). Correlates of psychological abuse perpetration in college dating relationships. *Journal of College Counseling*, 13, 4-16.
- Gormley, B., & Lopez, F. G. (2003, August). Contributions of stress, gender, adult attachment orientations, and defense mechanisms to psychological abuse in college dating relationships. Poster session presented at the annual convention of the American Psychological Association, Toronto, Canada.
- Gorski, R. A., Gordon, J. R., Shryne, J. E., & Southam, A. M. (1978). Evidence for amorphological sex difference within the medial preoptic area of the rat brain. *Brain Research*, 148, 333-346.
- Goudriaan, A. E., Lapauw, B., Ruige, J., Feyen, E., Kaufman, J-M., Brand, M., et al. (2010). The influence of high-normal testosterone levels on risk-taking in healthy males in a 1-week letrozole administration study. *Psychoneuroendocrinology*, 35, 1416-1421.
- Goutou, M., Sakka, C., Stakias, N., Sefanidis, I., & Koukoulis, G. N. (2009). AR CAG repeat length is not associated with serum gonadal steroids and lipid levels in healthy men. *International Journal of Andrology*, *32*, 616-622.
- Graber, J. A., & Brooks-Gunn, J. (1996). Transitions and turning points: Navigating the passage from childhood through adolescence. *Developmental Psychology*, *32*, 768-776.
- Granger, D. A., Shirtcliff, E. A., Zahn-Waxler, C., Usher, B., Klimes-Dougan, B., & Hastings, P. (2003). Salivary testosterone diurnal variation and psychopathology in adolescent males and females: Individual differences and developmental effects. *Development and Psychopathology*, 15, 431-449.
- Gray, P.B. (2003). Marriage, parenting, and testosterone variation among Kenyan Swahili men. *American Journal of Physical Anthropology*, *122*, 279–286.
- Gray, P. B., Chapman, J. F., Burnham, T. C., McIntyre, M. H., Lipson, S. F., & Ellison, P. T. (2004). Human male pair bonding and testosterone. *Human Nature*, 15, 119-131.
- Gray-Little, B., Williams, V.S.L., & Hancock, T. D. (1997). An item response theory analysis of the Rosenberg Self-Esteem Scale. *Personality and Social Psychology Bulletin, 23*, 443-451.

- Granot, D., & Mayseless, O. (2001). Attachment security and adjustment to school in middle childhood. *International Journal of Behavioral Development*, 25, 530-541.
- Griffin, D. W., & Bartholomew, K. (1994). Models of the self and other: Fundamental dimensions underlying measures of adult attachment. *Journal of Personality and Social Psychology*, 67, 430-445.
- Griskevicius, V., Tybur, J. M., Delton, A. W., & Robertson, T. E. (2011). The influence of mortality and socioeconomic status on risk and delayed rewards: A life history theory approach. *Journal of Personality and Social Psychology, 100*, 1015-1026.
- Gupta, S. K., Lindemulder, E. A., & Sathyan, G. (2000). Modeling of circadian testosterone in healthy men and hypogonadal men. *Journal of Clinical Pharmacology*, *40*, 731-738.
- Halpern, C. T., Campbell, B., Agnew, C. R., Thompson, V., & Udry, J. R. (2002). Associations between stress reactivity and sexual and nonsexual risk taking in young adult human males. *Hormones and Behavior*, 42, 387-398.
- Hamilton, C. E. (2000). Continuity and discontinuity of attachment from infancy through adolescence. *Child Development*, *71*, 690–694.
- Hampson, E., Ellis, C. L., & Tenk, C. M. (2008). On the relation between 2D:4D and sex-dimorphic personality traits. *Archives of Sexual Behavior*, *37*, 133-144.
- Hampson, E., & Sankar, J.S. (2012a). Hand preference in humans is associated with testosterone levels and androgen receptor gene polymorphism. *Neuropsychologia*, 50, 2018-2025.
- Hampson, E., & Sankar J.S. (2012b). Re-examining the Manning hypothesis: Androgen receptor polymorphism and the 2D:4D digit ratio. *Evolution and Human Behavior*, *33*, 557-561.
- Hamson, D. K., Jones, B. A., Csupity, A. S., Ali, F. M., & Watson, N. V. (2014). Androgen insensitive male rats display increased anxiety-like behavior on the elevated plus maze. *Behavioural Brain Research*, 259, 158-163.
- Hankin, B. L., Kassel, J. D., & Abela, J. R. Z. (2005). Adult attachment dimensions and specificity of emotional distress symptoms: Prospective investigations of cognitive risk and interpersonal stress generation as mediating mechanisms. *Personality and Social Psychology Bulletin*, 31, 136-151.
- Hanoch, Y., Gummerum, M., & Rolison, J. (2012). Second-to-fourth digit ratio and impulsivity: A comparison between offenders and nonoffenders. *PLoS ONE*, *7*, 10.

- Härkönen, K., Huhtaniemi, I., Mäkinen, J., Hübler, D., Irjala, K., Koskenvuo, M., et al. (2003). The polymorphic androgen receptor gene CAG repeat, pituitary-testicular function and andropausal symptoms in ageing men. *International Journal of Andrology*, 26, 187-194.
- Harris, J. A. (1997). A further evaluation of the aggression questionnaire: Issues of validity and reliability. *Behaviour Research and Therapy*, 35, 1047-1053.
- Hawley, P. H., Shorey, H. S., & Alderman, P. M. (2009). Attachment correlates of resource-control strategies: Possible origins of social dominance and interpersonal power differentials. *Journal of Social and Personal Relationships*, 26, 1097-1118.
- Hayduk, L. A., & Littvay, L. (2012). Should researchers use single indicators, best indicators, or multiple indicators in structural equation models? *BMC Medical Research Methodology*, *12*, 159.
- Hazan, C., & Shaver, P. R. (1987). Romantic love conceptualized as an attachment process. *Journal of Personality and Social Psychology*, *52*, 511-524.
- Hazan, C., & Shaver, P. R. (1990). Love and work: An attachment-theoretical perspective. *Journal of Personality and Social Psychology*, *59*, 270-280.
- Hazan, C., & Shaver, P. R. (1994). Attachment as an organizational framework for research on close relationships. *Psychological Inquiry*, *5*, 1-22.
- Helleday, J., Edman, G., Ritzen, E. M., & Siwers, B. (1993). Personality characteristics and platelet MAO activity in women with congenital adrenal hyperplasia. *Psychoneuroendocrinology*, 18, 343-354.
- Hernandez, L., Gonzalez, L., Murzi, E., Páez, X., Gottberg, E., & Baptista, T. (1994). Testosterone modulates mesolimbic dopaminergic activity in male rats. *Neuroscience Letters*, *171*, 172-174.
- Hetherington, E. M., Bridges, M., & Insabella, G. M. (1998). What matters? What does not? Five perspectives on the association between marital transitions and children's adjustment. *American Psychologist, 53*, 167-184.
- Hill, E., Blow, F., Young, J., & Singer, K. (1994b). Family history of alcoholism and childhood adversity: Joint effects on alcohol consumption and dependence. *Alcoholism: Clinical and Experimental Research, 18*, 1083-1090.
- Hill, E. M., Jenkins, J., & Farmer, L. (2008). Family unpredictability, future discounting, and risk taking. *The Journal of Socio-Economics*, *37*, 1381-1396.

- Hill, E. M., Young, J. P., & Nord, J. L. (1994a). Childhood adversity, attachment security, and adult relationships: A preliminary study. *Ethology and Sociobiology*, 15, 323-338.
- Hollingshead, A. B. (1975). Four-factor index of social status. Unpublished manuscript, Yale University, New Haven, CT.
- Holtzworth-Munroe, A., Stuart, G. L., & Hutchinson, G. (1997). Violent versus nonviolent husbands: Differences in attachment patterns, dependency, and jealousy. *Journal of Family Psychology*, *11*, 314-331.
- Hönekopp, J. (2011). Relationships between digit ratio 2D:4D and self-reported aggression and risk taking in an online study. *Personality and Individual Differences*, 51, 77-80.
- Hönekopp, J., Voracek, M., & Manning, J. T. (2006). 2nd to 4th digit ratio (2D:4D) and number of sex partners: Evidence for effects of prenatal testosterone in men. *Psychoneuroendocrinology*, *31*, 30-37.
- Hönekopp, J., & Watson, S. (2011). Meta-analysis of the relationship between digit-ratio 2D:4D and aggression. *Personality and Individual Differences*, 51, 381-386.
- Howell, S. H., Portes, P., & Brown, J. H. (1997). Gender and age differences in child adjustment to parental separation. *Journal of Divorce and Remarriage*, *27*, 141-158.
- Howlett, C., Marshall, A. R., & Hughes, W. O. H. (2012). Digit ratios and dominance in female baboons (*Papio hamadryas and Papio ursinus*). *International Journal of Primatology*, 33, 1439-1452.
- Hsiao, P.W., Lin, D.L., Nakao, R., & Chang, C. (1999). The linkage of Kennedy's neuron disease to ARA24, the first identified androgen receptor polyglutamine region-associated coactivator. *Journal of Biological Chemistry*, 274, 20229-20234.
- Hsing, A.W., Gao, Y.T., Wu, G., Wang, X., Deng, J., Chen, Y., et al. (2000). Polymorphic CAG and GGN repeat lengths in the androgen receptor gene and prostate cancer risk: A population-based case-control study in China. *Cancer Research*, 60, 5111-5116.
- Hu, L., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling*, 6, 1–55.
- Hudson, S. M., & Ward, T. (1997). Intimacy, loneliness, and attachment style in sexual offenders. *Journal of Interpersonal Violence*, *12*, 323-339.

- Hurd, P. L., Vaillancourt, K. L., & Dinsdale, N. L. (2011). Aggression, digit ratio and variation in androgen receptor and monoamine oxidase A genes in men. *Behavior Genetics*, *41*, 543-556.
- Iacobucci, D. (2010). Structural equations modeling: Fit Indices, sample size, and advanced topics. *Journal of Consumer Psychology*, 20, 90-98.
- Infante, D. A., & Wigley, C. J. (1986). Verbal aggressiveness: An interpersonal model and measure. *Communication Monographs*, 53, 61-69.
- Irvine, R.A., Ma, H., Yu, M.C., Ross, R.K., Stallcup, M.R., & Coetzee, G.A. (2000). Inhibition of p160-mediated coactivation with increasing androgen receptor polyglutamine length. *Human Molecular Genetics*, 9, 267-274.
- Jackson, D. N. (2004). *Jackson Personality Inventory-Revised*. Port Huron, MI: Research Psychologists Press.
- Jackson, D. N. (2003). *Personality Research Form*. Port Huron, MI: Research Psychologists Press.
- Jackson, D. N. (1999). *Personality Research Form Manual*. Port Huron, MI: Research Psychologists Press.
- Jackson, D. N. (1994). *Jackson Personality Inventory-Revised Manual*. Port Huron, MI: Research Psychologists Press.
- Jackson, J. J., & Kirkpatrick, L. A. (2007). The structure and measurement of human mating strategies: Towards a multidimensional model of sociosexuality. *Evolution and Human Behavior, 28*, 382-391.
- James, J., Ellis, B. J., Schlomer, G. L., & Garber, J. (2012). Sex-specific pathways to early puberty, sexual debut, and sexual risk taking: Tests of an integrated evolutionary–developmental model. *Developmental Psychology*, 48, 687-702.
- Jaremka, L. M., Glaser, R., Loving, T. J., Malarkey, W. B., Stowell, J. R., & Kiecolt-Glaser, J. K. (2013). Attachment anxiety is linked to alterations in cortisol production and cellular immunity. *Psychological Science*, 24, 272-279.
- Johnson, R. T., Burk, J. A., & Kirkpatrick, L. A. (2007). Dominance and prestige as differential predictors of aggression and testosterone levels in men. *Evolution and Human Behavior, 28*, 345-351.
- Johnston, M. A. (1999). Influences of adult attachment in exploration. *Psychological Reports*, 84, 31-34.
- Jönsson, E. G., von Gertten, C., Gustavsson, J. P., Yuan, Q.-P., Lindblad-Toh, K.,

Forslund, K., et al. (2001). Androgen receptor trinucleotide repeat polymorphism and personality traits. *Psychiatric Genetics*, 11, 19-23.

- Kalin, N. H. (1999). Primate models to understand human aggression. *The Journal of Clinical Psychiatry*, 60, 29-32.
- Kaplan, H. S., & Gangestad, S. W. (2005). Life history theory and evolutionary psychology. In D. M. Buss (Ed.), *Handbook of Evolutionary Psychology* (pp. 68-95). Hoboken, NJ: Wiley.
- Karavasilis, L., Doyle, A. B., & Markiewicz, D. (2003). Associations between parenting style and attachment to mother in middle childhood and adolescence. *International Journal of Behavioral Development*, 27, 153-164.
- Kendall, D. A., Stancel, G. M., & Enna, S. J. (1982). The influence of sex hormones on antidepressant-induced alterations in neurotransmitter receptor binding. *Journal of Neuroscience*, 2, 354-360.
- Kim, K., & Smith, P. K. (1998). Retrospective survey of parental marital relations and child reproductive development. *International Journal of Behavioral Development*, 22, 729-751.
- Kim, K., Smith, P. K., & Palermiti, A.-L. (1997). Conflict in childhood and reproductive development. *Evolution and Human Behavior*, 18, 109-142.
- King, B. E., & Allgeier, E. R. (2000). The Sexual Desire Inventory as a measure of sexual motivation in college students. *Psychological Reports*, 86, 347–350.
- Kirkpatrick, L. A. (1998). Evolution, pair bonding, and reproductive strategies: A reconceptualization of adult attachment. In J. A. Simpson & W. S. Rholes (Eds.), *Attachment Theory and Close Relationships* (pp. 353-393). New York, NY: Guilford Press.
- Kirkpatrick, L. A., & Davis, K. E. (1994). Attachment style, gender, and relationship stability: A longitudinal analysis. *Journal of Personality and Social Psychology*, *66*, 502-512.
- Kirkpatrick, L. A., & Hazan, C. (1994). Attachment styles and close relationships: A four-year prospective study. *Personal Relationships*, 1, 123-142.
- Kline, R. B. (2004, 2011). *Principles and Practices of Structural Equation Modeling*. New York, NY: The Guilford Press.
- Knussmann, R. & Christiansen, K. (1989). Attributes of masculinity and androgen measures. *Homo*, 39, 45-50.

- Kouri, E. M., Lukas, S. E., Pope Jr., H. G., & Oliva, P. S. (1995). Increased aggressive responding in male volunteers following the administration of gradually increasing doses of testosterone cypionate. *Drug and Alcohol Dependence*, 40, 73-79.
- Krithivas, K., Yurgalevitch, S. M., Mohr, B. A., Wilcox, C. J., Batter, S. J., Brown, M., et al. (1999). Evidence that the CAG repeat in the androgen receptor gene is associated with the age-related decline in serum androgen levels in men. *Journal* of Endocrinology, 162, 137-142.
- Kritzer, M. F. (2004). The distribution of immunoreactivity for intracellular androgen receptors in the cerebral cortex of hormonally intact adult male and female rats: Localization in pyramidal neurons making corticocortical connections. *Cerebral Cortex, 14*, 268-280.
- Kuepper, Y., & Hennig, J. (2007). Behavioral aggression is associated with the 2D:4D ratio in men but not in women. *Journal of Individual Differences, 28*, 64-72.
- Kuzawa, C. W., Gettler, L. T., Muller, M. N., McDade, T. W., & Feranil, A. B. (2009). Fatherhood, pairbonding and testosterone in the Philippines. *Hormones and Behavior*, 56, 429-435.
- Kuzawa, C. W., McDade, T. W., Adair, L. S., & Lee, N. (2010). Rapid weight gain after birth predicts life history and reproductive strategy in Filipino males. *Proceedings of the National Academy of Science*, 107, 16800-16805.
- Laberge, G., Lussier, Y., & Godbout, N. (2012, August). Correlates of sexual risk taking behavior in young adults. Poster session presented at the annual convention of the American Psychological Association, Orlando, Florida.
- Lafontaine, M.-F., & Lussier, Y. (2005). Does anger towards the partner mediate and moderate the link between romantic attachment and intimate violence? *Journal of Family Violence, 20*, 349-362.
- La Grange, L., Jones, T. D., Erb, L., & Reyes, E. (1995). Alcohol consumption: Biochemical and personality correlates in a college student population. *Addictive Behaviors*, 20, 93-103.
- Lakatos, K., Toth, I., Nemoda, Z., Ney, K., Sasvari-Szekely, M., & Gervai, J. (2000). Dopamine D4 receptor (DRD4) polymorphism is associated with attachment disorganization in infants. *Molecular Psychiatry*, 5, 633-637.
- Lawson, D. M. (2008). Attachment, interpersonal problems, and family of origin functioning: Differences between partner violent and nonpartner violent men. *Psychology of Men & Masculinity, 9*, 90-105.

- Leak, G. K., & Parsons, C. J. (2001). The susceptibility of three attachment style measures to socially desirable responding. *Social Behavior and Personality*, 29, 21-30.
- Lei, M., & Lomax, R. G. (2005). The effect of varying degrees of nonnormality in structural equation modeling. *Structural Equation Modeling*, *12*, 1-27.
- Letcher, A., & Slesnick, N. (2014). Early relationships of adolescents: Evaluation and observation of romantic attachment and risk behavior. *Journal of Social and Personal Relationships*, *31*, 366-383.
- Levy, K. N., Blatt, S. J., & Shaver, P. R. (1998). Attachment styles and parental representations. *Journal of Personality and Social Psychology*, *74*, 407-419.
- Lewis, M. (1997). *Altering Fate: Why the Past Does not Predict the Future*. New York, NY: Guilford Press.
- Liu, J., Portnoy, J., & Raine, A. (2012). Association between a marker for prenatal testosterone exposure and externalizing behavior problems in children. *Development and Psychopathology*, 24, 771-782.
- Loo, R., & Thorpe, K. (2000). Confirmatory factor analyses of the full and short versions of the Marlowe–Crowne Social Desirability Scale. *Journal of Social Psychology*, 140, 628-635.
- Lucion, A. B., Charchat, H., Pereira, G. A., & Rasia-Filho, A. A. (1996). Influence of early postnatal gonadal hormones on anxiety in adult male rats. *Physiology and Behavior*, 60, 1419-1423.
- Lumia, A. R., Thorner, K. M., & McGinnis, M. Y. (1994). Effects of chronically high doses of the anabolic androgenic steroid, testosterone, on intermale aggression and sexual behavior in male rats. *Physiology and Behavior*, 55, 331-335.
- Lund, M. (1985). The development of investment and commitment scales for predicting continuity of personal relationships. *Journal of Social and Personal Relationships*, *2*, 3-23.
- Lutchmaya, S., Baron-Cohen, S., Raggat, P., Knickmeyer, R., & Manning, J. T. (2004). 2nd to 4th digit ratios, fetal testosterone and estradiol. *Early Human Development*, 77, 23-28.
- MacCallum, R. C., & Browne, M. W. (1993). The use of causal indicators in covariance structure models: Some practical issues. *Psychological Bulletin*, 114, 533-541.
- MacKinnon, D. P. (2008). *Introduction to statistical mediation analysis*. Mahwah, NJ: Erlbaum.

- MacKinnon, D. P., Fairchild, A. J., & Fritz, M. S. (2007). Mediation analysis. *Annual Review of Psychology*, 58, 593-614.
- Main, M., Kaplan, N., & Cassidy, J. (1985). Security in infancy, childhood, and adulthood: A move to the level of representation. In I. Bretherton & E. Waters (Eds.), Growing points of attachment theory and research. *Monographs of the Society for Research in Child Development, 50*, (1-2, Serial No. 209), 66-104.
- Main, M., & Solomon, J. (1990). Procedures for identifying infants as disorganized/disoriented during the Ainsworth Strange Situation. In M. T. Greenberg, D. Cicchetti, & E. M. Cummings (Eds.), Attachment in the preschool years: Theory, research, and intervention (pp. 121-160). Chicago, IL: University of Chicago Press.
- Main, M. & Goldwyn, R. (1984). *Adult Attachment Scoring and Classification System*. Berkeley, CA: University of California.
- Maki, P. M., Ernst, M., London, E. D., Mordecai, K. L., Perschler, P., Durso, S. C., et al. (2007). Intramuscular testosterone treatment in eldery men : Evidence of memory decline and altered brain function. *Journal of Clinical Endocrinology and Metabolism*, 92, 4107-4114.
- Maner, J. K., Miller, S. L., Coyle, J. M., & Kaschak, M. P. (2014). Confronting intrasexual rivals: 2D:4D digit ratio predicts behavioral and endocrinological responses to infidelity threat. *Social Psychological and Personality Science*, 5, 119-128.
- Maner, J. K., Miller, S. L., Schmidt, N. B., & Eckel, L. A. (2008). Submitting to defeat: Social anxiety, dominance threat, and decrements in testosterone. *Psychological Science*, 19, 764-768.
- Manni, A., Pardridge, W. M., Cefalu, W., Nisula, B. C., Bardin, C. W., Santner, S. J., et al. (1985). Bioavailability of albumin-bound testosterone. *Journal of Clinical Endocrinology and Metabolism*, 61, 705-710.
- Manning, J. T., Bundred, P. E., Newton, D. J., & Flanagan, B. F. (2003). The second to fourth digit ratio and variation in the androgen receptor gene. *Evolution and Human Behavior*, 24, 399-405.
- Manning, J. T. & Fink, B. (2008). Digit ratio (2D:4D), dominance, reproductive success, asymmetry, and sociosexuality in the BBC Internet Study. *American Journal of Human Biology*, 20, 451-461.
- Manning, J. T., Scutt, D., Wilson, J., & Lewis-Jones, D. I. (1998). The ratio of 2nd to 4th digit length: A predictor of sperm numbers and concentrations of

testosterone, luteinizing hormone and oestrogen. *Human Reproduction*, 13, 3000-3004.

- Marganska, A., Gallagher, M., & Miranda, R. (2013). Adult attachment, emotion dysregulation, and symptoms of depression and generalized anxiety disorder. *American Journal of Orthopsychiatry*, 83, 131-141.
- Marsh, H. W., & Hau, K.-T. (1999). Confirmatory factor analysis: Strategies for small sample sizes. In R. H. Hoyle (Ed.), *Statistical Strategies for Small Sample Research* (pp. 252-284). Thousand Oaks, CA: Sage.
- Martin, S. M., Manning, J. T., & Dowrick, C. F. (1999). Fluctuating asymmetry, relative digit length and depression in men. *Evolution and Human Behavior*, 20, 203-214.
- Mauricio, A. M., & Gormley, B. (2001). Male perpetration of physical violence against female partners. The interaction of dominance needs and attachment insecurity. *Journal of Interpersonal Violence*, 16, 1066-1081.
- Mazur, A., Booth, A. (2014). Testosterone is related to deviance in male army veterans, but relationships are not moderated by cortisol. *Biological Psychology*, *96*, 72-76.
- Mazur, A., & Michalek, J. (1998). Marriage, divorce, and male testosterone. *Social Forces*, 77, 315-330.
- McCarthy, J. D., & Hoge, D. R. (1984). The dynamics of self-esteem and delinquency. *American Journal of Sociology*, 90, 396-412.
- McIntyre, M. H. (2006). The use of digit ratios as markers of perinatal androgen action. *Reproductive Biology and Endocrinology*, *4*, 10-18.
- McIntyre, M. H., Barrett, E. S., McDermott, R., Johnson, D. D. P., Cowden, J., & Rosen, S. P. (2007). Finger length ratio (2D:4D) and sex differences in aggression during a simulated war game. *Personality and Individual Differences*, 42, 755-764.
- McIntyre, M., Gangestad, S. W., Gray, P. B., Chapman, J. F., Burnham, T. C., O'Rourke, M. T., et al. (2006). Romantic involvement often reduces men's testosterone levels—but not always: The moderating role of extrapair sexual interest. *Journal* of Personality and Social Psychology, 91, 642-651.
- McHenry, J., Carrier, N., Hull, E., & Kabbaj, M. (2014). Sex differences in anxiety and depression: Role of testosterone. *Frontiers in Neuroendocrinology*, *35*, 42-57.
- McLanahan, S., & Bumpass, L. (1988). Intergenerational consequences of family disruption. *American Journal of Sociology*, 94, 130-152.

McNair, D. M., & Heuchert, JW. P. (2005). Profile of Mood States: Technical Update.

San Diego, CA: Educational and Industrial Testing Service.

- McNair, D.M., Lorr, M., & Droppleman, L.F. (1992). *Profile of Mood States*. San Diego, CA: Educational and Industrial Testing Service.
- McNair, D. M., Lorr, M., Heuchert, JW. P., & Droppleman, L.F. (2003). *Profile of Mood States*. San Diego, CA: Educational and Industrial Testing Service.
- Meckelmann, V., Pfeifer, C., & Rauh, H. (2013). Family relationships in childhood, pubertal timing, and subsequent reproductive strategies among adolescents. *South African Journal of Psychology*, *43*, 22-33.
- Medland, S. E., Duffy, D. L., Spurdle, A. B., Wright, M. J., Geffen, G. M., Montgomery, G. W., et al. (2005). Opposite effects of androgen receptor CAG repeat length on increased risk of left-handedness in males and females. *Behavior Genetics*, 35, 735-744.
- Medland, S.E., Zayats, T., Glaser, B., Nyholt, D.R., Gordon, S.D., Wright, M.J., et al. (2010). A variant in *LIN28B* is associated with 2D:4D finger-length ratio, a putative retrospective biomarker of prenatal testosterone exposure. *American Journal of Human Genetics*, 86, 519-525.
- Mendelson, S. D. & McEwen, B. S. (1990). Chronic testosterone propionate treatment decreases the concentration of [3H]quipazine binding at 5-HT 3 receptors in the amygdala of the castrated male rat. *Brain Research*, 528, 339-343.
- Meschke, L. L., Johnson, P. J., Barber, B. L., & Eccles, J. S. (2003). Psychosocial factors predicting pubertal onset. In C. Hayward (Ed.), *Gender Differences at Puberty* (pp. 217–238). New York, NY: Cambridge University Press.
- Meston, C. M., Trapnell, P. D., & Gorzalka, B. B. (1996). Ethnic and gender differences in sexuality: Variations in sexual behavior between Asian and non-Asian university students. *Archives of Sexual Behavior*, 25, 33–72.
- Mickelson, K. D., Kessler, R. C., & Shaver, P. R. (1997). Adult attachment in a nationally representative sample. *Journal of Personality and Social Psychology*, 73, 1092-1106.
- Mikulincer, M., (1998). Adult attachment style and individual differences in functional versus dysfunctional experiences of anger. *Journal of Personality and Social Psychology*, 74, 513-524.
- Mikulincer, M., & Shaver, P. R. (2007). *Attachment in Adulthood: Structure, Dynamics, and Change*. New York, NY: Guilford Press.
- Miller, W. B., & Pasta, D. J. (2000). Early family environment, reproductive strategy,

and contraceptive behavior: Testing a genetic hypothesis. In Rodgers, J. L., Rowe, D. C., & Miller, W. B. (Eds.), *Genetic Influences on Human Fertility and Sexuality. Theoretical and Empirical Contributions from the Biological and Behavioral Sciences* (pp. 183-230). Norwell, MA: Kluwer Academic Publishers.

- Minnis, H., Reekie, J., Young, D., O'Connor, T., Ronald, A., Gray, A., et al. (2007). Genetic, environmental and gender influences on attachment disorder behaviours. *British Journal of Psychiatry*, 190, 490-495.
- Moffat, S. D., & Hampson, E. (1996). A curvilinear relationship between testosterone and spatial cognition in humans: Possible influence of hand preference. *Psychoneuroendocrinology*, 21, 323-337.
- Moffat, S. D., & Hampson, E. (2000). Salivary testosterone concentrations in lefthanders: An association with cerebral language lateralization? *Neuropsychology*, *14*, 71-81.
- Moffat, S. D., & Resnick, S. M. (2007). Long-term measures of free testosterone predict regional cerebral blood flow patterns in elderly men. *Neurobiology of Aging, 28*, 914-920.
- Moffitt, T. E., Caspi, A., Belsky, J., & Silva, P. (1992). Childhood experience and the onset of menarche: A test of a sociobiological model. *Child Development*, *63*, 47-58.
- Moss, E., Rousseau, D., Parent, S., St-Laurent, D., & Saintonge, J. (1998). Correlates of attachment at school age: Maternal reported stress, mother-child interaction, and behavior problems. *Child Development*, 69, 1390-1405.
- Muller, M. N., Marlowe, F. W., Bugumba, R., & Ellison, P. T. (2009). Testosterone and paternal care in East African foragers and pastoralists. *Proceedings of the Royal Society B: Biological Sciences*, 276, 347-354.
- Murphy, B., & Bates, G. W. (1997). Adult attachment style and vulnerability to depression. *Personality and Individual Differences*, 22, 835-844.
- Mustanski, B. S., Viken, R. J., Kaprio, J., Pulkkinen, L., & Rose, R. J. (2004). Genetic and environmental influences on pubertal development: Longitudinal data from Finnish twins at ages 11 and 14. *Developmental Psychology*, 40, 1188-1198.
- Nair, H., & Murray, A. D. (2005). Predictors of attachment security in preschool children from intact and divorced families. *The Journal of Genetic Psychology*, 166, 245-263.

- Nakamura, N., Fujita, H., & Kawata, M. (2002). Effects of gonadectomy on immunoreactivity for choline acetyltransferase in the cortex, hippocampus, and basal forebrain of adult male rats. *Neuroscience*, *109*, 473-485.
- Neave, N., Laing, S., Fink, B., & Manning, J. T. (2003). Second to fourth digit ratio, testosterone and perceived male dominance. *Proceedings of the Royal Society of London, Series B, 270,* 2167-2172.
- Nelson, E., Hoffman, C. L., Gerald, M. S., & Shultz, S. (2010). Digit ratio (2D:4D) and dominance rank in female rhesus macaques (*Macaca mulatta*). *Behavioural Ecology and Sociobiology*, 64, 1001-1009.
- Nenonen, H., Björk, C., Skjaerpe, P. A., Giwercman, A., Rylander, L., Svartberg, J., et al. (2010a). CAG repeat number is not inversely associated with androgen receptor activity *in vitro*. *Molecular Human Reproduction*, 16, 153-157.
- Nenonen, H. A., Giwercman, A., Hallengren, E., & Giwercman, Y. L. (2010b). Nonlinear association between androgen receptor CAG repeat length and risk of male subfertility—a meta-analysis. *International Journal of Andrology*, 33, 1-6.
- Nyenhuis, D. L., Yamomoto, C., Luchetta, T., Terrien, A., & Parmentier, A. (1999). Adult and geriatric normative data and validation of the profile of mood states. *Journal of Clinical Psychology*, 55, 79-86.
- O'Connor, D. B., Archer, J., Hair, W. M., & Wu, C. W. (2002). Exogenous testosterone, aggression, and mood in eugonadal and hypogonadal men. *Physiology and Behavior*, *75*, 557-566.
- O'Connor, D. B., Archer, J., & Wu, F. C. (2004). Effects of testosterone on mood, aggression, and sexual behavior in young men: a double-blind, placebo-controlled, cross-over study. *Journal of Clinical Endocrinology and Metabolism, 89*, 2837-2845.
- Okazaki, S. (2002). Influence of culture on Asian Americans' sexuality. *The Journal of Sex Research*, *39*, 34-41.
- Ökten, A., Kalyoncu, M., & Yaris, N. (2002). The ratio of second- and fourth-digit lengths and congenital adrenal hyperplasia due to 21-hydroxylase deficiency. *Early Human Development*, *70*, 47-54.
- Ortner, G. R., Wibral, M., Becker, A., Dohmen, T., Klingmüller, D., Falk, A., et al. (2013). No evidence for an effect of testosterone administration on delay discounting in male university students. *Psychoneuroendocrinology*, 38, 1814-1818.

- Osborne, J. W. (2013). *Best Practices in Data Cleaning*. Thousand Oaks, CA: SAGE Publications, Inc.
- Oskis, A., Loveday, C., Hucklebridge, F., Thorn, L., & Clow, A. (2011). Anxious attachment style and salivary cortisol dysregulation in healthy female children and adolescents. *Journal of Child Psychology and Psychiatry*, *52*, 111-118.
- Owen, M. T., & Cox, M. J. (1997). Marital conflict and the development of infantparent attachment relationships. *Journal of Family Psychology*, 11, 152-164.
- Panksepp, J. (1988). Affective Neuroscience: The Foundations of Human and Animal Emotions. New York, NY: Oxford University Press.
- Pardridge, W. M., & Demers, L. M. (1991). Bioavailable testosterone in salivary glands. *Clinical Chemistry*, 37, 139-140.
- Parkes, C. M. (2006). *Love and Loss. The Roots of Grief and its Complications*. New York, NY: Routledge.
- Pasterski, V., Hindmarsh, P., Geffner, M., Brook, C., Brain, C., & Hines, M. (2007). Increased aggression and activity level in 3- to 11-year-old girls with congenital adrenal hyperplasia. *Hormones and Behavior*, 52, 368-374.
- Paul, E. L., McManus, B., & Hayes, A. (2000). "Hookups": characteristics and correlates of college students' spontaneous and anonymous sexual experiences. *The Journal of Sex Research*, 37, 76–88.
- Paunonen, S. V. (1998). Hierarchical organization of personality and prediction of behavior. *Journal of Personality and Social Psychology*, 74, 538-556.
- Penke, L., & Asendorpf, J. B. (2008). Beyond global sociosexual orientation: A more differentiated look at sociosexuality and its effects on courtship and romantic relationships. *Journal of Personality and Social Psychology*, 95, 1113-1135.
- Peters, M., Simmons, L., & Rhodes, G. (2008). Testosterone is associated with mating success but not attractiveness or masculinity in human males. *Animal Behavior*, 76, 297–303.
- Phoenix, C. H., Goy, R. W., Gerall, A. A., & Young, W. C. (1959). Organizing action of prenatally administered testosterone proprionate on the tissues mediating mating behaviour in the female guinea pig. *Endocrinology*, 65, 369-382.
- Picardi, A., Vermigli, P., Toni, A., D'Amico, R., Bitetti, D., & Pasquini, P. (2002).
 Further evidence of the validity of the Italian version of the questionnaire
 "Experiences in Close Relationships" (ECR), a self-report instrument to assess adult attachment. *Italian Journal of Psychopathology*, *8*, 282-294.

- Pollet, T. V., Cobey, K. D., & van der Meij, L. (2013). Testosterone levels are negatively associated with fatherhood in males, but positively related to offspring count in fathers. *PLoS ONE*, *8*, 4.
- Pollet, T. V., van der Meij, L., Cobey, K. D., & Buunk, A. P. (2011). Testosterone levels and their associations with lifetime number of opposite sex partners and remarriage in a large sample of American elderly men and women. *Hormones* and Behavior, 60, 72-77.
- Pope Jr., H.G., Kouri, E.M., & Hudson, J.I. (2000). Effects of supraphysiological doses of testosterone on mood and aggression in normal men: a randomized controlled trial. Archives of General Psychiatry, 57, 133-140.
- Pouliot, W. A., Handa, R. J., & Beck, S. G. (1996). Androgen moduleates *N*-methyl-Dasparate-mediated depolarization in CA1 hippocampal pyramidal cells. *Synapse*, 23, 10-19.
- Preacher, K. J., & Hayes, A. F. (2008). Asymptotic and resampling strategies for assessing and comparing indirect effects in multiple mediator models. *Behavior Research Methods*, 40, 879-891.
- Preacher, K. J., & Kelley, K. (2011). Effect size measures for mediation models: Quantitative strategies for communicating indirect effects. *Psychological Methods*, 16, 93-115.
- Puts, D. A., Gaulin, S. J. C., Sporter, R. J., & McBurney, D. H. (2004). Sex hormones and finger length. What does 2D:4D indicate? *Evolution and Human Behavior*, 25, 182–199.
- Puts, D. A., Pope, L. E., Hill, A. K., Cárdenas, R. A., Welling, L. L. M., Wheatley, J. R. et al. (2015). Fulfilling Desire: Evidence for negative feedback between men's testosterone, sociosexual psychology, and sexual partner number. *Hormones and Behavior*, 70, 14-21.
- Puy, L., MacLusky, N. J., Becker, L., Karsan, N., Trachtenberg, J., & Brown, T. J. (1995). Immunocytochemical detection of androgen receptor in human temporal cortex: Characterization and application of polyclonal androgen receptor antibodies in frozen and paraffin-embedded tissues. *Journal of Steroid Biochemistry and Molecular Biology*, 55, 197-209.
- Quinton, D., Rutter, M., & Little, C. (1984). Institutional rearing, parenting difficulties, and marital support. *Psychological Medicine*, 14, 107-124.
- Quirin, M., Pruessner, J. C., & Kuhl, J. (2008). HPA system regulation and adult attachment anxiety: Individual differences in reactive and awakening cortisol. *Psychoneuroendocrinology*, *33*, 581-590.

- Quissell, D. O. (1993). Steroid hormone analysis in human saliva. *Annals of the New York Academy of Science, 694*, 143-145.
- Rahman, Q., Korhonen, M., & Aslam, A. (2005). Sexually dimorphic 2D:4D ratio, height, weight, and their relationship to number of sexual partners. *Personality* and Individual Differences, 39, 83-92.
- Rajender, S., Pandu, G., Sharma, J. D., Gandhi, K. P. C., Singh, L. & Thangaraj, K. (2008). Reduced CAG repeats length in androgen receptor gene is associated with violent criminal behavior. *International Journal of Legal Medicine*, 122, 367-372.
- Ranta, M., Dietrich, J., & Salmela-Aro, K. (2014). Career and romantic relationship goals and concerns during emerging adulthood. *Emerging Adulthood*, *2*, 17-26.
- Reavis, R. & Overman, W.H. (2001). Adult sex differences on a decision-making task previously shown to depend on the orbital prefrontal cortex. *Behavioral Neuroscience*, 115, 196-206.
- Rejeski, W. J., Brubaker, P. H., Herb, R., A., Kaplan, J. R., & Koritnik, D. (1988). Anabolic steroids and aggressive behavior in cynomolgus monkeys. *Journal of Behavioral Medicine*, 11, 95-105.
- Reiner, I., & Spangler, G. (2010). Adult attachment and gene polymorphisms of the dopamine D4 receptor and serotonin transporter (5-HTT). *Attachment & Human Development*, 12, 209-229.
- Reinisch, J. M. (1981). Prenatal exposure to synthetic progestins increases potential for aggression in humans. *Science*, 211, 1171-1173.
- Renken, B., Egeland, B., Marvinney, D., Manglesdorf, S., & Sroufe, L. A. (1989). Early childhood antecedents of aggression and passive-withdrawal in early elementary school. *Journal of Personality*, 57, 257-281.
- Repetti, R. L., Taylor, S. E., & Seeman, T. E. (2002). Risky families: Family social environments and the mental and physical health of offspring. *Psychological Bulletin*, *128*, 330-366.
- Reynolds, W. M. (1982). Development of reliable and valid short forms of the Marlowe-Crowne social desirability scale. *Journal of Clinical Psychology*, *38*, 119-125.
- Rifkin-Graboi, A. (2008). Attachment status and salivary cortisol in a normal day and during simulated interpersonal stress in young men. *Stress: The International Journal on the Biology of Stress, 11*, 210-224.
- Riggs, S. A., & Kaminski, P. (2010). Childhood emotional abuse, adult attachment, and

depression as predictors of relational adjustment and psychological aggression. *Journal of Aggression, Maltreatment and Trauma, 19*, 75–104.

- Rivas, M. P., Moreira, L. M. A., Santo, L. D. E., Marques, A. C. S. S., El-Hani, C. N., & Toralles, M. B. P. (2014). New studies of second and fourth digit ratio as a morphogenetic trait in subjects with congenital adrenal hyperplasia. *American Journal of Human Biology*, 26, 559-561.
- Roberts, J. E., Gotlib, I. H., & Kassel, J. D. (1996). Adult attachment security and symptoms of depression: The mediating roles of dysfunctional attitudes and low self-esteem. *Journal of Personality and Social Psychology*, 70, 310-320.
- Roberts, N., & Noller, P. (1998). The associations between adult attachment and couple violence: The role of communication patterns and relationship satisfaction. In J. A. Simpson & W. S. Rholes (Eds.), *Attachment Theory and Close Relationships* (pp. 317-350). New York, NY: Guilford Press.
- Robins, R. W., Hendin, H. M., & Trzesniewski, K. H. (2001). Measuring global selfesteem: construct validation of a single-item measure and the Rosenberg selfesteem scale. *Personality and Social Psychology Bulletin*, 27, 151-161.
- Rogers, N. L., Cole, S. A., Lan, H-C., Crossa, A., & Demerath, E. W. (2007). New saliva DNA collection method compared to buccal cell collection techniques for epidemiological studies. *American Journal of Human Biology*, 19, 319-326.
- Roisman, G. I., Holland, A., Fortuna, K., Fraley, R. C., Clausell, E., & Clarke, A. (2007). The adult attachment interview and self-reports of attachment style: An empirical rapprochement. *Journal of Personality and Social Psychology*, *92*, 678-697.
- Roney, J. R. & Maestripieri, D. (2004). Relative digit lengths predict men's behavior and attractiveness during social interactions with women. *Human Nature*, 15, 271-282.
- Roney, J. R., Simmons, Z. L., & Lukaszewski, A. W. (2010). Androgen receptor gene sequence and basal cortisol concentrations predict men's hormonal responses to potential mates. *Proceedings of the Royal Society B: Biological Sciences, 227*, 57-63.
- Rosenberg, M. (1965). Society and the Adolescent Self-Image. Princeton, NJ: Princeton University Press.
- Rosenstein, D. S., & Horowitz, H. A. (1996). Adolescent attachment and psychopathology. *Journal of Consulting and Clinical Psychology*, 64, 244-253.
- Ross, L. T., & Hill, E. M. (2002). Childhood unpredictability, schemas for unpredictability, and risk taking. *Social Behaviour and Personality*, *30*, 453-474.

- Roth, P. L., & Switzer, F. S. (1995). A monte carlo analysis of missing data techniques in a HRM setting. *Journal of Management*, *21*, 1003–1023.
- Roth, P. L., Switzer, F. S., & Switzer, D. M. (1999). Missing data in multiple item scales: A monte carlo analysis of missing data techniques. *Organizational Research Methods*, 2, 211-212.
- Rubinow, D. R., & Schmidt, P. J. (1996). Androgens, brain, and behavior. *American Journal of Psychiatry*, 153, 974-984.
- Salo, J., Jokela, M., Lehtimäki, T., & Keltikangas-Järvinen, L. (2011). Serotonin receptor 2A gene moderates the effect of childhood maternal nurturance on adulthood social attachment. *Genes, Brain and Behavior, 10*, 702-709.
- Sankar, J. S., & Hampson, E. (2012). Testosterone levels and androgen receptor gene polymorphism predict specific symptoms of depression in young men. *Gender Medicine*, 9, 232-243.
- Sannikka, E., Terho, P., Suominen, J., & Santi, R. (1983). Testosterone concentrations in human seminal plasma and saliva and its correlation with non-protein-bound and total testosterone levels in serum. *International Journal of Andrology*, 6, 319-330.
- Sapienza, P., Zingales, L., & Maestripieri, D. (2009). Gender differences in financial risk aversion and career choices are affected by testosterone. *Proceedings of the National Academy of Sciences*, 106, 15268-15273.
- Sarkisian, N. (2010). Introduction to structural equation modeling. Retrieved from: http://www.sarkisian.net/sc705/sem_intro.pdf
- Schachner, D. A., & Shaver, P. R. (2004). Attachment dimensions and sexual motives. *Personal Relationships*, 11, 179-195.
- Scharfe, E., & Bartholomew, K. (1994). Reliability and stability of adult attachment patterns. *Personal Relationships*, 1, 23-43.
- Scheffer, D., Chasiotis, A., Restemeier, R., Keller, H., & Schölmerich, A. (2000). [Retrospective measurement of representational aspects of childhood relationships: Construction of the Relationship Context Scale]. Zeitschrift für Entwicklungspsychologie und Pädagogische Psychologie, 32, 2–13.
- Schindler, A., Thomasius, R., Sack, P.-M., Gemeinhardt, B., Küstner, U., & Eckert, J. (2005). Attachment and substance use disorders: A review of the literature and a study in drug dependent adolescents. *Attachment & Human Development*, 7, 207-228.

Schindler, I., Fagundes, C. P., & Murdock, K. W. (2010). Predictors of romantic

relationship formation: Attachment style, prior relationships, and dating goals. *Personal Relationships, 17*, 97-105.

- Schmidt, P. J., Berlin, K. L., Danaceau, M.A., Neeren, A., Haq, N. A., Roca, C. A., et al. (2004). The effects of pharmacologically induced hypogonadism on mood in healthy men. *Archives of General Psychiatry*, 61, 997-1004.
- Schmitt, D. P. (2005). Is short-term mating the maladaptive result of insecure attachment? A test of competing evolutionary perspectives. *Personality and Social Psychology Bulletin*, 31, 747-768.
- Schmitt, D. P., Alcalay, L., Allensworth, M., Allik, J., Ault, L., Austers, I., et al. (2003). Are men universally more dismissing than women? Gender differences in romantic attachment across 62 cultural regions. *Personal Relationships*, 10, 307-331.
- Schneider, G., Nienhaus, K., Gromoll, J., Heuft, G., Nieschlag, E., & Zitzmann, M. (2010). Aging males' symptoms in relation to the genetically determined androgen receptor CAG polymorphism, sex hormone levels and sample membership. *Psychoneuroendocrinology*, 35, 578-587.
- Schneider, G., Nienhaus, K., Gromoll, J., Heuft, G., Nieschlag, E., & Zitzmann, M. (2011). Depressive symptoms in men aged 50 years and older and their relationship to genetic androgen receptor polymorphism and sex hormone levels in three different samples. *American Journal of Geriatric Psychiatry*, 19, 274-283.
- Schulz, K. M., Molenda-Figueira, H. A., & Sisk, C. L. (2009). Back to the future: The organizational-activational hypothesis adapted to puberty and adolescence. *Hormones and Behavior*, 55, 597-604.
- Schwarz, S., Mustafić, M., Hassebrauck, M., & Jörg, J. (2011). Short- and long-term relationship orientation and 2D:4D finger-length ratio. *Archives of Sexual Behavior*, 40, 565-574.
- Schwerdtfeger, A., Heims, R., & Heer, J. (2010). Digit ratio (2D:4D) is associated with traffic violations for male frequent car drivers. *Accident Analysis and Prevention*, 42, 269-274.
- Scott, L. N., Levy, K. N., & Pincus, A. L. (2009). Adult attachment, personality traits, and borderline personality disorder features in young adults. *Journal of Personality Disorders*, 23, 258-280.
- Sellers, J. G., Mehl, M. R., & Josephs, R. A. (2007). Hormones and personality: Testosterone as a marker of individual differences. *Journal of Research in Personality*, 41, 126-138.

- Senchak, M., & Leonard, K. E. (1992). Attachment styles and marital adjustment among newlywed couples. *Journal of Social and Personal Relationships*, 9, 51-64.
- Shaver, P. R., & Mikulincer, M. (2002). Attachment –related psychodynamics. *Attachment and Human Development, 4*, 133-161.
- Shaw, A. Z., Kotowski, M. R., Boster, F. J., & Levine, T. R. (2012). The effect of prenatal sex hormones on the development of verbal aggression. *Journal of Communication*, 62, 778-793.
- Shrout, P. E., & Bolger, N. (2002). Mediation in experimental and nonexperimental studies: New procedures and recommendations. *Psychological Methods*, 7, 422-445.
- Sibley, C. G., Fischer, R., & Liu, J. H. (2005). Reliability and validity of the revised experiences in close relationships (ECR-R) self-report measure of adult romantic attachment. *Personality and Social Psychology Bulletin, 31*, 1524-1536.
- Simmons, Z. L., & Roney, J. R. (2011). Variation in CAG repeat length of the androgen receptor gene predicts variables associated with intrasexual competitiveness in human males. *Hormones and Behavior*, 60, 306-312.
- Simonelli, L., Ray, W., & Pincus, A. L. (2004). Attachment models and their relationships with anxiety, worry and depression. *Counseling and Clinical Psychology Journal*, 1, 107-118.
- Simpson, J. A. (1990). Influence of attachment styles on romantic relationships. *Journal* of Personality and Social Psychology, 59, 971–980.
- Simpson, J. A., & Belsky, J. (2008). Attachment theory within a modern evolutionary framework. In J. Cassidy & P. R. Shaver (Eds.), *Handbook of Attachment. Theory, Research, and Clinical Applications* (pp. 131-157). New York, NY: Guilford Press.
- Simpson, J. A., & Gangestad, S. W. (1991). Individual differences in sociosexuality: evidence for convergent and discriminant validity. *Journal of Personality and Social Psychology*, 60, 870-883.
- Simpson, J. A., & Rholes, W. S. (2002). Fearful-avoidance, disorganization, and multiple working models: Some directions for future theory and research. *Attachment and Human Development*, 4, 223-229.
- Sinclair, S. J., Blais, M. A., Gansler, D. A., Sandberg, E., Bistis, K., & LoCicero, A. (2010). Psychometric properties of the Rosenberg Self-Esteem Scale: Overall and across demographic groups living within the United States. *Evaluation & the Health Professions, 33*, 56-80.

- Sisk, C. L., & Zehr, J. L. (2005). Pubertal hormones organize the adolescent brain and behavior. *Frontiers in Neuroendocrinology*, 26, 163-174.
- Sizonenko, P. C. (1978). Endocrinology in preadolescents and adolescents. I. Hormonal changes during normal puberty. *American Journal of Diseases of Children*, 132, 704-712.
- Shrive, F. M., Stuart, H., Quan, H., & Ghali, W. A. (2006). Dealing with missing data in a multi-question depression Scale: A comparison of imputation methods. *BMC Medical Research Methodology*, 6: 57.
- Skjaerpe, P. A., Giwercman, Y. L., Giwercman, A., & Svartberg, J. (2009). Androgen receptor gene polymorphism and sex hormones in elderly men: The Tromsø study. Asian Journal of Andrology, 11, 222-228.
- Slatcher, R., B., Mehta, P. H., & Josephs, R. A. (2011). Testosterone and self-reported dominance interact to influence human mating behavior. *Social Psychological* and Personality Science, 2, 531-539.
- Smallbone, S. W., & Dadds, M. R. (2000). Attachment and coercive sexual behavior. Sexual Abuse: A Journal of Research and Treatment, 12, 3-15.
- Smallbone, S. W., & Dadds, M. R. (2001). Further evidence for a relationship between attachment insecurity and coercive sexual behavior in nonoffenders. *Journal of Interpersonal Violence*, 16, 22-35.
- Smith, L. L. & Hines, M. (2000). Language lateralization and handedness in women prenatally exposed to diethylstilbestrol (DES). *Psychoneuroendocrinology*, 25, 497-512.
- Smith, M. R., Rudd, B. T., Shirley, A., Rayner, P. H., Williams, J. W., Duignan, N. M., et al. (1975). A radioimmunoassay for the estimation of serum dehydroepiandrosterone sulphate in normal and pathological sera. *Clinica Chimica Acta*, 65, 5-13.
- Smith, E. A., Udry, J. R., & Morris, N. M. (1985). Pubertal development and friends: A biosocial explanation of adolescent sexual behavior. *Journal of Health and Social Behavior*, 26, 183-192.
- Spangler, G., Johann, M., Ronai, Z., & Zimmermann, P. (2009). Genetic and environmental influence on attachment disorganization. *Journal of Child Psychology and Psychiatry*, 50, 952–961.
- Spector, I. P., Carey, M. P., & Steinberg, L. (1996). The sexual desire inventory: Development, factor structure, and evidence of reliability. *Journal of Sex & Marital Therapy*, 22, 175–190.
- Spector, I. P., Carey, M. P., & Steinberg, L. (1998). Sexual Desire Inventory. In C. M. Davis, W. L. Yarber, R. Bauserman, G. Schreer, & E. L. Davis (Eds.), *Handbook* of Sexuality-Related Measures (pp. 174–176). Thousand Oaks, CA: Sage Publishers.
- Spector, I. P., & Fremeth, S. M. (1996). Sexual behaviors and attitudes of geriatric residents in long-term care facilities. *Journal of Sex & Marital Therapy*, 22, 235-246.
- Sroufe, L. A. (1979). The coherence of individual development: Early care, attachment, and subsequent developmental issues. *American Psychologist*, *34*, 834-841.
- Sroufe, L. A., Egeland, B., & Kreutzer, T. (1990). The fate of early experience following developmental change: Longitudinal approaches to individual adaptation in childhood. *Child Development*, 61, 1363-1373.
- Stanton, S. J., Liening, S. H., & Schultheiss, O. C. (2011a). Testosterone is positively associated with risk taking in the Iowa Gambling Task. *Hormones and Behavior*, *59*, 252-256.
- Stanton, S. J., Mullette-Gillman, O. A., McLaurin, R. E., Kuhn, C. M., LaBar, K. S., Platt, M. L., et al. (2011b). Low- and high-testosterone individuals exhibit decreased aversion to economic risk. *Psychological Science*, 22, 447-453.
- Stenstrom, E., Saad, G., Nepomuceno, M. V., & Mendenhall, Z. (2011). Testosterone and domain-specific risk: Digit ratios (2D:4D and rel2) as predictors of recreational, financial, and social risk-taking behaviors. *Personality and Individual Differences*, 51, 412-416.
- Strachman, A., & Impett, E. A. (2009). Attachment orientations and daily condom use in dating relationships. *Journal of Sex Research*, *46*, 319-329.
- Strong, R. K., & Dabbs Jr., J. M. (2000). Testosterone and behavior in normal young children. *Personality and Individual Differences*, 28, 900-915.
- Su, Q.-R., Su, L.-Y., Su, H.-R., & Chen, Q. (2007). Polymorphisms of androgen receptor gene in childhood and adolescent males with first-onset major depressive disorder and association with related symptomatology. *International Journal of Neuroscience*, 117, 903-917.
- Svensson, A. I., Åkesson, P., Engel, J. A., & Söderpalm, B. (2003). Testosterone treatment induces behavioral disinhibition in adult male rats. *Pharmacology Biochemistry and Behavior*, 75, 481-490.
- Svensson, A. I., Söderpalm, B., & Engel, J. A. (2000). Gonadectomy enhances shockinduced behavioral inhibition in adult male rats: Implications for impulsive

behavior. Pharmacology Biochemistry and Behavior, 65, 831-736.

- Swerdloff, R. S., & Odell, W. D. (1975). Hormonal mechanisms in the onset of puberty. *Postgraduate Medical Journal*, *51*, 200-208.
- Swerdloff, R. S., Wang, C., Hines, M., & Gorski, R. (1992). Effect of androgens on the brain and other organs during development and aging. *Psychoneuroendocrinology*, 17, 375-383.
- Tabachnick, B. G., & Fidell, L. S. (2007). *Using Multivariate Statistics (5th ed.)*. Boston, MA: Allyn & Bacon.
- Takahashi, T., Sakaguchi, K., Oki, M., Homma, S., & Hasegawa, T. (2006). Testosterone levels and discounting delayed monetary gains and losses in male humans. *Neuroendocrinology Letters*, 27, 439-444.
- Temcheff, C. E., Serbin, L. A., Martin-Storey, A., Stack, D. M., Hodgins, S., Ledingham, J., et al. (2008). Continuity and pathways from aggression in childhood to family violence in adulthood: A 30-year longitudinal study. *Journal of Family Violence*, 23, 231-242.
- Thompson, R. A., Lamb, M. E., & Estes, D. (1982). Stability of infant-mother attachment and its relationship to changing life circumstances in an unselected middle-class sample. *Child Development*, *53*, 144-148.
- Torgersen, A. M., Grova, B. K., & Sommerstad, R. (2007). A pilot study of attachment patterns in adult twins. *Attachment and Human Development*, *9*, 127-138.
- Tracy, J. L., & Shaver, P. R., Albino, A. W., & Cooper, M. L. (2003). Attachment styles and adolescent sexuality. In P. Florsheim (Ed.), *Adolescent Romantic Relations* and Sexual Behavior: Theory, Research, and Practical Implications (pp. 137-159). Mahwah, NJ: Lawrence Erlbaum.
- Trivers, R. L. (1985). Social Evolution. Menlo Park, CA: Benjamin/Cummings.
- Troisi, A., & D'Argenio, A. (2004). The relationship between anger and depression in a clinical sample of young men: The role of insecure attachment. *Journal of Affective Disorders, 79*, 269-272.
- Troy, M. & Sroufe, L. A. (1987). Victimization among preschoolers: Role of attachment relationship history. *Journal of the American Academy of Child and Adolescent Psychiatry*, 26, 166-172.
- Turan, B., Guo, J., Boggiano, M. M., & Bedgood, D. (2014). Dominant, cold, avoidant, and lonely: Basal testosterone as a biological marker for an interpersonal style. *Journal of Research in Personality*, 50, 84-89.

- Tweed, R. G., & Dutton, D. G. (1998). A comparison of impulsive and instrumental subgroups of batterers. *Violence and Victims, 13*, 217-230.
- van Anders, S. M. (2012). Testosterone and sexual desire in healthy women and men. *Archives of Sexual Behavior, 41*, 1471-1484.
- van Anders, S. M. (2010). Chewing gum has large effects on salivary testosterone, estradiol, and secretory immunoglobulin A assays in women and men. *Psychoneuroendocrinology*, *35*, 305-309.
- van Anders, S. M., Hamilton, L. D., & Watson, N. V. (2007). Multiple partners are associated with higher testosterone in North American men and women. *Hormones and Behavior*, *51*, 454-459.
- van Anders, S. M. & Goldey, K. L. (2010). Testosterone and partnering are linked via relationship status for women and 'relationship orientation' for men. *Hormones and Behavior*, *58*, 820-826.
- van Anders, S. M., Goldey, K. L., Conley, T. D., Snipes, D. J., & Patel, D. A. (2012). Safer sex as the bolder choice: Testosterone is positively correlated with safer sex behaviorally relevant attitudes in young men. *The Journal of Sexual Medicine*, 9, 727-734.
- van Anders, S. M. & Watson, N. V. (2006). Relationship status and testosterone in North American heterosexual and non-heterosexual men and women: Cross-sectional and longitudinal data. *Psychoneuroendocrinology*, 31, 715-723.
- van Anders, S. M., & Watson, N. V. (2007). Testosterone levels in women and men who are single, in long-distance relationships, or same-city relationships. *Hormones and Behavior*, *51*, 286-291.
- van der Meij, L., Almela, M., Buunk, A. P., Dubbs, S., & Salvador, A. (2012). 2D:4D in men is related to aggressive dominance but not to sociable dominance. *Aggressive Behavior*, 38, 208-212.
- van Honk, J., Tuiten, A., Verbaten, R., van den Hout, M., Koppeschaar, H., Thijssen, J., et al. (1999). Correlations among salivary testosterone, mood, and selective attention to threat in humans. *Hormones and Behavior, 36*, 17-24.
- van IJzendoorn, M.H., & Bakermans-Kranenburg, M.J. (2006). DRD47-repeat polymorphism moderates the association between maternal unresolved loss or trauma and infant disorganization. *Attachment & Human Development*, 8, 291-307.
- van IJzendoorn, M.H., Moran, G., Belsky, J., Pederson, D., Bakermans-Kranenburg, M.J., & Kneppers, K. (2000). The similarity of siblings' attachments to their

mother. Child Development, 71, 1086-1098.

- Vermeersch, H., T'Sjoen, G., Kaufman, J. M., Vincke, J., & Van Houtte, M. (2010). Testosterone, androgen receptor gene CAG repeat length, mood, and behaviour in adolescent males. *European Journal of Endocrinology*, 163, 319-328.
- Vermeulen, A., & Verdonck, G. (1992). Representativness of a single point plasma testosterone level for the long term hormonal milieu in men. *Journal of Clinical Endocrinology and Metabolism, 4*, 939-942.
- Vittek, J., L'Hommedieu, D. G., Gordon, G. G., Rappaport, S. C., & Southren, A. L (1985). Direct radioimmunoassay (RIA) of salivary testosterone: correlation with free and total serum testosterone. *Life Sciences*, 37, 711-716.
- Voracek, M., & Stieger, S. (2009). Replicated nil associations of digit ratio (2D:4D) and absolute finger lengths with implicit measures of aggression. *Psicothema*, 21, 382-389.
- Wacker, J., Mueller, E. M., & Stemmler, G. (2013). Prenatal testosterone and personality: Increasing the specificity of trait assessment to detect consistent associations with digit ratio (2D:4D). *Journal of Research in Personality*, 47, 171-177.
- Wade, T. J., Shanley, A., & Imm, M. (2004). Second to fourth digit ratios and individual differences in women's self-perceived attractiveness, self-esteem, and bodyesteem. *Personality and Individual Differences*, 37, 799-804.
- Wallen, K. (2009). The Organizational Hypothesis: Reflections on the 50th anniversary of the publication of Phoenix, Goy, Gerall, and Young (1959). *Hormones and Behavior*, 55, 561-565.
- Wallen, K., Maestripieri, D., & Mann, D. R. (1995). Effects of neonatal testicular suppression with a GnRH antagonist on social behavior in group-living juvenile rhesus monkeys. *Hormones and Behavior*, 29, 322-337.
- Waltz, J., Babcock, J. C., Jacobson, N. S., & Gottman, J. M. (2000). Testing a typology of batterers. *Journal of Consulting and Clinical Psychology*, 68, 658–669.
- Wang, C., Plymate, S., Nieschlag, E., & Paulsen, C. A. (1981). Salivary testosterone in men: Further evidence of a direct correlation with free serum testosterone. *Journal of Clinical Endocrinology and Metabolism*, 53, 1021-1024.
- Waters, E., Merrick, S., Treboux, D., Crowell, J., & Albersheim, L (2000). Attachment security in infancy and early adulthood: A twenty-year longitudinal study. *Child Development*, 71, 684-689.

Webster, G. D., & Bryan, A. (2007). Sociosexual attitudes and behaviors: Why two

factors are better than one. Journal of Research in Personality, 41, 917-922.

- Westberg, L., Henningsson, S., Landén, M., Annerbrink, K., Melke, J., Nilsson, S., et al. (2009). Influence of androgen receptor repeat polymorphisms on personality traits in men. *Journal of Psychiatry and Neuroscience*, 34, 205-213.
- Wei, M., Mallinckrodt, B., Larson, L. M., & Zakalik, R. A. (2005). Adult attachment, depressive symptoms, and validation from self versus others. *Journal of Counseling Psychology*, 52, 368-377.
- Weisman, O., Zagoory-Sharon, O., & Feldman, R. (2014). Oxytocin administration, salivary testosterone, and father-infant social behaviour. *Progress in Neuro-Psychopharmacology and Biological Psychiatry*, 49, 47-52.
- White, L. K., Brinkerhoff, D. B., & Booth, A. (1985). The effect of marital disruption on child's attachment to parents. *Journal of Family Issues*, *6*, 5-22.
- White, R. E., Thornhill, S., & Hampson, E. (2006). Entrepreneurs and evolutionary biology: The relationship between testosterone and new venture creation. *Organizational Behavior and Human Decision Processes*, 100, 21-34.
- Wiesner, M., & Ittel, A. (2002). Relations of pubertal timing and depressive symptoms to substance use in early adolescence. *Journal of Early Adolescence, 22*, 5-23.
- Williams, J. M., & Dunlop, L. C. (1999). Pubertal timing and self-reported delinquency among male adolescents. *Journal of Adolescence*, 22, 157-171.
- Williams, J. H. G., Greenhalgh, K. D., & Manning, J. T. (2003). Second to fourth finger ratio and possible precursors of developmental psychopathology in preschool children. *Early Human Development*, 72, 57-65.
- Williams, C. L. & Meck, W. H. (1991). The organizational effects of gonadal steroids on sexually dimorphic spatial ability. *Psychoneuroendocrinology*, 16, 155-176.
- Williams, N. L., & Riskind, J. H. (2004). Adult romantic attachment and cognitive vulnerabilities to anxiety and depression: Examining the interpersonal basis of vulnerability models. *Journal of Cognitive Psychotherapy: An International Quarterly, 18*, 7-24.
- Wilson, M., & Daly, M. (1985). Competitiveness, risk-taking, and violence: The young male syndrome. *Ethology and Sociobiology*, 6, 59-73.
- Wismer Fries, A. B., Shirtcliff, E. A., & Pollak, S. D. (2008). Neuroendocrine disruption following early social deprivation in children. *Developmental Psychobiology*, 50, 588-599.

- Yang, C. J., Hooven, C. K., Boynes, M., Gray, P. B., & Pope Jr., H. G. (2007). Testosterone levels and mental rotation performance in Chinese men. *Hormones and Behavior*, 51, 373-378.
- Zeifman, D., & Hazan, C. (2008). Pair bonds as attachments. Reevaluating the evidence. In J. Cassidy & P. R. Shaver (Eds.), *Handbook of Attachment. Theory, Research, and Clinical Applications* (pp. 436-455). New York, NY: Guilford Press.
- Zheng, Z., & Cohn, M. J. (2011). Developmental basis of sexually dimorphic digit ratios. *Proceedings of the National Academy of Science, 108*, 16289-16294.
- Zitzmann, M. (2009). The role of the CAG repeat androgen receptor polymorphism in andrology. *Frontiers in Hormone Research*, *37*, 52-61.
- Zitzmann, M., & Nieschlag, E. (2003). The CAG repeat polymorphism within the androgen receptor gene and maleness. *International Journal of Andrology, 26*, 76-83.
- Zlokovich, M. S., & Snell, W. E. (1997). Contraceptive behavior and efficacy: The influence of illusion of fertility control and adult attachment tendencies. *Journal of Psychology & Human Sexuality*, *9*, 39-55.
- Zook, A., & Sipps, G.J. (1985). Cross-validation of a short form of the Marlowe–Crowne Social Desirability Scale. *Journal of Clinical Psychology*, *41*, 236–238.
- Zuloaga, D. G., Poort, J. E., Jordan, C. L., & Breedlove, S. M. Male rats with the testicular feminization mutation of the androgen receptor display elevated anxiety-related behavior and corticosterone response to mild stress. *Hormones and Behavior*, *60*, 380-388.

Appendices

Appendix A: Poster for Recruitment from Western Campus

MALES Needed

for a study of Testosterone, Personality, and Experiences in Close Relationships

Eligibility:

- Heterosexual MALE age 18 35
- Open to undergraduates and staff at UWO
- Must have had **at least** one romantic or sexual partner, either currently or in the past 1 year
- Fluent in English

The experiment will last approximately **1 hour**.

You will be compensated **\$15.00** for your participation.

You will be asked to:

- Fill out questionnaires about your personality and relationships with romantic or sexual partners
- Fill out questionnaires about childhood experiences with parents/caregivers
- Provide saliva to measure testosterone levels
- Provide saliva to determine the genetic makeup of your hormone receptor

For further information or to make an appointment, please e-mail: shipstudy@gmail.com

Appendix B: Order of Questionnaire Presentation

- 1) Part I Composite JPI-R/PRF-E/MC-C Questionnaire
- 2) Demographics Questionnaire



13) POMS

Note. JPI-R = Jackson Personality Inventory Revised; PRF-E = Personality Research Form E; MC-C = Marlowe-Crowne Social Desirability Scale Form C; ECR = Experiences in Close Relationships Questionnaire; SOI = Sociosexual Orientation Inventory; RQ = Relationship Questionnaire; SDI = Sexual Desire Inventory; RAQ = Retrospective Attachment Questionnaire; RSE = Rosenberg Self-Esteem Scale; AQ = Aggression Questionnaire; POMS = Profile of Mood States

Appendix C: Demographics Questionnaire and Dating History Questionnaire

Demographics Questionnaire

Please provide the information requested below. If, for any reason, you wish not to answer a particular item, you are free to leave it blank. All information provided is strictly confidential and will be used for research purposes only. Your name or other identifying information will not appear anywhere on this questionnaire.

Subject Number:	Date: Testing time:			
Age:				
1. In the 30 minute period before your appointme	ent today, did	you:		
 (a) have anything to eat? (b) have a beverage other than water? (c) have a cigarette? (d) brush your teeth? (e) chew gum? 	YES YES YES YES YES	NO NO NO NO		
2. In the past week before your appointment toda	ay, did you:			
(a) have an acute illness (e.g., cold or flu)? Y	ES	NO	
3. What time do you normally wake up in the mo	rning?			
On weekdays: On weekends:				
4. What time did you wake up today?				
5. Has anything happened in your life over the pa few days that made you feel unusually stressed	ast Y. 1?	ES	NO	
6. What is your height? (feet &	inches) OR	L	(cm)	
7. What is your weight? (pound	s) OR _		(kg)	
8. Do you attend/work at Western or Fanshawe?	WESTE	RN F.	ANSHAWE	

9. Are you a smoker, non-smoker, or an occasional smoker? **SMOKER** NON-SMOKER OCCASIONAL SMOKER 10. How often do you normally consume alcohol? (Circle one number from 0 to 4). 0 1 2 3 4 1-2 times Never 1-2 times 3-4 times Almost a **month** a week a week every day 11. What is the average number of drinks you have when/if you drink? (Circle one number). 0 1 2 3 4 one to three four to seven eight to twelve more than 12 None 12. How long has it been since you last consumed an alcoholic beverage (e.g., beer, wine, spirit) of any kind? 13. Have you ever had any accidents with either of your hands that could affect the growth of your fingers? 14. Are you currently taking any prescription or YES NO non-prescription medications? If so, please list the medications you are taking. 15. Do you currently have any physical condition(s) that might cause your hormone levels to be unusual? YES NO (e.g., diabetes, thyroid, etc.). 16. What is your ethnicity? East Asian Middle Eastern White Black _____ South Asian Pacific Islander First Nations Hispanic Other

17. How many brothers do you have? _____ What are their ages? _____

18. How many sisters do you have? _____ What are their ages? _____

19. Up to age 16, I was raised by (check all applicable answers and indicate ages):

Both biological parents living together		From ages	to
Both biological parents, living separately		From ages	to
A biological parent and a step-parent		From ages	to
One biological parent living alone		From ages	to
Another family member (not a parent)		From ages	to
Adoptive parents		From ages	to
Foster parents		From ages	to
Other			
(please specify:)	From ages	to
20. Is English your first language?	YES		NO
If no, what is your first language?			

21. The following information should be filled out for the parents who raised you. If you grew up in a single-parent home, fill out the information only for the parent you lived with:

MOTHER

Mother's Usual Occupation or Job (be very specific; e.g., Insurance adjuster):

Mother's Highest Education (check one):

 Grade 6 or less

 Grade 7 to 9

 Some high school

 High school graduate

 At least one year of college or other specialized training

 College or university graduate

 Master's degree, Ph.D., M.D., or other professional

 degree

Province or country where your mother went to elementary and high school:

FATHER

Father's Usual Occupation or Job: (be very specific; e.g., Insurance adjuster):

Father's Highest Education (check one):

 Grade 6 or less
 Grade 7 to 9
 Some high school
 High school graduate
 At least one year of college or other specialized training
 College or university graduate
 Master's degree, Ph.D., M.D., or other professional
degree

Province or country where your father went to elementary and high school:

22. What is <u>your</u> highest level of education (check one):

Grade 6 or less
Grade 7 to 9
Some high school
High school graduate
At least one year of college or other specialized training
College or university graduate
Master's degree, Ph.D., M.D., or other professional
degree
5

23. If not a student, what is your usual occupation?

Dating History Questionnaire

1. Your marital/partner status:	 Single	 Steady live-in partner
	 Steady partner but living apart	 More than one steady partner at present
	 Married	 Divorced
	 Separated	 Widowed

2. How long did your most **recent** romantic or sexual relationship last? (Do not include your current relationship, if any):

3. What is the **typical** length of a romantic or sexual relationship for you?

4. If you are sexually active, at what age did you first have sexual intercourse?

5. In total, how many romantic or sexual relationships have you had in your lifetime? (Please count all intimate relationships, whether or not they included sexual behavior)

6. How many children do you have? (e.g., 0, 1, 2, unknown etc.)

7. Is there any reason why your answers today might not be typical?

Appendix D: Detailed Description of Indicators of Latent Constructs in the LHT-Based Model

Early Family Environment

Because all the life history theories reviewed highlight the role of parental sensitivity as potentially important early environmental predictor of psychological and somatic development, measures of "family affection" and "family abuse" were included as indicator variables. Using items from Part I of the RAQ, which asks adults about their childhood relationships with parents (Parkes, 2006), the "family affection" index (α = .58) was defined as the sum of four items measuring the level of attention and affection the participant received from or felt towards his parents during childhood: Was your parent inconsistent, sometimes responding, and at other times ignoring your needs for attention and affection? Were you unusually close to your parent? Was either parent unable to show warmth or to hug or cuddle you? Did you have mixed feelings of love and hate, affection and resentment, towards either parent? Responses to these yes/no questions were summed for both parents, yielding a total of 8 items.

The "family abuse" index (α = .48) was made up of the sum of 12 yes/no items from Part I of the RAQ (6 for each parent), measuring the level of physical or psychological abuse in the participant's family: Did your parent ever assault or injure his or her partner? Did either parent obtain your obedience by threatening to leave you or give you away? Did either parent drink more alcohol than was good for them? Was either parent inclined to tease you or make you feel small? Did either parent beat you or physically punish you more than most parents? Did either parent sexually interfere with you or expect you to touch their genitals?

Because some researchers (e.g., Clarke-Stewart et al., 2000; James et al., 2012) have suggested parental psychopathology may influence the ecology of the early family environment, "Mother's Psychiatric Health" ($\alpha = .64$) and "Father's Psychiatric Health" ($\alpha = .56$) were included as manifest variables. Five items from the RAQ Part I made up each of these measures: Was either parent nervous, insecure or a worrier? Was your parent subject to episodes of gloom or depression? Did your parent ever receive psychiatric treatment? Was your parent ever admitted to a hospital for psychiatric treatment? Did either parent threaten to kill themselves? The total score was the sum of the yes/no responses to each of these questions, computed for each parent independently.

Within their theories, both Belsky et al. (1991) and Chisholm (1999) suggest that the availability of financial resources may have an impact on the nature of the early family context, and empirical studies have provided further support for this viewpoint (e.g., Arim et al., 2011; Chasiotis et al., 1998; Hill et al., 1994a; James et al., 2012). Consequently, SES of the household was included as an indicator variable. Within the demographics questionnaire, participants provided occupation and education information for the parents who raised them from ages 0-16 years. Following Hollingshead (1975), the level of education attained by each parent was rated on a 1-7 scale, where 1 = "Grade 6 or less" and 7 = "Master's degree, Ph.D., M.D., or other professional degree", and occupational status was rated on 1-9 scale, categorized according to the earning potential of the profession, such that a more lucrative profession (e.g., Physician: score of 9) received a higher occupational status score than a less lucrative profession (e.g., Dishwasher: score of 1). Parents who were unemployed (including stay-at-home parents) received a score of 0. The list of professions falling into the different occupational categories was provided by Hollingshead (1975).

SES was computed according to the procedure outlined by Deonandan et al. (2000) where the highest occupation score in the household (which could come from either parent) was summed with the average of the education score for both parents. In this way, the influence of a parent who was not contributing financially to the participant's early environment (e.g., homemaker) but may have been contributing to the overall SES by way of their educational status (e.g., helping to provide an intellectually stimulating home environment) could be taken into consideration. If the participant indicated that he was raised by a single parent, then total SES was computed as the sum of the occupation and education score for that one parent. If the participant reported that he was raised in a divorced household where he lived with each parent separately, then SES = (sum of occupation and education of mother + sum of the occupation and

education of father)/2. For participants who reported that a parent and a stepparent raised them, SES was computed the same as for a two-parent household, since both adults were deemed to be contributing to the home environment.

Because the life history theories of attachment and reproduction appear to discuss the potential importance of financial stability in the early family environment, a second measure of SES based purely on employment was also computed, emphasizing financial resources. For two-parent households this translated into the sum of the occupation scores for each of the two parents; for one-parent households this was equal to the occupation score for the one parent who raised the participant; and for divorced households where both parents were raising the child independently this was equal to the sum of the employment score for both parents divided by 1.5. In this way, a two-parent (or parent and stepparent) household had the potential of receiving the highest score (highest level of financial resources), a one-parent household had the potential of receiving the lowest score (lowest level of financial resources), and a divorced household fell in between, as the child would only be able to gain exposure to each parent's financial resources than a child living in a single-parent home.

Parental presence/absence has been explored both theoretically and experimentally as a factor within the early family environment that may predict subsequent psychological (i.e., childhood attachment), somatic, and reproductive strategy development (e.g., Chisholm, 1999; Draper & Harpending, 1982; Hill et al., 1994a; James et al., 2012; Miller & Pasta, 2000). There is evidence, however, that not just parental absence, but also the nature of the family structure (i.e., two-parent household, divorced, single-parent) may affect attachment style, such that a less intact family structure is associated with higher levels of attachment insecurity in children (e.g., Nair & Murray, 2005; Clarke-Stewart et al., 2000). Children from divorced families, on average, have been found to experience more problems and have a lower level of well-being than children from continuously intact two-parent families (Amato & Keith, 1991), and these effects may extend into adulthood (e.g., Cherlin, Chase-Lansdale, & McRae, 1998; Hetherington, Bridges, & Insabella, 1998). Marital breakup may also have negative psychological consequences by way of its effect on the child's access to financial resources and by potentially diminishing the capacity of the parent to rear the child due to the emotional and financial distress of the divorce (Clarke-Stewart et al., 2000). Young children from single-parent/never married households have also been found to be less securely attached than their two-parent household counterparts. Moreover, they display more negative interactions with mothers than children from separated/divorced homes (Clarke-Stewart et al., 2000) and are more economically disadvantaged than children from divorced households (e.g., Bianchi, 1995). With respect to families with a biological parent and a cohabiting stepparent, there is some evidence that biological mother-partner families are at higher risk of emotionally and physically maltreating children compared to biological two-parent households, but the maltreatment risk is higher in single parent (i.e., single mother) families, particularly if the mother works (Berger, 2004). By contrast, Gibson-Davis (2008) found that biological mothers with a cohabiting male stepparent rated their partners' level of involvement in family life (i.e., positive engagement, frequency of spanking, and instrumental family support) similarly to mothers married to the biological father.

Some research suggests that changes in family structure have greater developmental effects when the disruption occurs earlier in childhood (e.g., Allison & Furstenberg, 1989; Howell, Portes, & Brown, 1997), but other studies do not find any apparent age-related effect (e.g., Furstenberg & Teitler, 1994; White, Brinkerhoff, & Booth, 1985). McLanahan and Bumpass (1988) postulate that family structure disruption occurring in early as well as late childhood/early adolescence may be important. Consequently, three indicator variables measuring the effect of family structure on the quality of participants' environmental experience were included. The first variable encapsulated the ages of 0-4 years, the second variable ages 5-9 years, and the third variable ages 10-16 years. Based on participants' responses to a demographic item inquiring about the family structures in which they were raised across the ages of 0-16 years, participants received a score on a 1-4 scale for each of the 3 age ranges, where a higher score was indicative of being raised in a more stressful family structure. Thus for each of the age ranges a score of 1 was given for having two biological (or adoptive) parents living together, a score of 2 was given for a biological parent and a stepparent, a score of 3 was given for both biological parents living separately, and a score of 4 was given for either one biological parent living alone, being raised by another family member who was not a biological parent, or for non-adoptive foster parents.

Because the presence/absence of a non-biological stepparent has been investigated in studies of the psychological and somatic outcomes of family antecedents (e.g., Arim et al., 2012), an indicator variable categorizing whether participants did or did not have a stepparent over the course of the first 16 years of their lives (0 = no stepparent; 1 = stepparent) was included. Finally, because the death of a parent during childhood can have serious emotional and physical consequences, and can certainly disrupt a growing child's family structure, an indicator variable categorizing whether or not participants had lost a parent to death during the first 16 years of their lives (0 = both parents alive; 1 =experienced a parent death) was included. Information regarding parent death was obtained from the second question on Part I of the RAQ, which asks if the participant's parents are alive, and if not, how old was the participant when his parent(s) died.

Each manifest (observed) variable was coded such that a higher score was indicative of a poorer quality early family environment. In the case of SES, where a higher score indicated better socioeconomic status, scores were reversed so that a higher score corresponded with a lower SES.

Retrospective Childhood Attachment

Although individual scores for the three insecure attachment patterns (i.e., anxious/ambivalent, avoidant, disorganized) can be separately computed, because Belsky et al. (1991) and Chisholm (1999) speculate about the causes and consequences of overall attachment insecurity (*vs.* security) without differentiating among the varieties of insecure attachment in their theoretical models of attachment and reproduction, the decision was made to only compute the RAQ total insecure attachment score as a single indicator of retrospective childhood attachment style.

Non-Sexual Reproductive Strategy

Although not discussed as a formal part of the Young Male Syndrome, the rationale for including self-esteem and trait anxiety as indicators of the non-sexual reproductive strategy is that higher levels of self-esteem and lower levels of anxiety may provide an advantage for young males who need to defend their social status in anticipation of early reproduction, and thus are conceptually similar to the other traits that comprise the Young Male Syndrome (Del Giudice, 2009). Because the opportunistic reproductive approach, which may run parallel to a more avoidant adult attachment style in men, may, in part, be characterized by more independent and less affiliative social behaviour, level of need for autonomy was also hypothesized to be an indicator of participants' non-sexual reproductive strategy.

Each manifest (observed) variable of the Non-Sexual Reproductive Strategy was coded such that a higher score was indicative of a more opportunistic non-sexual personality profile (i.e., more aggressive, more impulsive, less anxious). In the case of the anxiety subscale of the POMS, where higher scores indicated more anxiety, scores were reversed so that a higher score corresponded to a lower level of anxiety.

Sexual Reproductive Strategy

Data for the age at first intercourse, typical relationship length, and total number of lifetime partners indicator variables were obtained from responses to questions on the Dating History Questionnaire. Of note, because sexual experience was not a pre-requisite for study participation, some participants had not yet had sexual intercourse. Review of the frequency distribution for the "age at first intercourse" variable revealed that nearly all participants in the sample had had their first sexual experience by the age of 19 (89% of the sample). In order to not code as missing those participants who were sexually inactive, inactive participants who were aged 18 or 19 years at the time of testing were assigned a score of 19 years on this variable, given the data from the observed frequency distribution. For those participants who were older than 19 years and had never had sex, they were assigned a score that corresponded to their current age. In total, only 7 data points (3.6% of the sample) had to be handled in this manner.

As mentioned in Section 3.2.1, a current partnering status observed variable was also included. Participants were scored on a 3-point scale ranging from 0-2. A score of 0 was given to participants who indicated on the Dating History Questionnaire that they had a steady partner but living apart, a steady live in partner, or were married. A score of 1 was given to participants who were single, divorced, or separated. A score of 2 was given to participants who indicated that they had more than one steady partner at present. Thus, a higher score corresponded to a conceivably more opportunistic mating status. Moreover, the coding also corresponded to what the testosterone literature predicts is a higher T partnering status (van Anders & Watson, 2007).

With respect to the set of core indicators derived from the SOI, although Item 2 ("How many different partners do you foresee yourself having sex with during the next five years?") and Item 3 ("With how many partners have you had sex on *one and only one* occasion, in your lifetime?") of the SOI were included as independent indicators, Item 1 ("With how many different partners have you had sex (sexual intercourse) within the past year?") was omitted because of its redundancy with the "total number of lifetime partners" indicator variable.

The SOI assesses both behavioural and attitudinal facets of sociosexuality (Webster & Bryan, 2007). Items 4-8 corresponded to the attitudinal portion of the SOI. One item, Item 7, was omitted from the computation of the attitudinal SOI indicator variable because of its long and complicated wording which has been found in previous work to lead to measurement problems with less attentive participants (Penke & Asendorpf, 2008). Within the current sample, many participants appeared to respond to Item 7 in a manner inconsistent with their responses on the remaining attitudinal items, possibly due to the fact that Item 7 is the only question on the SOI that is negatively worded (i.e., a higher score corresponds to a more restricted sociosexual orientation), making it semantically difficult to decode. A principal components analysis was conducted to confirm that the remaining 4 items loaded onto a single factor, and participants' standardized factor scores became the indicator variable, a common procedure used to handle the different response scaling on some of the SOI items (Simpson & Gangestad, 1991).

As noted in Section 3.2.1, there is a lack of clarity in the life history models about the role of adult fearful attachment (e.g., Del Giudice, 2009). Many studies (e.g., Brennan & Shaver, 1995; Cooper et al., 1998; Feeney & Noller, 1992; Simpson, 1990) investigating the relationship between adult attachment and sexual behaviours have employed measures that prevent a distinction between the two avoidant attachment classifications (i.e., dismissing-avoidant and fearful-avoidant) (Shaver & Mikulincer, 2002; Simpson & Rholes, 2002). That being said, some research has found that fearful attachment is positively correlated with indices of short-term mating in men, as well as more out-of-control or addictive sexual behaviour (e.g., Schmitt, 2005; Carnes, 1991; Faisandier, Taylor, & Salisbury, 2012). It is believed that fearful-avoidants' motivation for avoiding intimacy is due to a lack of trust towards others and/or to prevent being hurt by partners (Bartholomew, 1990), which differs from dismissing-avoidants who adopt an avoidant orientation to maintain self-reliance and independence. Nevertheless, men with high fearful attachment may still be apt to engage in sex opportunistically and without commitment in an effort to avoid intimacy, regardless of the motive (but see Schachner & Shaver, 2004). For these reasons, participants' RQ continuous fearful score was included as an indicator variable in the Sexual Reproductive Strategy latent construct, along with the other continuous measures of romantic attachment avoidance, anxiety, and overall security.

Although a measure of commitment in a current romantic relationship was administered, and would certainly be a relevant indicator of the Sexual Reproductive Strategy latent construct, the Commitment Scale (Lund, 1985) by definition can only be completed by participants currently in a relationship. At the time of testing only N = 100were in a current relationship and thus completed the Commitment questionnaire. The proportion of missing data on this variable was, therefore, too high to justify imputation of any sort, even though the data were considered to be missing at random (missingness that is not related to the missing value itself (i.e., commitment scale question), but is related to the values of another variable in the data set (i.e., current partnering status) (Kline, 2011). Consequently, it was not possible to include level of actual relationship commitment as a manifest variable in the model.

Each manifest (observed) variable of the Sexual Reproductive Strategy was coded such that a higher score was indicative of a shorter-term reproductive strategy. Participants' responses to age at first intercourse and typical relationship length were reversed so that a higher score corresponded to an earlier age at first intercourse and a shorter typical relationship length. Similarly, the anxious attachment observed variables were reversed so that a higher score reflected lower levels of anxiety.

Appendix E: Ethics Approval Forms



Department of Psychology The University of Western Ontario Room 7418 Social Sciences Centre, London, ON, Canada N6A 5C1 Telephone:

Use of Human Subjects - Ethics Approval Notice

Review Number	12 02 15	Approval Date	12 02 16
Principal Investigator	Elizabeth Hampson/Janani Sankar	End Date	13 02 16
Protocol Title	Associations between testosterone, personality, a	and experiences in close relatio	nships
Sponsor	n/a	4	

This is to notify you that The University of Western Ontario Department of Psychology Research Ethics Board (PREB) has granted expedited ethics approval to the above named research study on the date noted above

The PREB is a sub-REB of The University of Western Ontario's Research Ethics Board for Non-Medical Research Involving Human Subjects (NMREB) which is organized and operates according to the Tri-Council Policy Statement and the applicable laws and regulations of Ontario. (See Office of Research Ethics web site: http://www.uwo.ca/research/ethics/)

This approval shall remain valid until end date noted above assuming timely and acceptable responses to the University's periodic requests for surveillance and monitoring information.

During the course of the research, no deviations from, or changes to, the protocol or consent form may be initiated without prior written approval from the PREB except when necessary to eliminate immediate hazards to the subject or when the change(s) involve only logistical or administrative aspects of the study (e.g. change of research assistant, telephone number etc). Subjects must receive a copy of the information/consent documentation.

Investigators must promptly also report to the PREB:

a) changes increasing the risk to the participant(s) and/or affecting significantly the conduct of the study; b) all adverse and unexpected experiences or events that are both serious and unexpected;

c) new information that may adversely affect the safety of the subjects or the conduct of the study.

If these changes/adverse events require a change to the information/consent documentation, and/or recruitment advertisement, the newly revised information/consent documentation, and/or advertisement, must be submitted to the PREB for approval.

Members of the PREB who are named as investigators in research studies, or declare a conflict of interest, do not participate in discussion related to, nor vote on, such studies when they are presented to the PREB.



Chair, Psychology Expedited Research Ethics Board (PREB)

The other members of the 2011-2012 PREB are: Mike Atkinson (Introductory Psychology Coordinator), Rick Goffin, Riley Hinson Albert Katz (Department Chair), Steve Lupker, and Karen Dickson (Graduate Student Representative)

CC: UWO Office of Research Ethics

This is an official document. Please retain the original in your files



Department of Psychology The University of Western Ontario Room 7418 Social Sciences Centre, London, ON, Canada N6A 5C1 Telephone:

Use of Human Subjects - Ethics Approval Notice

Review Number	12 02 15	Approval Date	12 02 16 Extension of end date approved on 13 01 31
Principal Investigator	Elizabeth Hampson/Janani Sankar	End Date	13 02 16 New End Date 14 01 30
Protocol Title	Associations between testosterone, personality,	and experiences in close relatio	nships
Sponsor	n/a		

This is to notify you that The University of Western Ontario Department of Psychology Research Ethics Board (PREB) has granted expedited ethics approval to the above named research study on the date noted above.

The PREB is a sub-REB of The University of Western Ontario's Research Ethics Board for Non-Medical Research Involving Human Subjects (NMREB) which is organized and operates according to the Tri-Council Policy Statement and the applicable laws and regulations of Ontario. (See Office of Research Ethics web site: http://www.uwo.ca/research/ethics/)

This approval shall remain valid until end date noted above assuming timely and acceptable responses to the University's periodic requests for surveillance and monitoring information.

During the course of the research, no deviations from, or changes to, the protocol or consent form may be initiated without prior written approval from the PREB except when necessary to eliminate immediate hazards to the subject or when the change(s) involve only logistical or administrative aspects of the study (e.g. change of research assistant, telephone number etc). Subjects must receive a copy of the information/consent documentation.

Investigators must promptly also report to the PREB: a) changes increasing the risk to the participant(s) and/or affecting significantly the conduct of the study;

b) all adverse and unexpected experiences or events that are both serious and unexpected;

c) new information that may adversely affect the safety of the subjects or the conduct of the study.

If these changes/adverse events require a change to the information/consent documentation, and/or recruitment advertisement, the newly revised information/consent documentation, and/or advertisement, must be submitted to the PREB for approval.

Members of the PREB who are named as investigators in research studies, or declare a conflict of interest, do not participate in discussion related to, nor vote on, such studies when they are presented to the PREB.



Chair, Psychology Expedited Research Ethics Board (PREB)

The other members of the 2011-2012 PREB are: Mike Atkinson (Introductory Psychology Coordinator), Rick Goffin, Riley Hinson Albert Katz (Department Chair), Steve Lupker, and Karen Dickson (Graduate Student Representative)

CC: UWO Office of Research Ethics

This is an official document. Please retain the original in your files

Janani S. Sankar Curriculum Vitae

Post-Secondary Education

2015	Doctor of Philosophy, Clinical Psychology The University of Western Ontario Research Advisor: Elizabeth Hampson, Ph.D., C. Psych. Dissertation: The Role of Androgens in Life History Theories of Attachment
2009	Master of Science, Clinical Psychology The University of Western Ontario Research Advisor: Elizabeth Hampson, Ph.D., C. Psych. Thesis: Influence of Testosterone and Androgen Receptor Polymorphism on Visual-Spatial Cognition in Adult Men
2007	 Bachelor of Science, Psychology, First Class Honours McGill University Research Advisor (U4): Barbara Sherwin, Ph.D. Thesis: Verbal Memory Function during Pregnancy and the Early Postpartum Period Research Advisor (U3): Frances Aboud, Ph.D. Thesis: Peer Relations and Friendship Quality in Two-Way Immersion Programs

Publications and Presentations

Peer-Reviewed Publications

- Sankar, J. S., & Hampson, E. (2012). Testosterone levels and androgen receptor gene polymorphism predict specific symptoms of depression in young men. *Gender Medicine*, *9*, 232-243.
- Hampson, E., & Sankar, J. S. (2012). Hand preference in humans is associated with testosterone levels and androgen receptor gene polymorphism. *Neuropsychologia*, 50, 2018-2025.
- Hampson, E., & Sankar J.S. (2012). Re-examining the Manning hypothesis: Androgen receptor polymorphism and the 2D:4D digit ratio. *Evolution and Human Behavior*, *33*, 557-561.
- Aboud F.E., & Sankar J. (2007). Friendship and identity in a language-integrated school. *International Journal of Behavioral Development, 31,* 445-453.

Conference Proceedings

- Sankar, J.S. on behalf of Advocacy through Action (2013, June), *Evaluation of a student-driven mental health literacy and advocacy initiative: The Finding Your Way public lecture series*. Poster session presented at the 74th Annual Convention of the Canadian Psychological Association, Quebec City, Quebec.
- Sankar, J.S. on behalf of Advocacy through Action (2013, June), *Increasing the footprint of psychology in the community: A conversation on grass-roots mental health literacy and advocacy initiatives.* Round-table conversation hour presented at the 74th Annual Convention of the Canadian Psychological Association, Quebec City, Quebec.
- Sankar, J. S., Hampson, E. (2011, March), *Testosterone levels and androgen receptor gene polymorphism predict depressive symptomatology in young men*. Poster session presented at the 1st annual Rotman Research Institute and the Kunin-Lunenfeld Applied Research Unit Neuroscience of Emotion and Emotion-Related Disorders Conference, Toronto, Ontario.
- Sankar, J. S., Hampson, E. (2010, February), Associations between handedness and spatial reasoning in men: Possible influence of androgen receptor gene polymorphism. Poster session presented at the 38th Annual Meeting of the International Neuropsychological Society, Acapulco, Mexico.
- Sankar, J.S., Hampson, E. (2009, June), *Testing the link between the 2D:4D digit ratio and spatial orientation and visualization in adult men.* Poster session presented at the 13th Annual Meeting of the Society for Behavioral Neuroendocrinology, East Lansing, Michigan.

Published Abstracts:

- Sankar, J. S., Hampson, E. (2011). Testosterone levels and androgen receptor gene polymorphism predict depressive symptomatology in young men. *Rotman-Baycrest and KLARU Neuroscience of Emotion and Emotion-Related Disorders Abstracts*, 32.
- Sankar, J. S., Hampson, E. (2010). Associations between handedness and spatial reasoning in men: Possible influence of androgen receptor gene polymorphism. *Journal of the International Neuropsychological Society*, 16 (Suppl. 1), 214.
- Sankar, J.S., Hampson, E. (2009). Testing the link between the 2D:4D digit ratio and spatial orientation and visualization in adult men. *Society for Behavioral Neuroendocrinology Abstracts*, P3.67, 194.

Recognition of Scholarship

Honours and Awards:

• Queen Elizabeth II Graduate Scholarship in Science and Technology, 2012 (\$10,000)

- NSERC Alexander Graham Bell Canada Graduate Scholarship, 2009-2012 (\$105,000)
- Ontario Graduate Scholarship, 2009 (\$15,000; declined)
- Ontario Graduate Scholarship, 2008 (\$15,000)
- Western Graduate Research Scholarship, 2008 (\$8,000)
- CIHR Canada Graduate Scholarship, 2007 (\$17,500)
- Fond de Recherche en Santé du Québec, 2006 (\$5,366)
- Dow-Hickson Undergraduate Scholarship, 2006 (\$2,000)
- Emily Ross Crawford Undergraduate Scholarship, 2006 (\$1,000)

Related Work Experience

Teaching Assistantships:

- Abnormal Psychology—Tutorial Instructor (2012-2013)
- Exceptional Children: Developmental Disorders (2012)
- Neuropsychology and Cognitive Neuroscience (2009)
- Sex Differences in Human Brain and Behaviour (2008)
- Introduction to Psychology (2007-2008)

Supervisory Roles:

• Independent Studies Research Advisor, The University of Western Ontario (2012)

Project Title: Do Experiences in Close Relationships Change Across the Seasons?

Professional Service

Professional Association Membership:

- Canadian Psychological Association, Student Affiliate (2012-present)
- Ontario Psychological Association, Student Affiliate (2011-present)
- London Regional Psychological Association, Student Member (2010-present)
- Society for the Study of Clinical Psychology, Student Member (2010-present)
- Society for Behavioral Neuroendocrinology, Student Member (2009-2012)